

United States International Trade Commission

Recent Trends in U.S. Services Trade:

2025 Annual Report

July 2025 Publication Number: 5643 Investigation Number: 332-605 United States International Trade Commission

Commissioners

Amy A. Karpel, Chair

David S. Johanson

Jason E. Kearns

Catherine DeFilippo Director, Office of Operations

Jeremy Wise Director, Office of Industry and Competitiveness Analysis

> Address all communications to Office of External Relations (<u>externalrelations@usitc.gov</u>) United States International Trade Commission Washington, DC 20436



United States International Trade Commission

Recent Trends in U.S. Services Trade:

2025 Annual Report

July 2025 Publication Number: 5643 Investigation Number: 332-605 This report was prepared principally by:

Project Leader

Isaac Wohl

Deputy Project Leaders

Sharon Fisher, John Stefanick

Office of Industry and Competitive Analysis

Eric Forden, Theron Gray, Junie Joseph, Tamar Khachaturian, Jennifer Powell, Robert Stark, Rudy Telles Jr.

Content Reviewers

Tamara Gurevich, Kristin Smyth

Editorial Reviewers

Rick Docksai, Judy Edelhoff, Brian Rose

Legal Reviewer

Eric Heath

Statistical Reviewers, Data Support, and Interactive Tables

Ann Marie Carton, Maureen Letostak, Cindy Payne

Production Support

David Lahrmer, Fara Ndiaye, Muriel Williams

Under the Direction of

Martha Lawless, Chief Services Division Office of Industry and Competitiveness Analysis

Preface

This report is the 29th in a series of annual reports on recent trends in U.S. services trade published by the U.S. International Trade Commission. The purpose of this report is to apprise Congress and the President of global industry trends, regional developments, and competitiveness issues related to trade in services. Services represent the largest part of the U.S. economy, and the United States runs a significant surplus in services trade. This report was self-instituted in response to congressional interest in establishing a systematic means of examining and reporting on developments in services trade, by services sector, and with leading U.S. trading partners. The Commission instituted the investigation producing this series of reports under section 332(b) of the Tariff Act of 1930.¹ These reports draw on discussions with representatives from industry, academia, and international organizations as well as published sources to inform the Commission's legislative and executive branch requestors and the public.

¹ 19 U.S.C. § 1332(b). On August 27, 1993, the Commission instituted Investigation No. 332-345, *Annual Reports on U.S. Trade Shifts in Selected Industries*. On December 20, 1994, the Commission expanded the scope of the annual *Trade Shifts* report to include more detailed coverage of services industries. The Commission began publishing *Recent Trends in U.S. Services Trade* as a separate report in 1997.

Table of Contents

Preface		5
Table of Conte	ents	7
Abbreviations	s and Acronyms	13
Executive Sun	nmary	15
Chapter 1	Introduction	19
Data Sources,	, Categories, and Limitations	20
U.S. Services	Sector	24
Global Service	es Trade	26
U.S. Cross-Bo	rder Services Trade	28
Cross-Border	Trade: Preliminary 2024 Data	33
Affiliate Trans	actions	34
Bibliography .		38
Chapter 2	Professional Services	41
Overview		41
Cross-Border	Exports and Imports	41
Affiliate Trans	actions	46
Professional S	Services by Sector: Background and Trade Trends	48
Legal Servio	ces	48
Architectur	al and Engineering Services	52
Accounting	and Auditing Services	57
Education S	Services	60
Manageme	ent Consulting	64
Advertising	Services	68
Special Topic:	Trade in Research and Development Services	73
Industry Ov	/erview	73
Cross-Bord	er Trade of R&D Services	75
Sales and P	Purchases by Foreign R&D Services Affiliates	77
Tax Incentiv	ves and R&D Trade and Investment	79
Effects of P	rofit Shifting on Services Trade	80
Outlook		82
Bibliography .		83
Chapter 3	Professional Services Suppliers: Integrating Artificial Int	elligence
into Supply N	lodels to Lower Costs and Address Skills Gaps	93
Legal Services	s: U.S. Law Firms Remain Competitive Globally Although Market Con	ditions and

AI Adoption and Impacts	. 94
Declining U.S. Law Firm Market Presence and Exports of Legal Services to China Reflect	
Difficult Market and Regulatory Conditions	. 98
Outlook	100
Architectural and Engineering Services: Rising Costs and Technology Are Affecting Supplier 100	S
Labor Costs and Attracting Talent Challenge Suppliers	101
Architectural and Engineering Services Suppliers Increasingly Adopt High-Tech	
Solutions	102
Outlook	106
Accounting and Audit Services: Global Firms Are Leveraging New Technologies and	
Outsourcing to Maintain Competitiveness	106
Global Accounting and Audit Firms Seek Ways to Increase Labor Productivity and	
Lower Costs	107
U.S. Accounting Firms Are Outsourcing and Offshoring	109
Global Accounting Firms Continue to Digitalize Their Operations	109
Global Accounting Firms Continue to Implement AI While Exploring GenAI Use Cases	112
Outlook	116
Bibliography	117
Chapter 4 Demand for Professional Services: Changing Demographics,	
ncome Growth, and Continued Digitalization of Businesses	129
Education Services: U.S. Exports Are Delivered Through Multiple Modes	130
U.S. Exports of Education Services Driven by Strong Enrollments of International	
Students	131
Increases in Online Education Help Offset Declining Enrollments	133
International Branch Campuses	136
Outlook	138
Management Consulting Services: Rising Demand, Shifting Priorities	138
Strong Demand for Management Consulting Services Has Been Accompanied by	
Heightened Competition	
	139
Shifting Demand Has Brought Changes in Business Models	139 141
Shifting Demand Has Brought Changes in Business Models Digitalization and AI Drive Consulting Demand	139 141 142
Shifting Demand Has Brought Changes in Business Models Digitalization and AI Drive Consulting Demand Supply Chain Optimization and Sustainability Initiatives Further Support Growth	139 141 142 143
Shifting Demand Has Brought Changes in Business Models Digitalization and AI Drive Consulting Demand Supply Chain Optimization and Sustainability Initiatives Further Support Growth Outlook	139 141 142 143 144
Shifting Demand Has Brought Changes in Business Models Digitalization and AI Drive Consulting Demand Supply Chain Optimization and Sustainability Initiatives Further Support Growth Outlook Advertising Services: A Shift from Traditional to Digital Media	139 141 142 143 144 144
Shifting Demand Has Brought Changes in Business Models Digitalization and AI Drive Consulting Demand Supply Chain Optimization and Sustainability Initiatives Further Support Growth Outlook Advertising Services: A Shift from Traditional to Digital Media The Decline of Linear Television and Rise of Digital Television and Video Streaming Services	 139 141 142 143 144 144 145

Adoption of	of Artificial Intelligence (AI) in Advertising	150
Outlook		151
Bibliography		153
Chapter 5	Roundtable Summary	163
Persistent La	bor Shortages Reported in Services Industries	163
Education an	d Development Initiatives	
Characteristi	cs of the U.S. Services Workforce	
Role of Forei	gn Nationals in the U.S. Services Workforce	165
The Role of A	I in Reshaping Services	165
AI and Work	orce Adaptation	166
Challenges a	nd Opportunities in AI Implementation	166
Appendix A S	selected Services-Related Commission Publications and Staf	f
Research		169
Appendix B C	Corresponding Tables: Corresponding Data Tables for Figures	s 173
Appendix C C	Categories by BEA Industries	187

Boxes

Box 1.1 Modes of Supply for Services Trade Under the World Trade Organization's General	I Agreement on
Trade in Services	20
Box 2.1 Management Consulting and Bureau of Economic Analysis (BEA) Trade Data	68

Figures

Figure 1.1 Modes of supply and demand in U.S. services trade	22
Figure 1.2 Real value added by U.S. industry, 2019–23	26
Figure 1.3 Global services: cross-border exports of commercial services, by country, 2023	27
Figure 1.4 Global services: cross-border imports of commercial services, by country, 2023	28
Figure 1.5 U.S. private services: cross-border exports and imports, 2019–23	29
Figure 1.6 U.S. private services: cross-border exports, by category, 2023	
Figure 1.7 U.S. private services: cross-border imports, by category, 2023	31
Figure 1.8 U.S. private services: cross-border exports, by country, 2023	32
Figure 1.9 U.S. private services: cross-border imports, by country, 2023	33
Figure 1.10 U.S. services: affiliate sales and purchases, 2018–22	35
Figure 1.11 U.S. services: affiliate sales by U.Sowned foreign affiliates, by industry, 2022	36
Figure 1.12 U.S. services: purchases from foreign-owned U.S. affiliates, by industry, 2022	37
Figure 2.1 Professional services: U.S. cross-border exports and imports, 2019–23	42
Figure 2.2 Professional services: U.S. cross-border exports, by country, 2023	43
Figure 2.3 Professional services: U.S. cross-border imports, by country, 2023	44
Figure 2.4 Professional services: U.S. cross-border exports, by industry, 2023	45
Figure 2.5 Professional services: U.S. cross-border imports, by industry, 2023	46
Figure 2.6 Professional services: sales of U.Sowned foreign affiliates, by industry, 2022	47

Figure 2.7 Professional services: purchases from foreign-owned U.S. affiliates, by industry, 2022	48
Figure 2.8 Legal services: U.S. cross-border exports and imports, 2019–23	51
Figure 2.9 Legal services: U.S. affiliate sales and purchases, 2018–22	52
Figure 2.10 Architectural, engineering, scientific, and other technical services: U.S. cross-border expo	orts
and imports, 2019–23	55
Figure 2.11 Architectural, engineering, and other technical services: U.S. affiliate sales and purchases	,
2018–22	56
Figure 2.12 Accounting and auditing services: U.S. cross-border exports and imports, 2019–23	59
Figure 2.13 Accounting and auditing services: U.S. affiliate sales and purchases, 2018–22	60
Figure 2.14 Education services: U.S. cross-border exports and imports, 2019–23	63
Figure 2.15 Educational services: U.S. affiliate sales and purchases, 2018–22	64
Figure 2.16 Management consulting services: U.S. cross-border exports and imports, 2019–23	66
Figure 2.17 Management, scientific, and technical consulting services: U.S. affiliate sales and purchas	es,
2018–22	67
Figure 2.18 Advertising and related services: U.S. cross-border exports and imports, 2019–23	71
Figure 2.19 Advertising and related services: U.S. affiliate sales and purchases, 2018–22	72
Figure 2.20 Business enterprise research and development (BERD) spending in the top five countries,	,
2017 and 2021	74
Figure 2.21 R&D services: U.S. cross-border exports and imports, 2019–23	76
Figure 2.22 R&D services: U.S. cross-border exports to selected countries and imports from selected	
countries, 2023	77
Figure 2.23 R&D services: U.S. affiliate sales and purchases, 2018–22	78
Figure 3.1 Total number of accountants and auditors in the United States and total revenue for	
accounting services in the United States, 2019–23	. 108
Figure 4.1 Share of total U.S. social media ad spending, by platform, 2022–24	. 148

Tables

Table 1.1 Total U.S. private cross-border services exports (preliminary), by category, 2023–24	34
Table 1.2 Total U.S. private cross-border services imports (preliminary), by category, 2023–24	34
Table 2.1 Top five global legal firms ranked by revenue, fiscal year 2023	50
Table 2.2 Top five global design firms by revenue, 2023	54
Table 2.3 Top five global accounting networks by revenue, 2024	58
Table 2.4 Top five global advertising providers by advertising revenue, 2022	70
Table 2.5 Top 10 firms by global R&D spending, 2022	75
Table 2.6 Research and Development (R&D) statistics, 2021, selected countries	79
Table B.1 Real value added by U.S. industry, 2019–23	174
Table B.2 Global services: cross-border exports of commercial services, by country, 2023	174
Table B.3 Global services: cross-border imports of commercial services, by country, 2023	174
Table B.4 U.S. private services: cross-border exports and imports, 2019–23	. 175
Table B.5 U.S. private services: cross-border exports, by category, 2023	. 175
Table B.6 U.S. private services: cross-border imports, by category, 2023	. 175
Table B.7 U.S. private services: cross-border exports, by country, 2023	. 175
Table B.8 U.S. private services: cross-border imports, by country, 2023	. 176
Table B.9 U.S. services: affiliate sales and purchases, 2018–22	. 176
Table B.10 U.S. services: affiliate sales by U.Sowned foreign affiliates by industry, 2022	. 176
Table B.11 U.S. services: purchases from foreign-owned U.S. affiliates, 2022	177

Table B.12 Professional services: cross-border exports and imports, 2019–23	177
Table B.13 Professional services: U.S. cross-border exports, by country, 2023	177
Table B.14 Professional services: U.S. cross-border imports, by country, 2023	178
Table B.15 Professional services: U.S. cross-border exports, by service industry, 2023	178
Table B.16 Professional services: U.S. cross-border imports, by industry, 2023	179
Table B.17 Professional services: sales by U.Sowned foreign affiliates, by industry, 2022	179
Table B.18 Professional services: purchases from foreign-owned U.S. affiliates, 2022	179
Table B.19 Legal services: U.S. cross-border exports and imports, 2019–23	180
Table B.20 Legal services: U.S. affiliate sales and purchases, 2018–22	180
Table B.21 Architectural, engineering, scientific, and other technical services: U.S. cross-border e	xports
and imports, 2019–23	180
Table B.22 Architectural, engineering and related services: U.S. affiliate sales and purchases, 201	3–22181
Table B.23 Accounting and auditing services: U.S. cross-border exports and imports, 2019–23	181
Table B.24 Accounting and auditing services: U.S. affiliate sales and purchases, 2018–22	181
Table B.25 Education services: U.S. cross-border exports and imports, 2019–23	181
Table B.26 Education services: U.S. affiliate sales and purchases, 2018–22	182
Table B.27 Management consulting services: U.S. cross-border exports and imports, 2019–23	182
Table B.28 Management, scientific, and technical consulting: U.S. affiliate purchases, 2018–22	182
Table B.29 Advertising services: U.S. cross-border exports and imports, 2019–23	183
Table B.30 Advertising and related services: U.S. affiliate sales and purchases, 2018–22	183
Table B.31 Business enterprise research and development (BERD) spending in the top five countr	ies,
2017 and 2021	183
Table B.32 R&D services: U.S. cross-border exports and imports, 2019–23	183
Table B.33 R&D services: U.S. cross-border exports and imports, select countries, 2023	184
Table B.34 R&D services: U.S. affiliate sales and purchases, 2018–22	184
Table B.35 Total number of accountants and auditors in the United States and total revenue for	
accounting services in the United States, 2019–23	184
Table B.36 Share of total U.S. social media ad spending, by platform, 2022–24	185
Table C.1 Cross-border services trade by sector or industry	188
Table C.2 Professional services and related travel services cross-border trade by subsector or inde	ustry for
chapter 2	189
Table C.3 Affiliate services trade by sector or industry	190

Abbreviations and Acronyms

Terms	Definitions
AI	artificial intelligence
ALSP	alternative legal services provider
BCG	Boston Consulting Group
BE-120	Benchmark Survey of Transactions in Selected Services and Intellectual Property
	with Foreign Persons
BEA	Bureau of Economic Analysis (USDOC)
BERD	business enterprise research and development
BIM	building information modeling
BLS	Bureau of Labor Statistics (USDOL)
BPM6	Balance of Payments and International Investment Position Manual, Sixth Edition
COVID-19	coronavirus disease 2019
CPA	certified public accountant
EY	Ernst & Young
FAST	free advertising-supported streaming television
FTE	full-time equivalent
GATS	General Agreement on Trade in Services
GDP	gross domestic product
GenAl	generative artificial intelligence
IBC	international branch campus
IMF	International Monetary Fund
IP	intellectual property
IRA	Inflation Reduction Act
IT	information technology
ML	machine learning
MNE	multinational enterprise
NAICS	North American Industry Classification System
NYU	New York University
PwC	PricewaterhouseCoopers
R&D	research and development
RRF	Recovery and Resilience Facility
RPA	robot process automation
STEM	science, technology, engineering, and mathematics
STRI	Services Trade Restrictiveness Index
SVOD	subscription video-on-demand
tech	technology
UAE	United Arab Emirates
UK	United Kingdom
USDOC	U.S. Department of Commerce
USDOL	U.S. Department of Labor
USITC	U.S. International Trade Commission

Terms	Definitions
USTR	Office of the U.S. Trade Representative
WTO	World Trade Organization

Executive Summary

This year's report examines recent trends in U.S. services trade and competitive conditions in the global services market in chapter 1, with a focus on professional services starting in chapter 2. It describes recent developments in the U.S. and global markets for major professional services sectors, including legal, architectural and engineering, accounting and auditing, education, management consulting, advertising, and research and development (R&D) services. Important emerging trends in global trade for these professional services are described in chapters that highlight two overarching themes. Chapter 3 illustrates a supply-side theme that is relevant for many professional services: how skills gaps and the introduction of artificial intelligence (AI) technologies are affecting the competitiveness of professional services suppliers. Chapter 4 focuses on how changing demographics (including aging and income growth) in advanced and emerging markets, businesses' ongoing digitalization, and the move by many manufacturing firms to reorganize their supply chains in the wake of the coronavirus disease 2019 (COVID-19) pandemic are driving demand for certain professional services.

Trade in services falls into two categories: cross-border transactions, and transactions with U.S.-owned or foreign-owned affiliates. In 2023, the United States continued to be the world's largest exporter and importer of services, with cross-border services exports totaling \$993.8 billion (or 12.7 percent of global services exports) and imports totaling \$722.7 billion (or 10.0 percent of global services imports). Sales of services by foreign affiliates of U.S.-owned firms totaled \$2.1 trillion in 2022 (the latest year available); purchases of services from U.S. affiliates of foreign-owned firms totaled \$1.5 trillion.

Report Highlights

The United States Continues to Run a Trade Surplus in Both Cross-Border Services Trade and Foreign Affiliate Sales

In 2023, U.S. cross-border services exports exceeded imports, resulting in a trade surplus of \$271.1 billion. The largest U.S. cross-border trade surpluses were in the professional services and financial services categories. The largest market for U.S. services exports in 2023 was the United Kingdom (UK), followed by Canada, Ireland, Switzerland, and UK overseas territories (including the British Virgin Islands, the Cayman Islands, Montserrat, and the Turks and Caicos Islands). In that same year, the largest source of U.S. services imports was the UK, followed by Canada, Germany, Mexico, and India.

Similarly, in 2022, sales of services by U.S.-owned foreign affiliates exceeded purchases from foreignowned U.S. affiliates by \$596.3 billion. Overall, the UK was the largest market for services sales by U.S.owned foreign affiliates, followed by Ireland and Canada. Affiliates of German-owned firms accounted for the largest share of services purchased from all foreign-owned affiliates in the United States, followed by U.S. affiliates owned by Canadian firms and UK firms.

Professional Services Accounted for 38 Percent of U.S. Cross-Border Services Exports and 29 Percent of Imports in 2023

Professional services accounted for \$378.2 billion of U.S. cross-border services exports in 2023, or 38.1 percent of total U.S. services exports, and \$184.7 billion, or 29.3 percent, of imports, resulting in a cross-border surplus of \$193.5 billion. The top markets for U.S. cross-border exports of professional services were Ireland, Switzerland, and Canada. The top sources of imports were the UK, India, and Canada. In 2022, foreign professional services affiliates of U.S.-owned companies recorded services sales of \$176.5 billion, accounting for 8.3 percent of total U.S. affiliate sales of services. Purchases of services from U.S. professional services affiliates of foreign-owned companies totaled \$171.2 billion, or 11.3 percent of total U.S. affiliate purchases of services.

Tax-Advantaged Markets Attract R&D Services Spending and Trade

Trends in international trade in R&D services reflect the geographic pattern of spending on both basic and applied R&D by firms in R&D-intensive sectors such as technology, health care, and manufacturing, as well as the impact of firms' global tax strategies. The largest destinations for U.S. R&D services exports are similar to, but not exactly aligned with, countries where total domestic R&D spending is high and where U.S.-owned foreign affiliates spend the most on R&D activities.² Varying taxation levels for R&D and intellectual property assets among the major destinations for U.S. exports is the other important driver of R&D services trade flows. Global R&D spending was \$2.5 trillion in 2022, led by the United States, China, and Japan.

In 2022, Amazon, Meta, and Alphabet were the firms with the most global R&D spending (\$73.2 billion, \$35.3 billion, and \$28.8 billion, respectively). U.S. cross-border exports of R&D services totaled \$118.9 billion in 2023; imports totaled \$57.3 billion. R&D services affiliate trade (foreign affiliate sales and purchases) is typically much smaller than cross-border trade. In 2022, sales by foreign R&D services affiliates of U.S.-owned firms totaled \$12.4 billion and purchases from U.S. R&D affiliates of foreign-owned firms totaled \$6.7 billion. The top destinations for U.S. R&D exports, like Ireland and Switzerland, have effective tax rates that encourage local R&D expenditures and the use of domestic and imported intellectual property.

² Statistics on foreign affiliate spending on R&D capture both spending on R&D services imports (including from the United States) and domestic spending on R&D in the country where the affiliate is located.

Professional Services Firms Are Integrating AI into Their Supply Models to Lower Costs and Address Skills Gaps

In recent years, professional services suppliers have moved to adopt new technologies to improve productivity, lower costs, and address increased constraints in skilled labor supply in their sectors. This has led to large changes in business models in some sectors. For example, the proliferation of legal technology has altered the legal services industry landscape through the growth of lower-priced and more technically sophisticated alternative legal services providers, legal technology companies, and client use of legal technology. In turn, law firms have increasingly adopted legal technology, and the use of Al tools specifically by law firms has gained momentum. U.S. firms supplying legal services in foreign markets are at the forefront of using technology integration to increase efficiency and maintain or enhance their competitive positions. At the same time, they are facing increasingly complex regulatory environments in certain countries, which makes it difficult to be competitive in those markets.

Architectural and engineering services are labor intensive, and in recent years, firms in this industry have been affected by higher labor costs and inflation. Architectural and engineering firms have also had difficulties in hiring and retaining skilled workers, and in response some are offering benefits like mentorship and internship opportunities, as well as flexible work schedules. New technologies like AI, building information modeling, and drones are increasingly used for design, risk assessment, 3D modeling, and other applications. One recent survey found that almost 50 percent of architectural and engineering firms are now using AI. This may reduce labor costs and increase productivity, though it also may require extensive upskilling and raise issues like plagiarism.

Global accounting firms are maintaining competitiveness by employing new technologies and outsourcing or offshoring (relocating services overseas) to improve productivity and lower costs. Accounting firms are automating and importing lower-skilled tax and auditing functions to lower the cost of supplying these services where revenue growth is likely to be limited. At the same time, they are seeking to drive revenue growth through higher-value advisory services. For example, in the United States, accounting services firms' revenue has steadily increased over the past five years, despite reductions in staffing due to resignations and retirements among other factors. Firms in this sector continue to leverage digital tools like robot process automation, cloud computing, and data analytics to improve efficiency. Accounting and auditing firms recognize that the implementation of AI and generative AI (GenAI) technologies will likely be a competitiveness factor going forward in part because these tools can accelerate the automation of lower-value tax and auditing functions.

Changing Demographics and Digitalization Affect Demand for Professional Services

Demand for professional services reflects changing demographics, including aging populations and rising incomes, and ever-expanding digitalization in the United States and foreign markets in recent years. For example, international students have been a growing constituency for U.S. colleges and

universities over the past five years, with particularly large numbers of students from India and China studying in California and New York. At the same time, the enrollment of U.S. students in colleges and universities, which increased steadily from 1980 to 2010, has started to decline, with 2 million fewer students in 2022–23 compared to 2010–11. Some institutions are responding by both cutting costs and trying to attract more international students. Digital provision of education services, including cross-border, has been growing in recent years. The use of massive open online courses increased dramatically in March 2020 because of COVID-19 pandemic-related lockdowns and prompted instructors and institutions to quickly acquire and learn digital technologies. Following the resumption of in-person instruction, some universities started to offer combinations of synchronous (real-time) and asynchronous (self-paced) classes. International branch campuses have also expanded recently, especially in China and the United Arab Emirates (UAE), though some have struggled to attract students or funding and have been forced to close.

Clients increasingly want management consulting services that incorporate AI and other new technologies, which has driven traditional strategy consulting firms to shift from the previous focus on human capital to investing significantly in building expertise in technology areas. Moreover, large multinational enterprises like MasterCard and Pfizer have launched "insourced" consulting practices with expanded internal capacities, further shaping demand for the services of traditional management consulting companies. At the same time, established firms providing management consulting services have been affected by rising competition from information technology firms and small niche companies, as well as the emergence of freelance consulting platforms. Changes in demand have also motivated management consulting firms to offer a broader range of tailored and customized services and to move toward value-based pricing instead of hour-based pricing.

The advertising services industry has been experiencing a sharp decline in demand for linear television content and a sharp rise in demand for video streaming services content. This industry also continues to see growing demand from the increased use of social media platforms, as indicated by growth in subscriber bases and digital advertising spending. Digital advertising firms are increasingly using AI technologies to target consumers more effectively and create relevant content more quickly.

Finally, the Commission held its annual Services Roundtable on October 30, 2024. Participants discussed how widespread workforce gaps and aging demographics are affecting U.S. services, how some graduates of U.S. schools and universities are struggling to find employment within the first two years, and how AI is expected to augment or replace many tasks in professional services.

Chapter 1 Introduction

This report, *Recent Trends in U.S. Services Trade (Recent Trends)*, published annually by the U.S. International Trade Commission (Commission or USITC), examines trends in U.S. services trade, the global services market, and competitive conditions, both in the aggregate and in selected industries. In addition, each year *Recent Trends* focuses on a specific category of services.³ This year, the report focuses on professional services, including accounting and audit services, advertising services, architectural and engineering services, education services, legal services, management consulting services, and research and development (R&D) services. Professional services were last covered in the 2021 report.⁴

This report is organized into five chapters. Chapter 1 gives an overview of the domestic U.S. services sector, global cross-border trade in services, and U.S. services trade (both cross-border and affiliate transactions) by services sector.⁵ It also provides a description of the data sources used throughout the report. Chapter 2 offers an overview of trends in cross-border trade, as well as foreign affiliate sales and purchases in the professional services category. Chapter 2 also looks in detail at the data and drivers of trade trends in R&D services.

Chapters 3 and 4 identify important and emerging trends in selected subsectors of professional services. Chapter 3 focuses on how skills gaps, job openings and employment limits, and the introduction of new artificial intelligence (AI) technologies are creating both disruptions and opportunities that affect the competitiveness of suppliers of professional services. Chapter 4 focuses on the effects of changing demographics, including aging and income growth, in many advanced and emerging markets; businesses' ongoing digitalization; and the move by many manufacturing firms to reorganize their supply chains in the wake of the global pandemic on the demand for and patterns of trade in professional services. Finally, chapter 5 summarizes the views expressed by participants at the Commission's 18th annual Services Roundtable, held on October 30, 2024.

The report also includes three appendixes. Appendix A summarizes recent services-related Commission publications and staff research. Appendix B presents underlying data for the figures included in this report. Appendix C provides information on how the Commission categorizes services data from the Bureau of Economic Analysis (BEA) at the U.S. Department of Commerce (USDOC). Additionally,

³ Beginning in 2013, *Recent Trends* has rotated its coverage on an annual basis among these four services industry categories.

⁴ USITC, *Recent Trends in U.S. Services Trade: 2021 Annual Report*, April 2021. Three other services categories are usually covered in a four-year rotation: digital and electronic services, distribution services, and financial services (last covered in 2022, 2023, and 2024, respectively). For a comprehensive listing of the services that comprise the broader service categories of professional services, digital and electronic services, distribution services, and financial services, and financial services, and financial services.

⁵ "Cross-border" refers to U.S. exports and imports of services; "affiliate transactions" refers to services supplied and received by multinational enterprises through their majority-owned affiliates; see box 1.1 below.

interactive figures and tables associated with this report are available on the Commission's website, allowing users to explore U.S. services trade trends over time and for selected industries and countries.⁶

Data Sources, Categories, and Limitations

Data on trade in services are more limited than data on trade in goods, partly because of the lack of U.S. customs data on services trade. As a result, this report relies on a variety of sources to present a comprehensive picture of trade in services. Official U.S. services trade data used in this report come from the BEA, which publishes annual data on both cross-border trade and affiliate transactions. These annual data are produced from a combination of surveys of U.S. firms; other administrative data sources, such as financial reports from the U.S. Department of State; and data from non-U.S. government sources.⁷ Cross-border trade and foreign affiliate transactions cover a vast majority of total U.S. services trade via all four modes of supply specified in the World Trade Organization's (WTO's) General Agreement on Trade in Services (GATS).⁸ Box 1.1 explains the four modes of supply for services trade.

Box 1.1 Modes of Supply for Services Trade Under the World Trade Organization's General Agreement on Trade in Services

The General Agreement on Trade in Services identifies four modes of supply for services trade, or four ways in which trade in services is most commonly measured.^a

Mode 1 is cross-border supply. In this mode, an individual or firm in one country supplies a service to an individual or firm in another country (i.e., the service crosses national borders). An example is a firm emailing (i.e., exporting) a digital file of an architectural design to a foreign client. This mode also captures transportation services, such as a ship delivering goods from a port in one country to a port in another country.

Mode 2 is consumption abroad. In this mode, an individual from one country travels to another country and consumes a service in that country. For example, a U.S. export of tourism service occurs when a foreign tourist purchases souvenirs, food, and lodging while vacationing in the United States.

Mode 3 is commercial presence. In this mode, a firm based in one country establishes a local affiliate in another country and supplies services through that affiliate. An example is a U.S.-based retailer establishing a local affiliate in a foreign country for the purpose of selling products to consumers in that market as well as consumers in other countries.

Mode 4 is the temporary presence of natural persons. In this mode, an individual from one country travels short term to another country to supply a service—for instance, as a consultant, contract

⁶ Interactive figures can be found at:

https://www.usitc.gov/publications/industry econ analysis 332/2025/recent trends us services trade 2025 an nual report.

⁷ For a list of the data the BEA uses to compile trade in services statistics, see USDOC, BEA, "U.S. International Economic Accounts: Concepts and Methods," June 2024, 17–19.

⁸ Annex 1B, GATS, Art. 1(1–2) (April 15, 1994); WTO, "Basic Purpose and Concepts," accessed September 26, 2024.

employee, or intracompany transferee at a foreign affiliate. An example is a U.S.-based airplane technician traveling abroad to service or repair an aircraft.

^a Annex 1B, GATS, Art. 1(1–2) (April 15, 1994); WTO, "Basic Purpose and Concepts" accessed September 26, 2024.

Figure 1.1 summarizes how U.S. services are exported and imported via these four modes of supply, as well as how the modes are differentiated in BEA data.⁹

⁹ After income generated through affiliate transactions has been repatriated to the United States, it appears as direct investment income in the U.S. balance of payments. USDOC, BEA, *U.S. International Economic Accounts*, June 2024, 123.

Figure 1.1 Modes of supply and demand in U.S. services trade

How services are supplied by the United States



How services are received by the United States



Source: USDOC, BEA, "The Basics of How International Services Are Supplied and Received," October 19, 2022.

Note: A U.S. multinational enterprise that owns a foreign affiliate can provide services to customers in that country and customers in other countries. Mode 2 services involve consumers traveling abroad, whereas Modes 1, 3, and 4 involve services suppliers traveling or operating abroad.

As defined by the BEA, cross-border services trade occurs when suppliers in one country sell services to consumers in another country, with people, information, or money crossing national borders. GATS

mode 1 and mode 2 transactions, as well as some mode 4 transactions, are generally grouped together in BEA data on cross-border trade. Firms also provide services to foreign consumers through affiliates established in host (i.e., foreign) countries,¹⁰ and after income generated through affiliate transactions has been repatriated to the United States, it appears as direct investment income in the U.S. balance of payments.¹¹ These mode 3 transactions are included, with some exceptions, in BEA affiliate transactions data.¹²

This report focuses on BEA "private services" data.¹³ As a result, the export and import data presented throughout this report exclude government transactions, which primarily consist of services supplied in support of operations of the U.S. military and U.S. embassies in foreign countries.

Aggregated data on cross-border trade in services appear in the balance of payments statistics published quarterly for the United States by the BEA and annually in the WTO's global services trade data.¹⁴ The term "commercial services" used in WTO services trade data is roughly equivalent to the term "private services" used in BEA services trade data.¹⁵ Like BEA cross-border trade data, WTO cross-border trade data roughly correspond to modes 1, 2, and 4, as specified in GATS.¹⁶

The BEA also uses survey data to publish more detailed annual statistics for cross-border services trade and foreign affiliate transactions of the United States. These data are sorted by partner country and by industry at the finest level of detail that BEA survey and confidentiality policies allow.¹⁷ Statistics on cross-border trade and foreign affiliate transactions are available for several subcategories of professional services: advertising, architectural and engineering, accounting and audit, education, legal, management consulting, R&D, and waste treatment services. More information on the data coverage for professional services is available in chapter 2.

Beginning in 2009, the BEA began efforts to restructure its international trade data following the release by the International Monetary Fund (IMF) of *Balance of Payments and International Investment Position*

¹⁰ USDOC, BEA, U.S. International Economic Accounts, June 2024, 264.

¹¹ USDOC, BEA, U.S. International Economic Accounts, June 2024, 123.

¹² USDOC, BEA, *U.S. International Economic Accounts*, June 2024, 265. Some statistics on services supplied through mode 4 may also be commingled with statistics on compensation of employees. Similarly, services supplied by employees that have been temporarily transferred to foreign affiliates may appear in foreign affiliate sales rather than cross-border statistics. The channel of delivery that service providers use is determined primarily by the nature of the service. For example, retail services are generally supplied through affiliates but transportation services are generally supplied through cross-border trade.

¹³ In BEA data, "private services" include all (total) services excluding the category government goods and services, not included elsewhere.

¹⁴ WTO, "Trade in Commercial Services," accessed September 26, 2024; USDOC, BEA, table 1.1, "U.S. International Transactions," September 19, 2024.

¹⁵ BEA private services are the same as the WTO's commercial services in aggregate. The WTO, however, defines services according to the fourth (1977) and fifth (1993) editions of the IMF's *Balance of Payments and International Investment Position Manual*; the BEA defines them according to the sixth version (2009) of this manual. WTO, "Technical Notes," accessed September 26, 2024; USDOC, BEA, *U.S. International Economic Accounts*, June 2024. ¹⁶ WTO, *World Trade Statistical Review 2023*, 2023.

¹⁷ Data are suppressed for certain industries or sectors for which disclosure could potentially reveal confidential information about individual companies. USDOC, BEA, *U.S. International Economic Accounts*, June 2024, 24, 248–49.

Manual, Sixth Edition (BPM6), which provides a statistical framework on balance of payments and international investment data. The BEA implemented changes to gradually bring U.S. services trade statistics in line with the BPM6, expanding detail in its international trade in services statistics starting in 2014 and continuing through 2022.¹⁸ In 2022, the BEA updated its *Benchmark Survey of Transactions in Selected Services and Intellectual Property with Foreign Persons* (BE-120 survey) covering selected services and intellectual property with foreign persons.¹⁹

The BEA collects and publishes survey-based statistics in two different ways. For cross-border services trade, the BEA collects data based on the type of service traded. For services supplied through affiliates, the BEA collects data based on the affiliates' primary industry.²⁰ As a result, comparability at the sector level between statistics for cross-border trade and foreign affiliate sales is limited. For example, a firm like Deloitte would likely report cross-border trade data for different segments of its business, such as legal and management consulting services, but will typically be classified in its affiliate transactions as an "accounting, auditing, and bookkeeping" service provider.²¹ As a result, any foreign affiliate sales of its legal services could appear under the accounting, auditing, and bookkeeping services category, rather than the legal services category, in BEA affiliate transactions data.²²

This report uses the latest available services trade data for each source described above. As of the time this report was prepared, WTO cross-border trade in services data were available through 2023. Annual data on cross-border trade from the BEA were available through 2023 (with preliminary data available for 2024); however, BEA data on affiliate transactions were available through only 2022. Accordingly, data on market conditions for each specific industry covered in this report may span different years.

U.S. Services Sector

The services sector is the largest sector of the U.S. economy, and the United States is the world's top cross-border exporter and importer of services.²³ In real value-added terms, U.S. private service-

¹⁸ USDOC, BEA, U.S. International Economic Accounts, June 2024, 10–15.

¹⁹ USDOC, BEA, 2022 Benchmark Survey of Transactions, August 2022. The BEA conducts Benchmark Surveys (BE-10 and BE-12) every five years to collect information from all services trading companies.

²⁰ USDOC, BEA, U.S. International Economic Accounts, June 2024, 253.

²¹ Deloitte, "Corporate Law," accessed February 3, 2025; "Accounting and Reporting Services," accessed March 27, 2025; "Global Office Locations," accessed March 27, 2025.

²² Affiliate transactions are reported by industry rather than type of product, and the affiliate transaction data include ancillary services provided by firms in the agriculture, manufacturing, and mining industries. USDOC, BEA, *U.S. International Economic Accounts*, June 2024, 242.

²³ WTO, "Trade in Commercial Services," accessed September 26, 2024.

supplying industries contributed \$16.5 trillion, or 82.3 percent, to U.S. gross domestic product (GDP) in 2023.²⁴ By comparison, goods-producing industries contributed \$3.7 trillion, or 18.3 percent.

Services-supplying industries also supplied the majority of full-time equivalent (FTE) employees in the U.S. economy in 2023, with private services industries encompassing 102.7 million FTE employees, or 70.3 percent of total employees.²⁵ Goods-producing industries, by contrast, accounted for private employment of 22.3 million FTE employees, or 15.3 percent of total employees.

Between 2019 and 2023, real value added by private U.S. services-supplying industries increased by 13.0 percent from \$14.6 trillion to \$16.4 trillion (figure 1.2). In contrast, real value added by private goods-producing industries increased by 2.1 percent, from \$3.6 trillion to \$3.7 trillion. The number of FTE employees in U.S. services-supplying industries increased by 4.2 percent from 2019 through 2023, whereas employment in goods-producing industries increased by 2.1 percent.²⁶

²⁴ USDOC, BEA, "Real Value Added by Industry," September 26, 2024; USDOC, BEA, table 6.5D, "Full-Time Equivalent Employees by Industry," September 27, 2024. Real value added is a measure of an industry's contribution to GDP and is the difference between the value of an industry's gross output and the cost of intermediate inputs, adjusted for inflation. Services-supplying industries include utilities; wholesale trade; retail trade; transportation and warehousing; information; finance, insurance, real estate, rental, and leasing; professional and business services; educational services, health care, and social assistance; and arts, entertainment, recreation, accommodation, and food services. Goods-producing industries include agriculture, forestry, fishing, and hunting; mining; construction; and manufacturing. These percentages are based on the private sector only; for totals including the U.S. government, U.S. services represented 72.9 percent of total value added and goods represented 16.2 percent. Because of rounding, these numbers may not add to 100 percent.
²⁵ USDOC, BEA, table 6.5D, "Full-Time Equivalent Employees by Industry," September 27, 2024.





In trillions of dollars. Underlying data for this figure can be found in appendix B, table B.1.

Source: USDOC, BEA, "Real Value Added by Industry," September 26, 2024.

Notes: Estimates are inflation-adjusted 2017 dollars. Private goods-producing industries include agriculture, forestry, fishing, and hunting; mining; construction; and manufacturing. Private services-producing industries include private utilities; wholesale trade; retail trade; transportation and warehousing; information; finance, insurance, real estate, rental, and leasing; professional and business services; educational services, health care, and social assistance; arts, entertainment, recreation, accommodation, and food services; and other services, except government services.

Global Services Trade

The United States was the world's largest cross-border exporter of commercial services in 2023, supplying 12.7 percent of global services exports. Other leading services exporters were the United Kingdom (UK) (7.4 percent), Germany (5.5 percent), Ireland (5.1 percent), and China (4.9 percent) (figure 1.3).

Figure 1.3 Global services: cross-border exports of commercial services, by country, 2023

In percentages. Underlying data for this figure can be found in appendix B, table B.2.



Export total: \$7.8 trillion

Source: WTO, "Trade in Commercial Services," accessed September 26, 2024. Notes: Exports of commercial services exclude public sector transactions. Because of rounding, shares may not add to 100 percent.

The United States was also the largest global importer of services in 2023, representing 10.0 percent of total cross-border services imports. As shown in figure 1.4, other leading importers of commercial services included China (7.6 percent), Germany (6.9 percent), the UK (5.4 percent), and Ireland (5.4 percent).

Figure 1.4 Global services: cross-border imports of commercial services, by country, 2023

In percentages. Underlying data for this figure can be found in appendix B, table B.3.



Import total: \$7.2 trillion

Source: WTO, "Trade in Commercial Services," accessed September 26, 2024. Notes: Imports of commercial services exclude public sector transactions. Because of rounding, shares may not add to 100 percent.

U.S. Cross-Border Services Trade

In 2023, the United States had a cross-border trade surplus of \$271.1 billion and was the largest net exporter of commercial services by a wide margin.²⁷ In 2023, U.S. cross-border services exports increased by 8.1 percent from \$919.8 billion in 2022 to \$993.8 billion, continuing a rebound started in 2021 from a COVID-19 pandemic-related low of \$704.3 billion in 2020. U.S. imports of services increased by 5.0 percent from \$688.3 billion in 2022 to \$722.7 billion in 2023 (figure 1.5).

²⁷ USDOC, BEA, table 2.1, "U.S. Trade in Services, by Type of Service," July 3, 2024.



Figure 1.5 U.S. private services: cross-border exports and imports, 2019–23

Source: USDOC, BEA, table 2.1, "U.S. Trade in Services, by Type of Service," July 3, 2024.

In 2023, professional services continued to compose the largest share of U.S. cross-border services exports, while travel services composed the largest share of imports.²⁸ As shown in figure 1.6, crossborder exports of professional services during 2023 totaled \$378.2 billion, or 38.1 percent of total U.S. cross-border services exports, followed by financial services (\$200.4 billion; 20.2 percent) and travel services (\$174.8 billion; 17.6 percent).

²⁸ For a comprehensive listing of the services that comprise the broader service categories of professional services, digital and electronic services, distribution services, and financial services, see appendix C.

Figure 1.6 U.S. private services: cross-border exports, by category, 2023

In percentages. Underlying data for this figure can be found in appendix B, table B.5.



Export total: \$993.8 billion

Source: USDOC, BEA, table 2.1, "U.S. Trade in Services, by Type of Service," July 3, 2024. Note: Because of rounding, shares may not add to 100 percent.

U.S. cross-border imports of travel services accounted for \$211.8 billion, or 29.3 percent of total U.S. cross-border imports in 2023, followed by professional services (\$184.7 billion; 25.6 percent) and financial services (\$127.3 billion; 17.6 percent); see figure 1.7.





Source: USDOC, BEA, table 2.1, "U.S. Trade in Services, by Type of Service," July 3, 2024. Note: Because of rounding, shares may not add to 100 percent.

In 2023, all U.S. services categories except travel services recorded a cross-border trade surplus, with the largest surplus in professional services (\$193.5 billion), followed by financial services (\$73.1 billion) and digital and electronic services (\$33.2 billion).²⁹ The distribution services category registered a deficit in 2022 (\$28.1 billion) but a surplus in 2023 (\$4.3 billion).

In 2023, the UK was the largest recipient of U.S. cross-border private services exports, totaling \$90.4 billion, or 9.1 percent of total U.S. services exports.³⁰ Other leading recipients of U.S. services exports included Canada (\$85.4 billion; 8.6 percent), Ireland (\$84.3 billion; 8.5 percent), and Switzerland (\$49.6 billion; 5.0 percent); see figure 1.8.

²⁹ USDOC, BEA, table 2.1, "U.S. Trade in Services, by Type of Service," July 3, 2024.

³⁰ USDOC, BEA, table 2.2, "U.S. Trade in Services, by Type of Service and by Country or Affiliation," July 3, 2024.



Source: USDOC, BEA, table 2.2, "U.S. Trade in Services, by Type of Service and by Country or Affiliation," July 3, 2024. Notes: Exports of commercial services exclude public sector transactions. Because of rounding, shares may not add to 100 percent. The BEA category "United Kingdom Islands, Caribbean" includes the following UK overseas territories: the British Virgin Islands, the Cayman Islands, Montserrat, and the Turks and Caicos Islands.

In 2023, the largest source of U.S. private services imports was the UK (\$85.2 billion; 11.8 percent), followed by Canada (\$54.0 billion; 7.5 percent), Germany (\$45.8 billion; 6.3 percent), and Mexico (\$44.8 billion; 6.2 percent); see figure 1.9.



Source: USDOC, BEA, table 2.2, "U.S. Trade in Services, by Type of Service and by Country or Affiliation," July 3, 2024. Notes: Imports of commercial services exclude public sector transactions. Because of rounding, shares may not add to 100 percent.

Cross-Border Trade: Preliminary 2024 Data

In addition to the cross-border trade data presented above for 2023, preliminary cross-border services trade data are also available for 2024 using aggregated services categories. Preliminary data on U.S. cross-border services trade for 2024 show that total private services exports were 8.3 percent higher in 2024 compared to 2023 (table 1.1). During this period, the most notable increases were in travel and passenger fares (for which exports increased by 12.3 percent); telecommunications, computer, and information services (12.2 percent); and insurance services (10.6 percent). In contrast, personal, cultural, and recreational services saw a 14.8 percent decrease in exports between 2023 and 2024.³¹

³¹ USDOC, BEA, table 3.1, "U.S. International Trade in Services," March 20, 2025.

Services category	2023	2024	change (%)
Travel and passenger fares	225.9	253.6	12.3
Financial services	175.5	190.7	8.7
Professional and management consulting services	150.4	163.5	8.7
Telecommunications, computer, and information services	70.6	79.3	12.2
Research and development services	54.8	53.7	-1.9
Technical, trade-related, and other business services	48.0	50.2	4.7
Air transport (excludes passenger fares)	36.0	39.1	8.6
Personal, cultural, and recreational services	30.7	26.2	-14.8
Insurance services	25.0	27.6	10.6
Sea transport	19.8	20.5	3.8
Other services, n.i.e.	157.1	172.2	9.6
Total cross-border exports	993.8	1,076.6	8.3

Table 1.1 Total U.S. private cross-border services exports (preliminary), by category, 2023–24 In billions of dollars. n.i.e. = not included elsewhere.

Source: USDOC, BEA, table 3.1, "U.S. International Trade in Services," March 20, 2025.

Preliminary data on U.S. cross-border services imports in 2024 also show significant increases and were 8.8 percent higher in 2024 compared to 2023 (table 1.2). "Other services not included elsewhere" posted the largest growth in imports (20.6 percent), followed by insurance services (15.2 percent), air transport services (13.1 percent), and travel and passenger fares (10.8 percent). In contrast, financial services imports decreased slightly, by 2.5 percent.

Table 1.2 Total U.S. private cross-border services imports (preliminary), by category, 2023–24
In billions of dollars. n.i.e. = not included elsewhere.

Services category	2023	2024	change (%)
Travel and passenger fares	224.0	248.1	10.8
Financial services	62.7	61.1	-2.5
Professional and management consulting services	74.1	80.9	9.3
Telecommunications, computer, and information services	60.1	60.4	0.5
Research and development services	33.6	35.8	6.7
Technical, trade-related, and other business services	37.5	39.6	5.8
Air transport (excludes passenger fares)	28.9	32.7	13.1
Personal, cultural, and recreational services	33.0	34.9	5.7
Insurance services	64.6	74.4	15.2
Sea transport	44.4	46.4	4.6
Other services, n.i.e.	59.8	72.1	20.6
Total cross-border imports	722.7	786.6	8.8

Source: USDOC, BEA, table 3.1, "U.S. International Trade in Services," March 20, 2025.

Affiliate Transactions

Sales of services by the affiliates of U.S. companies abroad exceeded purchases within the United States from the affiliates of foreign companies in every year from 2018 through 2022, the last year for which

such data are available.³² Sales of U.S. affiliates abroad increased from \$1.7 trillion in 2018 to \$2.1 trillion in 2022 (figure 1.10). Similarly, purchases from the affiliates of foreign firms in the United States grew from \$1.2 trillion in 2018 to \$1.5 trillion in 2022. Like affiliate sales, affiliate purchases also declined slightly in 2020.³³



Figure 1.10 U.S. services: affiliate sales and purchases, 2018–22

Sources: USDOC, BEA, table 4.1, "Services Supplied to Foreign Persons by U.S. MNEs," October 8, 2024; USDOC, BEA, table 5.1, "Services Supplied to U.S. Persons by Foreign MNEs," October 8, 2024.

Notes: MNEs = multinational enterprises. Sales by U.S.-owned foreign affiliates include services supplied by majority-owned foreign affiliates of U.S. parent firms. Purchases from foreign-owned U.S. affiliates include services supplied by majority-owned U.S. affiliates of foreign parent firms.

In 2022, digital and electronic services accounted for the largest share of U.S.-owned foreign affiliates sales, amounting to \$637.4 billion, or 30.1 percent of total foreign affiliate sales, followed by distribution services (\$563.3 billion; 26.6 percent) and financial services (\$319.2 billion; 15.1 percent); see figure 1.11.³⁴ The top markets for the sales of services through U.S. affiliates abroad were the UK (\$346.6 billion), Ireland (\$227.9 billion), and Canada (\$164.3 billion).³⁵

³² For affiliate transactions, sales of services by U.S. affiliates abroad are treated as services exports and purchases from the affiliates of foreign firms in the United State are treated as services imports.

³³ USDOC, BEA, table 4.1, "Services Supplied to Foreign Persons by U.S. MNEs," October 8, 2024; USDOC, BEA, table 5.1, "Services Supplied to U.S. Persons by Foreign MNEs," October 8, 2024.

³⁴ For a comprehensive listing of the services that compose the broader service categories of financial services, professional services, digital and electronic services, and distribution services, see appendix C.

³⁵ USDOC, BEA, table 4.1, "Services Supplied to Foreign Persons by U.S. MNEs," October 8, 2024.

Figure 1.11 U.S. services: affiliate sales by U.S.-owned foreign affiliates, by industry, 2022

In percentages. Underlying data for this figure can be found in appendix B, table B.10.



Sales by U.S.-owned foreign affiliates total: \$2.1 trillion

Source: USDOC, BEA, table 4.1, "Services Supplied to Foreign Persons by U.S. MNEs," October 8, 2024. Notes: MNEs = multinational enterprises. Professional services are underreported because health care and social assistance services data are suppressed. "Manufacturing" includes ancillary services provided by goods manufacturers. Other services include services supplied by majority-owned foreign affiliates of U.S. parent firms. Distribution services, digital and electronic services, professional services, and all other services are underreported because of suppression of data. Because of rounding, shares may not add to 100 percent.

Digital and electronic services accounted for the largest share of purchases from foreign-owned affiliates in the United States in 2022, with such purchases totaling \$637.4 billion, or 30.1 percent of total foreign affiliate purchases. In that same year, distribution services represented \$563.3 billion (26.6 percent) of total affiliate purchases, whereas financial services totaled \$319.2 billion (15.1 percent); see figure 1.12. German companies with U.S.-based affiliates were the largest source of foreign affiliate purchases in 2022 (\$205.6 billion), followed by Canadian-owned affiliates (\$188.5 billion) and UK-owned affiliates (\$188.1 billion).³⁶

³⁶ USDOC, BEA, table 5.1, "Services Supplied to U.S. Persons by Foreign MNEs," October 8, 2024.
Figure 1.12 U.S. services: purchases from foreign-owned U.S. affiliates, by industry, 2022

In percentages. Underlying data for this figure can be found in appendix B, table B.11.



Purchases from foreign-owned U.S. affiliates total: \$1.5 trillion

Source: USDOC, BEA, table 5.1, "Services Supplied to U.S. Persons by Foreign MNEs," October 8, 2024. Notes: MNEs = multinational enterprises. Digital and electronic services and all other services are underreported because of the suppression of data. "Manufacturing" includes ancillary services provided by goods manufacturers. "All other services" includes ancillary services provided in the mining, agriculture, and other sectors, as well as suppressed data. Beginning with the 2018 *Recent Trends in U.S. Services Trade* report, software publishing was reallocated from "All other services" to "Digital and Electronic Services" to better reflect the industry composition. Therefore, digital and electronic services data in this report, and in all *Recent Trends* reports published in 2018 and later, cannot be directly compared with such data in USITC reports published before 2018. Because of rounding, shares may not add to 100 percent.

Bibliography

- Deloitte. "Accounting and Reporting Services." Accessed March 27, 2025. <u>https://www2.deloitte.com/us/en/pages/audit/solutions/accounting-reporting-services.html</u>.
- Deloitte. "Corporate Law." Accessed February 3, 2025. https://www.deloitte.com/global/en/services/legal/services/corporate-law.html.
- Deloitte. "Global Office Locations." Accessed March 27, 2025. https://www.deloitte.com/global/en/offices.html.
- U.S. Department of Commerce (USDOC). Bureau of Economic Analysis (BEA). 2022 Benchmark Survey of Transactions in Selected Services and Intellectual Property with Foreign Persons: Form BE-120. August 2022. https://www.bea.gov/sites/default/files/2023-01/BE-120.pdf.
- U.S. Department of Commerce (USDOC). Bureau of Economic Analysis (BEA). "Real Value Added by Industry." September 26, 2024. <u>https://apps.bea.gov/iTable/?reqid=150&step=2&isuri=1&categories=gdpxind#eyJhcHBpZCI6MT</u> <u>UwLCJzdGVwcyI6WzEsMiwzXSwiZGF0YSI6W1siY2F0ZWdvcmllcyIsIkdkcHhJbmQiXSxbIIRhYmxIX0x</u> <u>pc3QiLCIxMCJdXX0=</u>.
- U.S. Department of Commerce (USDOC). Bureau of Economic Analysis (BEA). "The Basics of How International Services are Supplied and Received by the United States." *Survey of Current Business*, October 19, 2022. <u>https://apps.bea.gov/scb/issues/2022/10-october/1022-</u> <u>international-services-supply-types.htm</u>.
- U.S. Department of Commerce (USDOC). Bureau of Economic Analysis (BEA). U.S. International Economic Accounts: Concepts and Methods. June 2024. https://www.bea.gov/resources/methodologies/international/pdf/iea-concepts-methods.pdf.
- U.S. Department of Commerce (USDOC). Bureau of Economic Analysis (BEA). Table 1.1. "U.S. International Transactions." September 19, 2024. <u>https://apps.bea.gov/iTable/?ReqID=62&step=1#eyJhcHBpZCI6NjIsInN0ZXBzIjpbMSwyLDYsNl0sI</u> <u>mRhdGEiOltbIIByb2R1Y3QiLCIxIl0sWyJUYWJsZUxpc3QiLCIxIl0sWyJGaWx0ZXJfIzEiLFsiMSIsIjIiLCIzli</u> <u>wiNCIsIjUiXV0sWyJGaWx0ZXJfIzIiLFsiMSJdXSxbIkZpbHRlcl8jMyIsWyIwII1dLFsiRmIsdGVyXyM0Iixb</u> IjAiXV0sWyJGaWx0ZXJfIzUiLFsiMCJdXV19.
- U.S. Department of Commerce (USDOC). Bureau of Economic Analysis (BEA). Table 2.1. "U.S. Trade in Services, by Type of Service." July 3, 2024. <u>https://apps.bea.gov/iTable/?ReqID=62&step=1#eyJhcHBpZCI6NjIsInN0ZXBzIjpbMSw5LDZdLCJkY</u> <u>XRhIjpbWyJQcm9kdWN0liwiNCJdLFsiVGFibGVMaXN0liwiMjQ1II1dfQ==</u>.
- U.S. Department of Commerce (USDOC). Bureau of Economic Analysis (BEA). Table 2.2. "U.S. Trade in Services, by Type of Service and by Country or Affiliation." July 3, 2024. <u>https://apps.bea.gov/iTable/?ReqID=62&step=1#eyJhcHBpZCI6NjIsInN0ZXBzIjpbMSw5LDEwLDd</u> <u>dLCJkYXRhIjpbWyJQcm9kdWN0IiwiNCJdLFsiVGFibGVMaXN0IiwiMzA1NjgiXSxbIIRhYmxITGIzdFNIY</u> <u>29uZGFyeSIsIjMwNTY5II1dfQ==</u>.

- U.S. Department of Commerce (USDOC). Bureau of Economic Analysis (BEA). Table 3.1. "U.S. International Trade in Services." March 20, 2025. <u>https://apps.bea.gov/iTable/?ReqID=62&step=1#eyJhcHBpZCI6NjIsInN0ZXBzIjpbMSwyLDZdLCJkY</u> <u>XRhIjpbWyJQcm9kdWN0liwiMSJdLFsiVGFibGVMaXN0liwiNTEiXV19</u>.
- U.S. Department of Commerce (USDOC). Bureau of Economic Analysis (BEA). Table 4.1. "Services Supplied to Foreign Persons by U.S. MNEs Through Their MOFAs, by Industry of Affiliate and by Country of Affiliate." October 8, 2024. <u>https://apps.bea.gov/iTable/?ReqID=62&step=1#eyJhcHBpZCI6NjIsInN0ZXBzIjpbMSw5LDZdLCJkY</u> XRhljpbWyJQcm9kdWN0liwiNCJdLFsiVGFibGVMaXN0liwiMzYzll1dfQ==.
- U.S. Department of Commerce (USDOC). Bureau of Economic Analysis (BEA). Table 5.1. "Services Supplied to U.S. Persons by Foreign MNEs Through Their MOUSAs, by Industry of Affiliate and by Country of UBO." October 8, 2024. https://apps.bea.gov/iTable/2RegID=62&step=1#evibcHBpZCI6NiisIpN0ZXBzlipbMSw5LDZdLCIkX

https://apps.bea.gov/iTable/?ReqID=62&step=1#eyJhcHBpZCI6NjIsInN0ZXBzIjpbMSw5LDZdLCJkY XRhIjpbWyJQcm9kdWN0IiwiNCJdLFsiVGFibGVMaXN0IiwiMzY5II1dfQ==.

- U.S. Department of Commerce (USDOC). Bureau of Economic Analysis (BEA). Table 6.5D. "Full-Time Equivalent Employees by Industry." September 27, 2024. <u>https://apps.bea.gov/iTable/?reqid=19&step=2&isuri=1&1921=survey&gl=1#eyJhcHBpZCI6MTk</u> <u>sInN0ZXBzIjpbMSwyLDNdLCJkYXRhIjpbWyJDYXRIZ29yaWVzIiwiU3VydmV5II0sWyJOSVBBX1RhYm</u> xIX0xpc3QiLCIxOTciXV19.
- World Trade Organization (WTO). "Basic Purpose and Concepts." General Agreement on Trade in Services (GATS) Training Module. Accessed September 26, 2024. https://www.wto.org/english/tratop e/serv e/cbt course e/c1s3p1 e.htm.
- World Trade Organization (WTO). Global Services Trade Data Hub. "Trade in Commercial Services." Accessed September 26, 2024. <u>https://www.wto.org/english/res_e/statis_e/gstdh_commercial_services_e.htm</u>.
- World Trade Organization (WTO). "Technical Notes." International Trade Statistics. Accessed September 26, 2024. <u>https://www.wto.org/english/res_e/statis_e/technotes_e.htm</u>.

World Trade Organization (WTO). *World Trade Statistical Review 2023*. 2023. https://www.wto.org/english/res_e/booksp_e/wtsr_2023_e.pdf. Recent Trends in U.S. Services Trade: 2025 Annual Report

Chapter 2 Professional Services

Overview

Professional services comprise a wide range of activities that typically require highly skilled labor. Specialists in this industry often require specific degrees, licenses, or credentials to operate within their respective fields—such as accounting, legal, architecture, engineering, health care, and education.³⁷ Sectors within professional services such as accounting may also be subject to local and national regulations to guarantee legal and ethical services provision.³⁸ These requirements ensure that only qualified specialists offer such services, though they may also limit market entry, particularly across borders where requirements may differ. International trade in professional services typically includes cross-border trade predominantly through digital means or via the movement of persons, and the transactions of affiliated firms.³⁹ Consumers of professional services for social services such as education and health care. Businesses and governments require professional services for business operations and administrative infrastructure such as accounting services, legal services, management consulting services, and research and development (R&D) services.

Within professional services, this report focuses primarily on R&D services, legal services, architectural and engineering services, accounting and audit services, education services, management consulting services, and advertising services.⁴⁰ This chapter begins with a presentation of U.S. cross-border trade and affiliate transactions in professional services over the last five years of available data, followed by detailed sector-specific industry and trade trends. Concluding the chapter, a special topic section provides a closer look at global trends in the provision of and trade in R&D services, a large and growing category of professional services.

Cross-Border Exports and Imports

In 2023, U.S. cross-border exports of professional services (including travel services for the purposes of education and health care) totaled \$378.2 billion following a period of moderate growth since 2020

³⁷ UNCTAD, *Trade and Development Aspects of Professional Services and Regulatory Frameworks*, November 25, 2004, 7–8.

³⁸ Govdysh, *Professional, Scientific and Technical Services in the US*, September 2024, 3.

³⁹ Mann, "Measuring Trade in Services by Mode of Supply," August 2019, 10. See box 1.1 for more information on services trade modes of supply.

⁴⁰ The overall professional services statistics presented in chapters 1 and 2 contain other services—such as health care, scientific and technical consulting, and maintenance and repair services not included elsewhere—which are not discussed separately. See appendix C for a listing of services sectors included in professional services cross-border trade and affiliate services transactions data.

(figure 2.1).⁴¹ Unlike some other services sectors, professional services trade overall experienced a slight decline in 2020 because of the COVID-19 pandemic, but quickly recovered by 2021. Compared to 2019, professional services exports in 2023 grew by 22.3 percent, accounting for 38.1 percent of total crossborder services trade (\$993.8 billion). Top export destinations for professional services in 2023 included Ireland (\$55.9 billion), Switzerland (\$35.3 billion), Canada (\$25.9 billion), and the Netherlands (\$25.2 billion) (figure 2.2).



Source: USDOC, BEA, table 2.1, "U.S. Trade in Services, by Type of Service," July 3, 2024. Note: Total imports and exports include services for education and health care as defined in appendix C, table C.2.

⁴¹ Cross-border trade in travel services for the purposes of education and health care are categorized within professional services trade as "related travel services" throughout this report. See appendix table C.2 for a listing of service sectors included in professional services cross-border trade data.



Source: USDOC, BEA, table 2.2, "U.S. Trade in Services, by Type of Service and by Country or Affiliation," July 3, 2024. Notes: Because of rounding, shares may not add to 100 percent. Total exports include services for education and health care as defined in appendix C, table C.2.

Professional services imports (including related travel services) totaled \$184.7 billion in 2023, resulting in a \$193.5 billion surplus in U.S. professional services trade. Although U.S. imports of professional services slightly decreased in 2023 after substantial growth in 2022, imports were 20.2 percent higher in 2023 compared to the 2019 level. By 2023, imports of professional services accounted for 29.3 percent of total cross-border services trade (\$722.7 billion). Top source countries for professional services imports that year included the United Kingdom (UK) (\$24.4 billion), India (\$16.9 billion), Canada (\$13.8 billion), and Germany (\$12.3 billion) (figure 2.3). Figure 2.3 Professional services: U.S. cross-border imports, by country, 2023

In percentages. Underlying data for this figure can be found in appendix B, table B.14.



Import total: \$184.7 billion

Source: USDOC, BEA, table 2.2, "U.S. Trade in Services, by Type of Service and by Country or Affiliation," July 3, 2024. Notes: Because of rounding, shares may not add to 100 percent. Total imports include services for education and health care as defined in appendix C, table C.2.

The largest category of cross-border trade in professional services was R&D services, which accounted for 31.4 percent of all U.S. professional cross-border services exports in 2023 (figure 2.4).⁴² Management consulting services (27.6 percent) and education services (14.5 percent) were the next-largest categories and jointly with R&D services comprised 73.5 percent of all professional services exports that year.⁴³ Education services include travel services for education (i.e., individuals traveling to the United States to attend U.S. schools), which were the vast majority of U.S. education services exports (over 90 percent in 2023).⁴⁴ Similarly, R&D services, management consulting services, and education services were the largest sources of imports, representing 67.5 percent of U.S. professional services imports in 2023 (figure 2.5).

⁴² In addition to R&D services such as work undertaken on a systematic basis to increase the stock of knowledge, this category includes associated license fees for use of intellectual property outcomes of R&D (i.e., patents, industrial processes, and trade secrets). See appendix C, table C.2 for a listing of service sectors included in professional services cross-border trade data.

⁴³ Management consulting services include business consulting and public relations services. See Management Consulting in this chapter. See appendix C, table C.2 for a listing of service sectors included in professional services cross-border trade data.

⁴⁴ The remaining cross-border education services fall under personal, cultural, and recreational services. See Education Services in this chapter. USDOC, BEA, table 2.2, "U.S. Trade in Services, by Type of Service and by Country or Affiliation," July 3, 2024.

Figure 2.4 Professional services: U.S. cross-border exports, by industry, 2023

In percentages. Underlying data for this figure can be found in appendix B, table B.15.



Export total: \$378.2 billion

Source: USDOC, BEA, table 2.1, "U.S. Trade in Services, by Type of Service," July 3, 2024.

Notes: Because of rounding, shares may not add to 100 percent. Research and development (R&D) services include licenses for the outcomes of R&D. Education includes related travel services. See appendix C, table C.2 for a listing of service sectors included in professional services cross-border trade data.

Figure 2.5 Professional services: U.S. cross-border imports, by industry, 2023

In percentages. Underlying data for this figure can be found in appendix B, table B.16.





Source: USDOC, BEA, table 2.1, "U.S. Trade in Services, by Type of Service," July 3, 2024.

Notes: Because of rounding, shares may not add to 100 percent. Research and development (R&D) services include licenses for the outcomes of R&D. Education services include related travel services. See appendix C, table C.2 for a listing of service sectors included in professional services cross-border trade data.

Affiliate Transactions

In contrast to trade patterns for other services sectors, affiliate transactions by professional services firms are generally smaller than cross-border trade flows.⁴⁵ In 2022, the latest year for which such data are available, sales of professional services by U.S.-owned foreign affiliates abroad totaled \$176.5 billion, representing 8.3 percent of total affiliate sales.⁴⁶ According to available data, architectural, engineering, and related services accounted for the largest share of U.S.-owned foreign affiliate sales (16.9 percent) in 2022, followed by accounting, tax preparation, bookkeeping, and payroll services (9.3 percent); scientific R&D services (7.0 percent); and legal services (5.5 percent) (figure 2.6).⁴⁷

 ⁴⁵ As outlined in chapter 1, these two types of trade flows are not directly comparable due to differences in how services are categorized in BEA cross-border trade and foreign affiliate transaction data. See appendix C for a listing of services included in the professional services cross-border trade and affiliate services transactions data.
 ⁴⁶ USDOC, BEA, table 4.1, "Services Supplied to Foreign Persons by U.S. MNEs," October 8, 2024. Professional services affiliate sales in 2019 and 2022 are underreported because health care and social assistance data are suppressed by the BEA to avoid disclosing individual company information.

⁴⁷ Advertising and related services and management, scientific, and technical consulting services also likely represented a substantial share of sales by U.S.-owned foreign affiliates abroad in 2022, but data for these sales have been suppressed by the BEA to avoid disclosing individual company information. In addition to data for advertising and management consulting services foreign affiliate sales, data for specialized design services and health care and social assistance services are also suppressed by the BEA in 2022.

Figure 2.6 Professional services: sales of U.S.-owned foreign affiliates, by industry, 2022

In percentages. Underlying data for this figure can be found in appendix B, table B.17.





Source: USDOC, BEA, table 4.1, "Services Supplied to Foreign Persons by U.S. MNEs," October 8, 2024. Notes: MNEs = multinational enterprises. Because of rounding, shares may not add to 100 percent. Of the professional services categories discussed in this report, data on advertising and related services and management, scientific, and technical consulting are suppressed in 2022 to avoid disclosure of individual company data. These sectors are included in the "All other" category.

Purchases from affiliates of foreign-owned companies located in the United States totaled \$171.2 billion in 2022, representing 11.3 percent of total affiliate purchases.⁴⁸ Of the professional services categories discussed in this report, advertising and related services represented the largest share of purchases from foreign-owned U.S. affiliates (25.0 percent) in that year, followed by management, scientific, and technical consulting services (17.8 percent) and architectural, engineering, and related services (12.9 percent) (figure 2.7).

⁴⁸ Professional services affiliate purchases are underreported in 2018 and 2019 because waste management and remediation services data are suppressed by the BEA to avoid disclosing individual company information. USDOC, BEA, table 5.1, "Services Supplied to U.S. Persons by Foreign MNEs," October 8, 2024.

Figure 2.7 Professional services: purchases from foreign-owned U.S. affiliates, by industry, 2022

In percentages. Underlying data for this figure can be found in appendix B, <u>table B.18</u>.



Purchases from foreign-owned U.S. affiliates total: \$171 billion

Source: USDOC, BEA, table 5.1, "Services Supplied to U.S. Persons by Foreign MNEs," October 8, 2024. Notes: MNEs = multinational enterprises. Because of rounding, shares may not add to 100 percent. Purchases of legal services and accounting, tax preparation, bookkeeping, and payroll services both represent less than 1.0% of total purchases from foreign-owned U.S. affiliates, and therefore have been included in the "All other" category.

Professional Services by Sector: Background and Trade Trends

This section provides background and presents U.S. trade trends for the major subcategories of professional services included in this report. Legal, architecture and engineering, and accounting and auditing services are further covered in chapter 3. Management consulting, advertising, and education services are further discussed in chapter 4. Additional details on services trade—including data on the industry composition of U.S. services trade with major trading partners and data on U.S. trade in services industries not covered in this report—are available in the interactive figures associated with this report on the Commission's website.⁴⁹

Legal Services

U.S. trade data for legal services are defined either by the industry classification that law firms fall under (which is the case for sales by affiliates) or by the type of service offered by companies (which is the case for cross-border trade). Although both definitions are broad and include the provision of services across

⁴⁹ Interactive figures are available at:

https://www.usitc.gov/publications/industry econ analysis 332/2025/recent trends us services trade 2025 an nual report.

a variety of specific fields of law, international trade in legal services typically involves business and international law fields.⁵⁰

Industry Overview

Attorneys and law firms engaging in international trade typically provide legal advisory services supporting international transactions or business operations, along with other matters involving international law, their home country's law, and third-country law. At the same time, domestic law practice, which pertains to host country law (or the law of the country receiving the service), is widely subject to trade barriers.⁵¹ In particular, foreign entry restrictions often affect domestic law practice (i.e., local qualification requirements to own a law firm) and interact with other barriers such as nationality or residency requirements (to practice domestic law) or impediments to recognizing foreign qualifications.⁵²

In 2023, the global legal services market was valued at \$746 billion.⁵³ The market grew by 4.3 percent from 2022 to 2023, faster than the 3.5 percent average annual growth seen during 2019–23. The United States accounted for 48.0 percent of global revenue in 2023, followed by Europe at 23.8 percent and the Asia-Pacific region at 15.4 percent.⁵⁴ The UK supplied the largest share within Europe (32.4 percent), while China accounted for the largest share within the Asia-Pacific region (62.6 percent).⁵⁵

The U.S. legal services market grew by 1.4 percent between 2022 and 2023 to \$357.8 billion, slower than the global market in the same period. However, the U.S. market grew by an average annual rate of 3.7 percent during 2019–23, similar to global revenue growth in the sector, and the U.S. share of global revenue remained steady over the period.⁵⁶

Firms in the United States and the UK accounted for 92 of the world's 100 top-grossing legal firms. The top five firms globally were all U.S. firms (table 2.1).⁵⁷ The top five firms had a varied presence across

⁵⁰ The industry classification corresponds to *North American Industry Classification System* (NAICS) 5411: Legal Services. See USDOC, BEA, *Guide to Industry Classifications for International Surveys 2022*, 2022, 82; USDOC, BEA, *Quarterly Survey of Transactions*, 2022, 24; Grosso et al., *Services Trade Restrictiveness Index (STRI)*, November 4, 2014, 7.

⁵¹ Grosso et al., Services Trade Restrictiveness Index (STRI), November 4, 2014, 7; OECD, OECD Services Trade Restrictiveness Index (STRI)—Legal Services 2023, January 2024, 3.

⁵² OECD, OECD Services Trade Restrictiveness Index (STRI)—Legal Services 2023, January 2024, 3; OECD, Services Trade Restrictiveness Index Simulator, accessed January 27, 2025.

⁵³ MarketLine, a UK company that provides business information services, calculates regional market value using the total revenue received (including all applicable taxes) by law firms for services rendered. For more information, including on geographic definitions, see MarketLine, *MarketLine Industry Profile Global Legal Services*, February 2024, 7, 9.

⁵⁴ MarketLine, *Global Legal Services Market Summary*, February 2024, 9–10.

⁵⁵ MarketLine, Asia-Pacific (APAC) Legal Services Market Summary, February 2024, 10; Europe Legal Services Market Summary, February 2024, 10.

⁵⁶ In 2019, the United States accounted for 47.5 percent of global revenue (compared with 48.0 percent in 2023, as noted above). MarketLine, *United States (US) Legal Services Market Summary*, February 2024, 9; *MarketLine Industry Profile Global Legal Services*, February 2024, 9.

⁵⁷ American Lawyer, "The 2024 Global 200 Ranked by Revenue," September 17, 2024.

countries (ranging from 6–43 countries) and percentage of lawyers in the United States (ranging from 17–81 percent).⁵⁸

	Country where		Countries in	Share of
	firm has most	Total revenue, FY	which firm has	lawyers in base
Firm	lawyers	2023 (billion \$)	offices	country (%)
Kirkland & Ellis	United States	7.2	6	81
Latham & Watkins	United States	5.7	14	67
DLA Piper	United States	3.8	31	34
Baker McKenzie	United States	3.3	43	17
Skadden, Arps, Slate, Meagher & Flom	United States	3.3	11	72

Table 2.1 Top five global legal firms ranked by revenue, fiscal year 2023In billions of dollars, numbers of countries, and percentages.

Source: American Lawyer, "The 2024 Global 200 Ranked by Revenue," September 17, 2024; American Lawyer, "The 2024 Global 200 Ranked by Head Count," September 16, 2024.

Note: The Global 200 rankings list the country where each firm has the most lawyers. This usually coincides with the firm headquarters but may not be the case when firms have alternate structures. DLA Piper and Baker McKenzie are two such firms with alternate organizational structures.

Cross-Border Trade

During 2019–23, U.S. cross-border exports of legal services increased at an annual rate of 7.7 percent on average, from \$13.1 billion in 2019 to \$17.5 billion in 2023 (figure 2.8). U.S. imports of legal services increased as well at an average annual rate of 4.7 percent, from \$4.4 billion in 2019 to \$5.3 billion in 2023. The top destination for legal services exports was the UK at \$3.7 billion in 2023, followed by Canada (\$1.7 billion) and Germany (\$1.1 billion). Top sources for legal services imports in 2023 were the UK (\$2.2 billion), Japan (\$607 million), Canada (\$284 million), and Germany (\$280 million).⁵⁹

⁵⁸ American Lawyer, "The 2024 Global 200 Ranked by Head Count," September 16, 2024.

⁵⁹ USDOC, BEA, table 2.2, "U.S. Trade in Services, by Type of Service and by Country or Affiliation," July 3, 2024.



Source: USDOC, BEA, table 2.1, "U.S. Trade in Services, by Type of Service," July 3, 2024.

Affiliate Transactions

Sales of legal services by affiliates of U.S. companies in foreign countries increased each year from \$7.6 billion in 2018 to \$10.0 billion in 2021, then slightly decreased to \$9.8 billion in 2022 (figure 2.9). Significantly smaller than foreign affiliate sales, purchases of legal services from U.S. affiliates of foreign firms declined between 2018 and 2020 to a period low of \$80 million in 2020. In subsequent years, purchases increased to \$204 million in 2022 and were 44.7 percent higher compared to 2018.



Figure 2.9 Legal services: U.S. affiliate sales and purchases, 2018–22

Source: USDOC, BEA, table 4.1, "Services Supplied to Foreign Persons by U.S. MNEs," October 8, 2024; USDOC, BEA, table 5.1, "Services Supplied to U.S. Persons by Foreign MNEs," October 8, 2024.

Note: MNEs = multinational enterprises.

Outlook

According to one forecast, the U.S. legal services market is projected to increase at a slower pace in 2025–28 relative to 2019–23.⁶⁰ As discussed in chapter 3, U.S. law firms have adapted their business models using new technologies and have closed offices in China because of regulatory and market conditions. Among other projections, according to reporting from Citi Wealth and Hildebrandt Consulting, U.S. law firms are expected to continue adopting technology to enhance the delivery of their services and growing or scaling back practice areas and investments in foreign markets, depending on their growth potential.⁶¹

Architectural and Engineering Services

Architects provide design and planning services for the construction and renovation of various types of structures and buildings; engineers employ engineering principles and the laws of science in the design and development of structures, systems, and processes, in addition to machines, instruments, and

⁶⁰ MarketLine, *Global Legal Services Market Summary*, February 2024, 9, 11; *United States (US) Legal Services Market Summary*, February 2024, 9, 11. On an average annual growth basis, the global growth forecast for 2025–28 is slightly higher (4.0 percent) compared to 2019–23 (3.5 percent) but the U.S. forecast for 2025–28 is lower (2.3 percent) compared to 2019–23 (3.7 percent). These forecasts appear to apply to the legal industry as defined by law firms.

⁶¹ Citi Wealth and Hildebrandt Consulting, 2025 Citi Hildebrandt Client Advisory, December 5, 2024, 8.

materials.⁶² Key consumers of architecture and engineering services include the construction industry; government; retailers; mining firms; utilities; schools and universities; the entertainment and recreation industry; and additional entities that need designs for buildings and other structures, spaces, and processes. As such, demand for architecture and engineering services depends heavily on factors that affect customers' willingness and ability to finance construction, infrastructure, and industrial projects.⁶³

Industry Overview

During 2019–23, a slight decrease in revenues earned by U.S. architects (from \$55.6 billion to \$54.2 billion) and a small increase in global architecture revenues (from \$253.0 billion to \$260.8 billion) yielded an overall decline in the U.S. share of the global architecture market from 22.0 percent to 20.8 percent. By comparison, revenues earned by U.S. engineering firms grew during the period, from \$324.7 billion in 2019 to \$350.3 billion in 2023. However, faster growth in global engineering revenues (from \$1.5 trillion to \$1.8 trillion) also led to a decreasing U.S. market share in this industry segment. Specifically, U.S. engineering firms accounted for 19.1 percent of global revenues in 2023, down from 21.5 percent in 2019.⁶⁴

In 2024, IBISWorld identified Gensler (headquartered in the United States), Nikken Sekkei Ltd. (Japan), AECOM (United States), and Perkins and Will Inc. (United States) as the world's top architects in terms of revenue, and U.S. firms Jacobs Engineering Group Inc., AECOM, and Fluor Corporation as the world's top engineering firms. Each of these firms accounts for less than 1 percent of global revenues in its respective industry, pointing to low levels of market concentration.⁶⁵ Separately, *Engineering News-Record* indicates that two U.S.-based firms—Jacobs Solutions Inc. and AECOM—rank among the world's five leading earners of design revenues (table 2.2).⁶⁶

⁶² Blau, Engineering Services in the US—Market Size (2004–2031), July 2016; Morea, Architects in the US—Market Research Report (2015–2030), June 2016; Census, "2012 NAICS Definition: 541310 Architectural Services," accessed January 22, 2025.

⁶³ Blau, Engineering Services in the US—Market Size (2004–2031), July 2016; Morea, Architects in the US—Market Research Report (2015–2030), June 2016.

⁶⁴ USITC calculations based on IBISWorld, *Architects in the US*, November 2024; *Engineering Services in the US*, August 2024; *Global Architectural Services*, June 2024; *Global Engineering Services*, May 2024.

⁶⁵ IBISWorld, *Global Architectural Services*, June 2024; *Global Engineering Services*, May 2024.

⁶⁶ Engineering News-Record's list of leading design firms includes entities that provide architectural, engineering,

planning, environmental, geospatial, and geotechnical engineering services, among others. Adolphus and Keller, "The Top 225 International Design Firms," September 2, 2024, 69.

	Headquarters	Revenue, 2023	
Firm	location	(billion \$)	Revenue growth 2022–23 (%)
Power Construction Corp. of China	China	22.3	12.5
China Energy Engineering Corp. Ltd.	China	16.8	11.6
Jacobs Solutions, Inc.	United States	12.7	10.3
WSP Global Inc.	Canada	9.1	9.2
AECOM	United States	8.7	7.8

Table 2.2 Top five global design firms by revenue, 2023In billions of dollars and percentages.

Source: Adolphus and Keller, "The Top 225 International Design Firms," 69; Adolphus et al., "The Top 225 International Design Firms," 48; USITC calculations.

Cross-Border Trade

Between 2019 and 2023, U.S. cross-border exports of architecture and engineering services (including scientific and other technical services) fluctuated, declining by 17.5 percent to \$10.1 billion in 2020, then later increasing to \$12.1 billion in 2023 (figure 2.10).⁶⁷ Engineering services exports remained relatively flat during 2019–23, contributing 91.9 percent of architecture and engineering services exports and totaling \$11.1 billion in 2023. Architecture services, which comprised 5.2 percent of U.S. exports of architecture and engineering services in 2023, fell from \$890 million in 2019 to \$627 million in 2023. U.S. cross-border imports of architecture and engineering services increased substantially from \$7.9 billion in 2019 to \$15.2 billion in 2022, before decreasing by 16.3 percent to \$12.7 billion in 2023. The vast majority of these imports were engineering services (97.5 percent in 2023), which grew from \$7.7 billion in 2019 to \$12.4 billion in 2023.⁶⁸

⁶⁷ Except where noted, data throughout this chapter on cross-border trade and affiliate transactions in architectural and engineering services include scientific and other technical services because detailed data are suppressed by the BEA to protect confidentiality.

⁶⁸ USDOC, BEA, table 2.1, "U.S. Trade in Services, by Type of Service," July 3, 2024.

Figure 2.10 Architectural, engineering, scientific, and other technical services: U.S. cross-border exports and imports, 2019–23



In billions of dollars. Underlying data for this figure can be found in appendix B, table B.21.

Source: USDOC, BEA, table 2.1, "U.S. Trade in Services, by Type of Service," July 3, 2024. Note: Architectural and engineering services include scientific and other technical services because detailed data are suppressed by the BEA to protect confidentiality.

In 2023, Mexico was the leading export destination for architecture and engineering services, totaling \$2.3 billion in 2023, followed by the UK (\$1.0 billion) and Germany (\$468 million). Top sources of architecture and engineering services imports were Mexico (\$2.1 billion), India (\$2.0 billion), and Germany (\$1.6 billion) in 2023.⁶⁹

Affiliate Transactions

Architecture and engineering services supplied by foreign-based affiliates of U.S. firms decreased from \$32.3 billion in 2018 to \$28.6 billion in 2021 before increasing by 4.3 percent to \$29.8 billion in 2022 (figure 2.11). Purchases from U.S.-based affiliates of foreign firms in this sector rose by 15.7 percent to \$22.2 billion in 2022 after being relatively stable between 2019 and 2021.

⁶⁹ USDOC, BEA, table 2.2, "U.S. Trade in Services, by Type of Service and by Country or Affiliation," July 3, 2024.





In billions of dollars. Underlying data for this figure can be found in appendix B, table B.22.

Services supplied by U.S. firms' foreign affiliates Services supplied by U.S. affiliates of foreign firms

Source: USDOC, BEA, table 4.1, "Services Supplied to Foreign Persons by U.S. MNEs," October 8, 2024; USDOC, BEA, table 5.1, "Services Supplied to U.S. Persons by Foreign MNEs," October 8, 2024.

Notes: MNEs = multinational enterprises. Data for architectural, engineering, and related services supplied by U.S. affiliates of foreign firms in 2020 are suppressed by the BEA to protect confidentiality.

Outlook

Over the next five years, factors such as lower interest rates, government infrastructure funding, and growth in residential construction are expected to have a positive effect on demand for architecture and engineering services.⁷⁰ The architecture and engineering workforce is expected to grow, with IBISWorld data suggesting that employment in the U.S. and global architectural services industries would increase at average annual rates of over 2 percent during 2025–30 and 2025–29, respectively.⁷¹ IBISWorld also projects that employment in the U.S. and global engineering services industries would increase at rates of 1.3 percent and 1.0 percent, respectively, from 2025–30.⁷² In a recent survey of engineering leaders, more than half of respondents reported that they intend to expand outsourcing in the near term, an approach used by many firms to fill skills gaps.⁷³ Rising construction activity in certain South American, Asian, and Middle Eastern countries is expected to increase opportunities for large architectural firms in the near future. Further, urbanization will reportedly boost global architectural services demand, with increasing population density forecast to spur demand for certain types of structures—such as thin

⁷⁰ IBISWorld, Architects in the US, November 2024; Engineering Services in the US, August 2024.

⁷¹ IBISWorld did not provide projections for the global architectural services industry for 2030.

⁷² USITC calculations based on IBISWorld, *Architects in the US*, November 2024; *Engineering Services in the US*,

August 2024; Global Architectural Services, June 2024; Global Engineering Services, May 2024.

⁷³ IBISWorld, *Engineering Services in the US*, August 2024.

skyscrapers⁷⁴—in developed-country cities.⁷⁵ One source projected that engineering markets in China and India would experience particularly rapid growth and that the global engineering industry would be impacted by production shifts to South America, Asia, and Eastern Europe.⁷⁶

Accounting and Auditing Services

Firms use accounting services to report financial data in a consistent and compliant format—for financial reporting, government records, and tax filings—and to demonstrate financial health to investors and banks. Small companies especially rely on accounting firms to complete routine financial reporting tasks, including financial statement and balance sheet preparation, tax preparation, and payroll and bookkeeping. Firms specifically rely on accounting firms to conduct audits that assess whether their financial situation has been accurately reported.⁷⁷ The goal of auditors is to verify that financial statements fairly represent the financial position of a company, assess the accuracy of a firm's internal control systems, and provide tax consultation services, which help companies comply with tax regulations and minimize their tax burdens.⁷⁸ These services are particularly valuable to companies that operate in multiple tax jurisdictions.⁷⁹

Industry Overview

IBISWorld estimates that global accounting revenue grew by 1.6 percent in 2024 to \$627.1 billion, slower than the 2.5 percent annual growth rate from 2019 to 2024.⁸⁰ Although accounting revenue remained relatively flat in 2020, the industry was able to increase revenue from 2019 to 2024 largely because of a 6.0 percent increase in 2022.⁸¹ The United States accounted for 29.8 percent of global market value in 2023, followed by Germany (18.7 percent), the UK (12.4 percent), China (7.6 percent), and Japan (1.8 percent).⁸² The largest global accounting firms operate as international "networks" composed of national firms that are legally separate. These local accounting firms act more like franchises than affiliates because they are locally owned but use a common brand.⁸³ Deloitte, PricewaterhouseCoopers (PwC),

⁷⁴ Thin skyscrapers are characterized by their slender profiles and substantial height, which are intended to increase ventilation, views, and light in a minimal footprint. Skyscrapers World, "The Rise of Thin Skyscrapers," February 6, 2025.

⁷⁵ IBISWorld, *Global Architectural Services*, June 2024.

⁷⁶ IBISWorld, *Global Engineering Services*, May 2024.

⁷⁷ Since the 1930s, audits have been required for public companies in the United States, as well as for many privately held firms seeking external financing. Turner and Morse, "Reforms of the Auditing Profession," December 28, 2020; Blanco et al., "Financial Statements Required for Securities Offerings," July 1, 2024.

⁷⁸ Boulay Group, "Guide to Public Company Auditing," February 2021, 2-3; Thomson Reuters, "Auditing in Accounting," January 12, 2023.

⁷⁹ Boulay Group, "Guide to Public Company Auditing," February 2021, 2; PKC Management Consulting, "Auditing for International Businesses," May 29, 2024.

⁸⁰ IBISWorld, *Global Accounting Services*, May 2024, 15.

 ⁸¹ This growth in 2022 was likely due to a post-pandemic rebound including increased corporate dealmaking.
 IBISWorld, *Global Accounting Services*, May 2024, 69; *Scottish Financial News*, "PwC Announces Record Global Revenues of \$50bn," October 4, 2022; Foley and O'Dwyer, "Deloitte Revenues Hit Record," September 8, 2022.
 ⁸² MarketLine, *Global Accountancy*, February 2024, 12.

⁸³ Large global accounting networks continue to consolidate or are considering the consolidation of the national companies in their network. Foley, "US Accounting Firms Rethink Global Networks," July 30, 2024.

Ernst & Young (EY), and KPMG are the largest firms in the industry, collectively known as the "Big Four" and generate nearly 34 percent of global revenue (table 2.3).

Firm	Headquarters	Revenue 2024 (billion \$)	Market share (%)			
Deloitte	United Kingdom	67.2	10.7			
PricewaterhouseCoopers	United Kingdom	55.4	8.8			
Ernst & Young	United Kingdom	51.2	8.2			
KPMG	Netherlands	38.4	6.1			
BDO	Belgium	15	2.4			

In billions of dollars and percentages.

Sources: USITC calculations based on IBISWorld, *Global Accounting Services*, May 2024, 51, 53, 67; MarketLine, *Global Accountancy*, February 2024, 23; BDO, "2024 Global Financial Results," December 18, 2024; Caines, "Robust Growth for KPMG," 2024; Deloitte, "Deloitte 2024 Global Impact Report," September 12, 2024, 5; Lloyd, "EY Reports Global Revenue," EY, October 17, 2024; PwC, "Financials," October 2024.

Cross-Border Trade

In 2023, U.S. cross-border exports of accounting and auditing services increased by 26.4 percent to \$4.2 billion after remaining relatively flat during 2020–22 (figure 2.12). U.S. cross-border imports of accounting and auditing services, however, grew each year throughout 2019–23 at an average annual rate of 13.8 percent, from \$5.0 billion in 2019 to \$8.4 billion in 2023. Top export destinations for accounting and auditing services in 2023 included Canada (\$468 million), Switzerland (\$272 million), and the Netherlands (\$154 million). For imports, top source countries included India (\$1.8 billion), the UK (\$1.3 billion), and Canada (\$667 million) in 2023.⁸⁴

⁸⁴ USDOC, BEA, table 2.2, "U.S. Trade in Services, by Type of Service and by Country or Affiliation," July 3, 2024.



Figure 2.12 Accounting and auditing services: U.S. cross-border exports and imports, 2019–23

Source: USDOC, BEA, table 2.1, "U.S. Trade in Services, by Type of Service," July 3, 2024.

Affiliate Transactions

During 2018–20, accounting and auditing services supplied by affiliates of U.S. companies in foreign countries decreased by 6.3 percent to a period low of \$13.9 billion in 2020, before recovering in 2021 to \$15.8 billion (figure 2.13). By 2022, foreign affiliate sales of accounting and auditing services grew to \$16.4 billion, 10.1 percent higher compared to 2018. Although considerably smaller than foreign affiliate sales, purchases of accounting and auditing services from U.S. affiliates of foreign firms increased 25.8 percent during the period, from \$427 million in 2018 to \$537 million in 2022.



Figure 2.13 Accounting and auditing services: U.S. affiliate sales and purchases, 2018–22

In billions of dollars. Underlying data for this figure can be found in appendix B, table B.24.

Source: USDOC, BEA, table 4.1, "Services Supplied to Foreign Persons by U.S. MNEs," October 8, 2024; USDOC, BEA, table 5.1, "Services Supplied to U.S. Persons by Foreign MNEs," October 8, 2024.

Services supplied by U.S. firms' foreign affiliates Services supplied by U.S. affiliates of foreign firms

Note: MNEs = multinational enterprises.

Outlook

According to IBISWorld, global accounting revenue is expected to continue growing at a rate of 2.0 percent annually from 2025 to 2030.⁸⁵ This projected growth is due to several factors, including greater internet usage around the world, increased global incomes, and the expansion of the customer base to include more businesses from lower-income economies.⁸⁶ According to one industry representative, the accounting industry is projected to be characterized by higher productivity (through digitization, automation, and outsourcing), the increased provision of consulting services (discussed further in chapter 3), and the continued implementation of artificial intelligence (AI) and generative AI (GenAI) technologies.⁸⁷

Education Services

In this report, education services are defined as formal academic instruction at primary, secondary, and post-secondary programs at educational institutions.⁸⁸ This report focuses on education services

⁸⁵ USITC calculations based on IBISWorld, *Global Accounting Services*, May 2024, 69.

⁸⁶ IBISWorld, *Global Accounting Services*, May 2024, 15, 24.

⁸⁷ Raghunathan, "Five Trends That Will Redefine Finance and Accounting," January 7, 2025. AI refers to the application of advanced algorithms such as machine learning (ML) in accounting services and processes. GenAI technologies use sophisticated large-content ML models to generate high-quality original output such as text and images. Zewe, "Explained: Generative AI," November 9, 2023.

⁸⁸ USDOC, BEA, U.S. International Economic Accounts, June 2024, 103.

provided by post-secondary institutions, primarily because international students studying at colleges and universities represent the vast majority of international trade in education services. The share of international students enrolled in U.S. online programs or at the branch campuses of U.S. universities, for example, represents a very small share of total international students.⁸⁹

Industry Overview

In 2023, the global market for education services, measured by total revenues earned by colleges and universities, was estimated at \$1.0 trillion, up 7.9 percent over the previous year.⁹⁰ During that year, the United States was the largest global market for education services, accounting for 23.6 percent of total global revenues. Other large global markets included China (20.7 percent), Japan (6.0 percent), Germany (4.7 percent), and India (3.2 percent). All other country markets accounted for less than 3.0 percent of global revenues, respectively, in 2023.⁹¹ At the global level, the education services industry is highly fragmented along national lines due to the sheer number of universities—estimated to be more than 25,000 institutions in over 100 countries—and the fact that most universities earn the majority of their revenue in home-country markets.⁹² Nonetheless, the largest universities worldwide, measured by enrolled students in 2024, include Indira Ghandi National Open University (India, 7.1 million students), National University (Bangladesh, 2.1 million students), Anadolu University (Türkiye, 2.0 million students), California Community Colleges System (United States, 1.8 million students), and Allama Iqbal Open University (Pakistan, 1.0 million students).⁹³

Cross-Border Trade

U.S. cross-border trade in education services is composed of two sub-categories, depending on how education services are consumed. On the export side, the first category—education services—falls under "other personal, cultural, and recreational services." This category includes instruction for any type of structured learning (and related services), such as correspondence courses and online education provided across borders (mode 1), as well as instruction offered by U.S. educators who travel to foreign markets to provide in-person education services (mode 4).⁹⁴ The second category—education-related travel services—refers to the expenditures by foreign students traveling to the United States to attend U.S. schools (mode 2). Expenses in this category cover tuition, living expenses, and miscellaneous

⁸⁹ For more information, see the "Education Services" section of chapter 4.

⁹⁰ The USITC calculated global revenues for education services using data from Barnes Reports. The global market is defined as revenues collected by colleges and universities in 61 countries in Africa, Asia, Europe, Latin America, the Middle East and North Africa, and North America. Barnes Reports, *Colleges & Universities*, 2023.

⁹¹ Barnes Reports, *Colleges & Universities*, 2023.

⁹² Bouchrika, "60 University Statistics: 2025 Data, Trends & Predictions," January 8, 2025; IBISWorld, *Colleges & Universities*, June 2024, 47.

⁹³ Wilson, "Largest Universities in the World," January 28, 2024.

⁹⁴ USDOC, BEA, *U.S. International Economic Accounts*, June 2024, 116. The education services category does not include educators teaching at a foreign campus of a home university, which would be captured in foreign affiliate sales.

expenses, including expenses for dependents if any.⁹⁵ Education-related travel services accounted for the majority (91.5 percent) of total cross-border education exports in 2023. Its share of total education exports has fluctuated markedly in recent years as a result of the COVID-19 pandemic.⁹⁶

Cross-border exports of education services in 2023 (including related travel services) were \$54.8 billion, and imports totaled \$13.7 billion, resulting in a surplus of \$41.2 billion (figure 2.14).⁹⁷ Education services exports increased by 20.9 percent from 2022 to 2023, continuing the recovery after the pandemic-related decline during 2020–21.⁹⁸ Notably, exports of education services excluding related travel services flourished during the pandemic, rising 173.0 percent from \$2.3 billion in 2019 to \$6.2 billion in 2021, as many schools switched to online learning. By 2023, these education services exports had fallen about 25 percent in comparison with their 2021 peak but were still 103.6 percent higher than the 2019 level. Education services imports (including related travel services) rose at an average annual rate of 23.7 percent, from \$8.9 billion in 2021 to \$13.7 billion in 2023, following a sharp drop between 2019 and 2020.⁹⁹ In regard to education-related travel services, both exports and imports reported steep declines during the pandemic, as travel restrictions reduced the number of international students.¹⁰⁰ Education-related travel exports had fully recovered by 2023 to \$50.2 billion, surpassing the 2019 figure by almost 5 percent.¹⁰¹ Likewise, education-related travel imports also exceeded their 2019 level by 2023, rising about 3 percent to \$11.2 billion.¹⁰²

⁹⁵ Air passenger service for travel between the United States and other countries and between foreign countries is included in the BEA's transport category (air transport). USDOC, BEA, *U.S. International Economic Accounts*, June 2024, 101, 103.

⁹⁶ The share of education travel in total education exports dropped to a low of 84.1 percent in 2021, down from 95.4 percent in 2019. USDOC, BEA, table 2.1, "U.S. Trade in Services, by Type of Service," July 3, 2024.

⁹⁷ Cross-border trade in travel services for the purposes of education are categorized within education services trade as "related travel services" throughout this report. See appendix table C.2 for a listing of service sectors included in professional services cross-border trade data.

⁹⁸ USDOC, BEA, table 2.1, "U.S. Trade in Services, by Type of Service," July 3, 2024.

⁹⁹ Imports of education services excluding related travel services rose about 50 percent from 2019 to 2023. USDOC, BEA, table 2.1, "U.S. Trade in Services, by Type of Service," July 3, 2024.

¹⁰⁰ Imports of education-related travel services exclude expenses for individuals enrolled in primary and secondary programs because of a lack of sources data. USDOC, BEA, *U.S. International Economic Accounts*, June 2024, 103; Open Doors, "Enrollment Trends," accessed February 3, 2025.

¹⁰¹ USDOC, BEA, table 2.1, "U.S. Trade in Services, by Type of Service," July 3, 2024.

¹⁰² USDOC, BEA, table 2.1, "U.S. Trade in Services, by Type of Service," July 3, 2024.



Figure 2.14 Education services: U.S. cross-border exports and imports, 2019–23

Source: USDOC, BEA, table 2.1, "U.S. Trade in Services, by Type of Service," July 3, 2024.

China and India were the top source countries for education-related travel to the United States, accounting for \$14.3 billion and \$11.8 billion, respectively, of exports in 2023. South Korea ranked third in exports (\$2.2 billion). Although education services data (excluding education-related travel) were suppressed for both China and India in 2023, Asia as a whole accounted for 55.1 percent of the global total.¹⁰³ On the import side, Europe accounted for the majority of education-related travel, about 55 percent (\$6.1 billion) of the total in 2023. Top destination countries for U.S. students traveling abroad for education include the UK (\$1.6 billion in imports), followed by Mexico (\$905 million) and Italy (\$826 million). Cross-border data on imports of education services (excluding education-related travel) are suppressed for several countries to avoid the disclosure of data of individual entities, but Europe and Asia combined provide about half (\$1.3 billion) of the total.¹⁰⁴

Affiliate Transactions

In 2022, education services supplied by foreign affiliates of U.S. firms (e.g., international branch campuses of U.S. universities) totaled \$4.0 billion; services purchased from U.S. affiliates of foreign firms were \$3.1 billion (figure 2.15). Foreign affiliate sales were essentially flat in 2022, after dropping 17.1 percent per year on average from 2018 to 2021. In contrast, purchases of education services from U.S. affiliates of foreign firms grew 59.0 percent in 2022, following an uneven performance during 2018–21 (with an average annual decline of 7.9 percent).

¹⁰³ USDOC, BEA, table 2.2, "U.S. Trade in Services, by Type of Service and by Country or Affiliation," July 3, 2024. ¹⁰⁴ USDOC, BEA, table 2.2, "U.S. Trade in Services, by Type of Service and by Country or Affiliation," July 3, 2024.



Figure 2.15 Educational services: U.S. affiliate sales and purchases, 2018–22

Source: USDOC, BEA, table 4.1, "Services Supplied to Foreign Persons by U.S. MNEs," October 8, 2024; USDOC, BEA, table 5.1, "Services Supplied to U.S. Persons by Foreign MNEs," October 8, 2024. Note: MNEs = multinational enterprises.

Outlook

Although the United States is currently the top global destination for international students, over the next few years increased competition from non-U.S. universities and a variety of factors related to the domestic U.S. market suggest that, at a minimum, the share of international students choosing to study in the United States may decrease, if not the actual number of such students.¹⁰⁵ As a mature industry, employment in the higher education sector has grown very slowly over the past five years. Between 2019 and 2023, for example, the number of employees working at colleges and universities increased at an annual rate of only 0.4 percent from 3.0 million in 2019 to 3.1 million in 2023, albeit with a decline of 2.3 percent in 2020 due to layoffs during the COVID-19 pandemic. Employment in the sector is expected to continue to grow at an average annual pace of only 0.9 percent to 3.2 million during 2023–28.¹⁰⁶

Management Consulting

Management consulting services provide advice to businesses, public sector entities, and nonprofits on a range of operational functions, including organizational design and strategy, human resources, marketing, financial planning and budgeting, and logistics and distribution.¹⁰⁷ Overlap is considerable

 ¹⁰⁵ Shorelight, "U.S. Loses International Student Market Share," May 22, 2023; Armstrong, "Why Fewer International Students Are Coming to the US," December 13, 2023; Irwin-Hunt, "The Great Competition for International Students," March 27, 2024; Fischer and Aslanian, "Fading Beacon," August 2, 2021.
 ¹⁰⁶ USITC calculations using data provided by IBISWorld. IBISWorld, *Colleges & Universities*, June 2024, 73, 75.
 ¹⁰⁷ IBISWorld, *Management Consulting in the US*, August 2024, 5.

with firms from other sectors that are not categorized as management consultants by the *North American Industry Classification System*,¹⁰⁸ particularly accountancy (as noted above in the Accounting and Auditing Services section, the Big Four firms are Deloitte, PwC, EY, and KPMG) and information services (DiscoverOrg, Accenture, Cognizant Technology Solutions, and Tata Consulting Services).¹⁰⁹

Industry Overview

IBISWorld estimates that global revenue from management consulting services was \$1.0 trillion in 2024, with U.S. revenue accounting for \$392.2 billion.¹¹⁰ During 2014–24, the U.S. share of global management consulting services revenues consistently stood at about 38 percent. Real growth in U.S. management consulting services revenues slowed to an estimated 1.8 percent in 2024, down from 4.1 percent on average in 2019–23. Those growth rates were in line with the global trends.¹¹¹

According to IBISWorld, the four largest players in the U.S. management consulting market are Deloitte, DiscoverOrg, Accenture, and Cognizant and the global leaders are Accenture, KPMG, and McKinsey & Company. The sector is highly segmented, however, and IBISWorld reports that no company has more than 3 percent of the market, either in the United States or globally. ¹¹² The top 20 firms in the United States account for about one-quarter of total revenues. ¹¹³ In 2018, the Big Four accounting firms had a share of the consulting market that was nearly twice the size of the six largest pure-play management consulting firms, ¹¹⁴ which include McKinsey & Company, Boston Consulting Group (BCG), and Bain & Company, among others. ¹¹⁵

The U.S. consulting sector is international by nature. The largest U.S. consulting firms have global operations, with staff located around the world.¹¹⁶ Practice groups, or areas of expertise within consulting firms, often have team members in multiple countries.¹¹⁷

Cross-Border Trade

In 2023, U.S. cross-border exports of management consulting services, which include business consulting and public relations services, totaled \$104.4 billion (figure 2.16).¹¹⁸ Imports stood at \$53.7 billion, resulting in a surplus of \$50.8 billion. Management consulting exports reported only slight growth during

¹¹⁴ Pure-play management consulting firms such as McKinsey & Company specialize in strategy consulting.

¹¹⁷ Williams and van Triest, "Researching Practice Areas of Consultancy Firms," May 24, 2023, 105.

¹⁰⁸ Management Consulting Services fall under NAICS 54161. See Census, "North American Industry Classification System (NAICS)," 2022, 466–69.

¹⁰⁹ Sussman, "Separating Auditing from Consulting," July 28, 2021; IBISWorld, *Management Consulting in the US*, August 2024, 6.

¹¹⁰ IBISWorld, *Global Management Consultants*, December 2024, 35; *Management Consulting in the US*, August 2024, 11.

¹¹¹ IBISWorld, *Global Management Consultants*, December 2024, 35; *Management Consulting in the US*, August 2024, 87.

¹¹² IBISWorld, *Global Management Consultants*, December 2024, 24; *Management Consulting in the US*, August 2024, 57.

¹¹³ IBISWorld, *Management Consulting in the US*, August 2024, 15.

¹¹⁵ Sussman, "Separating Auditing from Consulting," July 28, 2021.

¹¹⁶ Morgan et al., *The Globalization of Management Consultancy Firms*, May 2006, 5.

¹¹⁸ See appendix table C.2 for a listing of service sectors included in professional services cross-border trade data.

2023, increasing by less than 1 percent compared to the prior year following average annual growth of 16.5 percent during 2019–22. Despite the deceleration in growth in 2023, management consulting services were the leading driver of U.S. professional services exports during 2019–23. Top destinations for management consulting exports in 2023 included Ireland (\$23.1 billion), Switzerland (\$11.1 billion), and the UK (\$8.1 billion).¹¹⁹



Figure 2.16 Management consulting services: U.S. cross-border exports and imports, 2019–23

Source: USDOC, BEA, table 2.1, "U.S. Trade in Services, by Type of Service," July 3, 2024.

Cross-border imports of management consulting services rose 10.9 percent in 2023 compared with the 2022 level, accelerating from an average annual growth rate of 7.1 percent during 2019–22. Top sources of management consulting imports in 2023 included the UK (\$9.4 billion), India (\$5.0 billion), and Canada (\$4.3 billion).

Affiliate Transactions

Total sales of management consulting services (including scientific and technical consulting) by foreign affiliates of U.S. firms are suppressed by the Bureau of Economic Analysis (BEA) for the years 2014 to 2022 in order to safeguard individual companies' information.¹²⁰ BEA data indicate that purchases of management consulting services from foreign-owned affiliates in the United States fell from \$26.3 billion

¹¹⁹ USDOC, BEA, table 2.2, "U.S. Trade in Services, by Type of Service and by Country or Affiliation," July 3, 2024. ¹²⁰ Although total figures are suppressed, U.S. foreign affiliate sales data have been published by the BEA for many countries and regions. In 2022, top destinations for management consulting services included the UK (\$6.7 billion), Germany (\$5.2 billion), and Japan (\$3.6 billion). Summing the regional-level data reveals that foreign affiliate sales in management consulting totaled at least \$38.3 billion in 2022. USDOC, BEA, table 4.1, "Services Supplied to Foreign Persons by U.S. MNEs," October 8, 2024.

in 2018 to \$23.1 billion in 2020 before rising to \$30.4 billion in 2022 (figure 2.17). Though U.S. sales by affiliates of foreign firms declined by 12.2 percent in 2020, management consulting foreign affiliate sales rebounded in subsequent years and rose by 24.4 percent in 2022 compared to 2021.

Figure 2.17 Management, scientific, and technical consulting services: U.S. affiliate sales and purchases,



In billions of dollars. Underlying data for this figure can be found in appendix B, table B.28.

Source: USDOC, BEA, table 4.1, "Services Supplied to Foreign Persons by U.S. MNEs," October 8, 2024; USDOC, BEA, table 5.1, "Services Supplied to U.S. Persons by Foreign MNEs," October 8, 2024.

Notes: MNEs = multinational enterprises. Services supplied by foreign affiliates of U.S. firms in 2018-22 are not shown because data were suppressed by the BEA to protect confidentiality.

Outlook

2018-22

Management consulting revenues in the United States are broadly expected to continue to rise over the next five years, although the growth outlook varies considerably by source.¹²¹ According to an IBM global survey of corporate executives, the average consulting spending will increase to more than 4 percent of total revenue by 2026, up from 2.8 percent in 2024.¹²² The United States is expected to remain a leading source of growth for the consulting sector, alongside emerging markets such as China, India, and Indonesia.¹²³ From a labor perspective, the U.S. Bureau of Labor Statistics projects that employment of

¹²¹ IBISWorld projects annual real revenue growth of less than 1 percent in 2025–29, but MarketLine expects real growth of about 5 percent annually. See IBISWorld, Management Consulting in the US, August 2024, 11; MarketLine, US Management Consultancy Report, March 2024, 13.

¹²² Candy et al., "Consulting Reimagined, Powered by AI," October 2024, 6.

¹²³ IBISWorld, *Global Management Consultants*, December 2024, 18.

management analysts will rise 11 percent from 2023 to 2033, considerably faster than the average for all occupations.¹²⁴

Box 2.1 Management Consulting and Bureau of Economic Analysis (BEA) Trade Data

Although consulting is a global business, the complexity of services trade data makes it difficult to draw firm conclusions about international trade patterns. Official trade statistics from the BEA present two different broad types of services trade. First, cross-border services trade (modes 1, 2, and 4) statistics are based on the type of service traded. Second, statistics measuring services supplied through affiliates (mode 3) are based on the affiliate's primary industry (see chapter 1). Thus, comparability is limited between statistics for cross-border trade and foreign affiliate sales at the sector level. For example, a bank like JPMorgan Chase that also provides management consulting services could report cross-border trade in professional services but, because it is primarily a financial services firm, its sales of consulting services would likely not appear under the professional services category in the BEA's foreign affiliate transactions data.

BEA data on cross-border sales of business and management consulting and public relations services indicate that U.S. parent companies' exports to their foreign affiliates accounted for 71.0 percent of total management consulting exports in 2023. U.S. affiliates' trade with their foreign parent groups made up another 21.0 percent of exports, signaling that unaffiliated trade comprised only about 8 percent of total consulting exports.

A 2024 study published in the BEA's *Survey of Current Business* indicates that cross-border exports in the broader professional and management consulting services category (which also includes legal, accounting, and advertising and related services) are supplied by various industries. Manufacturing companies accounted for the largest share of cross-border consulting services exports in 2022 (32.0 percent), followed by firms in the information services (26.4 percent) and professional services (21.6 percent) industries.^a Examples of cross-border consulting exports by a U.S. manufacturing company could include the following: The U.S. headquarters of a manufacturing firm advises its foreign subsidiary on retooling efforts, or a U.S. company sends staff abroad for several months to work with a foreign subsidiary on supply chain optimization efforts. The same BEA study indicates that exporters in the broader professional and management consulting services category tend to be "very large" firms, with 10,000 or more employees. In 2021, very large companies accounted for about 71 percent of total professional and management consulting services exports.^b

^a Bruner and Grimm, "A Profile of U.S. Services Traders, 2006–2022," May 31, 2024, 9. ^b Bruner and Grimm, "A Profile of U.S. Services Traders, 2006–2022," May 31, 2024, 23.

Advertising Services

Advertising services include creating awareness, generating interest, and persuading consumers to take a particular action such as purchasing a product, visiting a website, or supporting a brand. Traditional advertising refers to print and broadcast media such as newspaper ads, televisions commercials, cinema

¹²⁴ USDOL, BLS, "Management Analysts," accessed January 27, 2025.

ads, billboard ads, and radio infomercials.¹²⁵ Digital advertising refers to the practice of promoting products or services through online channels by purchasing ad placements with the aim of driving traffic, increasing brand awareness, and generating leads or sales using data-driven strategies. This form of advertising uses the internet in various forms such as digital video, search engine, social media, digital out-of-home, digital banner, digital audio, digital classifieds, and online influencer advertising.¹²⁶ The leading providers of advertising services are technology and media companies due to their prominence in digital advertising services.¹²⁷ Beyond those firms, major global advertising firms include Dentsu Group (Japan), Interpublic Group of Companies (United States), Publicis Groupe (France), and WPP (UK), which provide traditional advertising, public relations, and market research services in addition to digital advertising services.¹²⁸

Industry Overview

Statista Market Insights estimates that U.S. advertising services market revenue was \$425.9 billion in 2024, an increase of over 70 percent from \$274.3 billion in 2019.¹²⁹ This increase has coincided with the growth in the global advertising services market, which rose 50.8 percent, from \$720.8 billion in 2019 to an estimated \$1.1 trillion in 2024.¹³⁰ According to the most recent data, the three companies with the most advertising revenue in the world in 2022 were headquartered in the United States (Google, Meta, and Amazon), and the next two companies were headquartered in China (ByteDance and Alibaba) (table 2.4).¹³¹

¹²⁵ American Marketing Association, "Marketing vs. Advertising," accessed January 15, 2025; MarketLine, *Global Advertising Market Summary*, April 2024, 7.

¹²⁶ Digital advertising comprised roughly 60 percent of global advertising revenue in 2023 and 2024. Statista, "Share of Digital in Advertising Revenue Worldwide," January 10, 2025; PwC, "Perspectives from the Global Entertainment & Media Outlook," July 16, 2024; MarketLine, North America (NAFTA) Advertising Market Summary, May 2024, 9.

¹²⁷ Statista, Advertising: Market Data & Analysis, November 2024, 251–55, 265–69.

¹²⁸ MarketLine, *Global Advertising Market Summary*, April 2024, 23.

¹²⁹ Statista, "Advertising: United States," October 2024.

¹³⁰ Statista, Advertising: Market Data & Analysis, November 2024, 10.

¹³¹ ByteDance is an internet technology company that owns TikTok. McNally, "Global Ad Revenue Cracks The \$1 Trillion Dollar Mark In 2024," December 8, 2024.

Company	Headquarters	Industry	Advertising revenue		
Google	United States	Technology	224.5		
Meta	United States	Social media	113.6		
Amazon	United States	E-commerce and technology	37.7		
ByteDance	China	Social media	31.7		
Alibaba	China	E-commerce and technology	31.5		
	//· ·· ·· ·· ·· ·· ·· ·· ·· ·· ·· ·· ··				

 Table 2.4 Top five global advertising providers by advertising revenue, 2022

 In billions of dollars

Source: Statista, "Leading Media Companies Worldwide in 2022," September 2023.

Cross-Border Trade

Advertising and related services consist of three subcategories: advertising services, research and public opinion polling services, and trade exhibition and sales convention services. In 2023, the advertising services subcategory accounted for nearly all advertising and related services exports (98.1 percent).¹³²

Since 2019, cross-border exports of advertising and related services grew by 12.9 percent to \$24.3 billion in 2023, notwithstanding modest declines in 2020 and 2022 (figure 2.18). Growth was considerably faster for the advertising services subcategory, which rose by 23.7 percent over the period; the total category was affected by decreases in the other two subcategories, particularly research and public opinion polling. Imports of advertising and related services fell by 28.2 percent in 2020 but grew at an average annual rate of 15.8 percent over the next three years to \$6.7 billion by the end of 2023. Canada was the top destination for U.S. exports of advertising and related services, at \$8.2 billion in 2023, followed by Ireland (\$3.3 billion) and Singapore (\$2.5 billion). On the import side, the top source countries were the UK (\$1.2 billion), Canada (\$907 million), and Ireland (\$826 million).¹³³

¹³² In 2023, research and public opinion polling accounted for 1.7 percent and trade exhibition and sales convention services were 0.2 percent of advertising and related services. USDOC, BEA, table 2.1, "U.S. Trade in Services, by Type of Service," July 3, 2024.

¹³³ USDOC, BEA, table 2.2, "U.S. Trade in Services, by Type of Service and by Country or Affiliation," July 3, 2024.





Source: USDOC, BEA, table 2.1, "U.S. Trade in Services, by Type of Service," July 3, 2024.

Affiliate Transactions

Foreign affiliate trade data are currently available through 2022; however, data on total advertising and related services supplied by foreign affiliates of U.S. firms were suppressed by the BEA in 2021 and 2022 to safeguard individual companies' information (figure 2.19). Trade statistics that are available by country and region indicate that the value of services supplied by foreign affiliates of U.S. firms reached at least \$16.7 billion in 2022, which is close to 2018 and 2019 levels. Services purchased from U.S. affiliates of foreign firms totaled \$42.9 billion in 2022, a 9.4 percent increase following average annual growth of less than 1 percent during 2018–21.



Figure 2.19 Advertising and related services: U.S. affiliate sales and purchases, 2018–22

In billions of dollars. Underlying data for this figure can be found in appendix B, table B.30.

Services supplied by U.S. firms' foreign affiliates Services supplied by U.S. affiliates of foreign firms

Source: USDOC, BEA, table 4.1, "Services Supplied to Foreign Persons by U.S. MNEs," October 8, 2024; USDOC, BEA, table 5.1, "Services Supplied to U.S. Persons by Foreign MNEs," October 8, 2024.

Notes: MNEs = multinational enterprises. Services supplied by foreign affiliates of U.S. firms in 2021 and 2022 are not shown because data were suppressed by the BEA to protect confidentiality.

Outlook

After estimated annual growth of 7.5 percent in 2024, global advertising revenue is forecast to continue rising at around 7.5 percent in 2025–29.¹³⁴ This projected growth in revenue is due to several factors, including the increase in digital consumption and the rise of video streaming services, social media, social commerce, and AI used in advertising.¹³⁵

¹³⁴ Statista, Advertising: Market Data & Analysis, November 2024, 10.

¹³⁵ PwC, "Perspectives from the Global Entertainment & Media Outlook," July 16, 2024; Statista, *Advertising: Market Data & Analysis*, November 2024; McKinsey & Company, "Social Commerce," October 19, 2022; PwC, "Digital Ad Spending Moves Closer to the Consumer," accessed January 13, 2025; O'Malley, "Shifts in Digital Advertising," April 16, 2024.
Special Topic: Trade in Research and Development Services

R&D services account for almost one-third of U.S. cross-border trade in professional services, but a much smaller share of affiliate transactions by professional services firms.¹³⁶ Because R&D services, their outcomes, and charges for the use of R&D outcomes are inputs into the production of final goods, measuring R&D services trade is a challenge. This special topic section updates the statistics and trends reported in *Recent Trends in U.S. Services Trade: 2021 Annual Report* and discusses the tax-driven nature of R&D trade and spending.

Industry Overview

R&D services include work aimed at expanding knowledge or producing novel or significantly improved goods and services. R&D investments drive innovation and product development in many sectors, especially technology, health care, and manufacturing. R&D is essential to economic growth and U.S. global competitiveness, with major funding coming from both the public and private sectors.¹³⁷

In 2022, the most recent year for which data are available, R&D global spending reached \$2.5 trillion.¹³⁸ IT hardware accounted for the largest share of global R&D spending at 24.1 percent, followed by health care at 22.0 percent and software and the internet at 21.9 percent.¹³⁹ IT hardware and health care maintained similar shares of global R&D spending throughout 2018–22. In contrast, the share of R&D spending on software increased by 6 percent and automobile R&D spending declined by 2 percent during the period.¹⁴⁰ U.S. firms in these sectors conduct a large part of their R&D activities in the United States, which they then transfer to their manufacturing subsidiaries often located in low-tax jurisdictions around the world.

The United States and China significantly outpaced all other countries in R&D spending. The Organisation for Economic Co-operation and Development (OECD) compiles data across industry sectors on business enterprise research and development expenditure for both basic and applied R&D.¹⁴¹ The countries accounting for the largest shares of business enterprise research and development spending in 2021 (latest available data) were the United States (\$565 billion), China (\$488 billion), Japan (\$135

¹³⁶ Cross-border exports and imports of R&D services include both licenses for the use of outcomes of research and development under the "charges for the use of intellectual property n.i.e." category and "research and development services: total" and its subcategories, which are classified under "other business services" in the BEA's cross-border trade statistics. USDOC, BEA, table 2.1, "U.S. Trade in Services, by Type of Service," July 3, 2024.
¹³⁷ Boroush and Guci, "Research and Development," April 28, 2022, 37, 55.

¹³⁸ Statista, "Research and Development (R&D) Worldwide," 2024, 4.

¹³⁹ The automobile and other transport industries contributed 14.0 percent; all other sectors combined accounted for 18.0 percent. Statista, "Research and Development (R&D) Worldwide," 2024, 14.

¹⁴⁰ USITC, *Recent Trends in U.S. Services Trade: 2021 Annual Report*, April 2021, 52; Statista, "Research and Development (R&D) Worldwide," 2024, 14.

¹⁴¹ OECD, *Frascati Manual 2015*, October 8, 2015, 212.

billion), South Korea (\$87 billion), and Germany (\$87 billion).¹⁴² During 2017–21, the United States and China increased their R&D spending by average annual rates of 8.7 percent and 9.3 percent, respectively, and South Korea increased its spending by an average annual rate of 5.6 percent. Japan and Germany roughly maintained their business enterprise research and development expenditures over the four-year period (figure 2.20).¹⁴³





In billions of U.S. dollars. Underlying data for this figure can be found in appendix B, table B.31.

Source: OECD, "Main Science and Technology Indicators—Business Enterprise Expenditure on R&D (BERD)," March 2024. Note: Because of classification issues, data for China are not strictly comparable to data for other economies.

In terms of R&D spending as a percentage of gross domestic product (GDP), the top two countries in 2021 were Israel and South Korea with 5.6 and 4.9 percent, respectively. The United States was third, with 3.5 percent of its GDP dedicated to R&D, up from 2.9 percent in 2017. Belgium (3.4 percent) and Sweden (3.4 percent) rounded out the top five countries in R&D spending as a percentage of GDP.¹⁴⁴

In 2022, global R&D spending was driven by firms in the information technology sector, especially firms based in the United States. In 2022, as in 2018, Amazon ranked first in R&D spending worldwide with

¹⁴² These numbers exclude spending by the government. In *Recent Trends in U.S. Services Trade: 2021 Annual Report*, business enterprise expenditure on R&D data for 2017 were represented by industry (manufacturing, services, and other). Because the United States and China no longer categorize R&D services spending specifically, the industry breakdown has been eliminated from this report. See USITC, *Recent Trends in U.S. Services Trade: 2021 Annual Report*, April 2021, 53; OECD, "Main Science and Technology Indicators—Business Enterprise Expenditure on R&D (BERD)," March 2024.

 ¹⁴³ OECD, "Main Science and Technology Indicators—Business Enterprise Expenditure on R&D (BERD)," March 2024.
 ¹⁴⁴ World Bank Group, "Research and Development Expenditure (% of GDP)," September 30, 2024.

\$73.2 billion, more than tripling its R&D expenditures during that four-year period. Notably, 9 of the top 10 R&D spenders were technology firms, with Volkswagen the exception in 10th place. This marks a shift from 2018, when three pharmaceutical companies were among the top 10 R&D spenders. Six U.S.-headquartered firms ranked among the top 10 R&D spenders in 2022, one fewer than in 2018. China's Huawei and Tencent Holdings entered the list of top 10 R&D spenders in 2022; no Chinese firms ranked among the top 10 in 2018 (table 2.5).¹⁴⁵

Firm	Headquarters	Primary industry	R&D spending, billion \$
Amazon	United States	Technology	73.2
Meta	United States	Technology	35.3
Alphabet	United States	Technology	28.8
Apple	United States	Technology	26.3
Huawei	China	Technology	23.2
Microsoft	United States	Technology	22.7
Samsung	South Korea	Technology	18.2
Tencent Holdings	China	Technology	17.7
Intel	United States	Technology	17.5
Volkswagen	Germany	Automobiles	17.1

Table 2.5 Top 10 firms by global R&D spending, 2022

In billions of dollars.

Source: Statista, "Companies with Highest R&D Spending Worldwide 2022," May 2023.

Cross-Border Trade of R&D Services

BEA data on cross-border trade in R&D services include "R&D services: total," a subcategory of "other business services." This subcategory includes two further subcategories, (1) "provision of customized and non-customized R&D services" and (2) "the sale/purchase of proprietary rights arising from R&D," and "licenses for the use of outcomes of R&D" under charges for the use of intellectual property n.i.e.¹⁴⁶ "R&D services" includes activities aimed at expanding knowledge and producing novel or significantly improved goods and services, as well as nonroutine testing and product development activities. "Licenses for the use of outcomes of R&D" includes charges for the use of a patent, process, or trade secrets to produce or distribute a product or service.¹⁴⁷ In 2023, licenses for use of outcomes of R&D made up 53.9 percent of total exports in these categories and 41.4 percent of total imports.¹⁴⁸ In 2023, total U.S. cross-border exports of R&D services (including both R&D services and licenses for use of outcomes of secrets to prote exports of R&D services (including both R&D services and licenses for use of outcomes of secrets to a patent exports of R&D services (including both R&D services and licenses for use of outcomes of secrets to a patent exports of R&D services (including both R&D services and licenses for use of outcomes of secrets to a patent exports of R&D services (including both R&D services and licenses for use of outcomes of secrets to a patent exports of R&D services (including both R&D services and licenses for use of outcomes of secrets to a patent exports accounted for the largest share of U.S. professional services exports that year.¹⁵⁰ Following growth in R&D services exports of 15.3 percent and 11.5 percent in 2021

¹⁴⁵ Statista, "Companies with Highest R&D Spending Worldwide 2022," May 2023; USITC, *Recent Trends in U.S. Services Trade: 2021 Annual Report*, April 2021, 52.

¹⁴⁶ USDOC, BEA, 2022 Benchmark Survey of Transactions, August 2022, 32.

¹⁴⁷ USDOC, BEA, U.S. International Economic Accounts, June 2024, 115; USDOC, BEA, 2022 Benchmark Survey of Transactions, August 2022, 32.

¹⁴⁸ USDOC, BEA, table 2.1, "U.S. Trade in Services, by Type of Service," July 3, 2024.

¹⁴⁹ USDOC, BEA, table 2.1, "U.S. Trade in Services, by Type of Service," July 3, 2024.

¹⁵⁰ USDOC, BEA, table 2.1, "U.S. Trade in Services, by Type of Service," July 3, 2024.

and 2022, respectively, U.S. exports decreased by 4.8 percent in 2023. Total cross-border imports of R&D services increased from 2019 to 2022 at an annual growth rate of 7.6 percent before declining in 2023 by 19.0 percent (figure 2.21).



Figure 2.21 R&D services: U.S. cross-border exports and imports, 2019–23

In billions of dollars. Underlying data for this figure can be found in appendix B, table B.32.

Source: USDOC, BEA, table 2.1, "U.S. Trade in Services, by Type of Service," July 3, 2024.

Note: Exports and imports of R&D services include sale of licenses for outcomes of research and development and nonroutine testing and product development activities.

The top destinations for total U.S. R&D exports in 2023 were Ireland (\$28.8 billion), Switzerland (\$22.9 billion), and the Netherlands (\$13.7 billion). U.S. exports of R&D services to Ireland and Switzerland were balanced between the subcategories, with licenses for the use of outcomes of R&D accounting for 48.1 percent and 60.2 percent of total exports to the countries, respectively. Exports of licenses for the use of outcomes of R&D to the Netherlands made up 91.3 percent of R&D services exports. The top sources of U.S. R&D imports in 2023 were Japan (\$8.5 billion), India (\$5.6 billion), and the UK (\$5.1 billion). Licenses for the use of outcomes of R&D made up 90.4 percent of U.S. imports of R&D services from Japan; the same category accounted for only 21.8 percent of U.S. imports from India. U.S. imports of R&D services from the UK were balanced, with licenses for the use of outcomes of R&D, accounting for 46.6 percent (figure 2.22).¹⁵¹

¹⁵¹ USDOC, BEA, table 2.2, "U.S. Trade in Services, by Type of Service and by Country or Affiliation," July 3, 2024.



Figure 2.22 R&D services: U.S. cross-border exports to selected countries and imports from selected

countries, 2023

Source: USDOC, BEA, table 2.2, "U.S. Trade in Services, by Type of Service and by Country or Affiliation," July 3, 2024. Note: "R&D services" includes activities aimed at expanding knowledge and producing novel or significantly improved goods and services, as well as nonroutine testing and product development activities. "Licenses for the use of outcomes of R&D" includes charges for the use of a patent, process, or trade secrets to produce or distribute a product or service. See appendix C, table C.2 for BEA subsectors of professional services.

Sales and Purchases by Foreign R&D Services Affiliates

The total value of foreign transactions by R&D services affiliates is substantially lower than cross-border trade in R&D services. U.S. sales by R&D affiliates (affiliate sales) are instances where a foreign R&D affiliate of a U.S. multinational enterprise provides services to its U.S. parent, other foreign affiliates, or external clients.¹⁵² For example, a foreign lab might develop technology and transfer it to U.S. headquarters or sell it to third parties. U.S. purchases from R&D affiliates (affiliate purchases) occur when a U.S. R&D affiliate of a foreign-based firm sells services (such as technology or research support) to its foreign parent company or provides services to local U.S. institutions or other affiliates.

In 2022, the latest year for which data are available, services supplied by foreign R&D affiliates of U.S. firms totaled \$12.4 billion and services purchased from U.S. R&D affiliates of foreign firms were \$6.7

¹⁵² Foreign affiliate transactions statistics cover only the activities of firms that specialize in R&D and as a result are likely to be weighted more toward basic R&D. For this reason, foreign affiliate transactions and cross-border trade in R&D services cannot be directly compared. For a more detailed discussion on foreign affiliate transactions statistics, see USITC, *Recent Trends in U.S. Services Trade: 2021 Annual Report*, April 2021, 55.

billion.¹⁵³ Foreign affiliate sales dropped by 54.3 percent in 2019, then grew steadily from 2020 to 2022 with an annual growth rate of 9.8 percent. U.S. purchases from foreign-owned affiliates grew from 2018 to 2022, with an annual growth rate of 22.4 percent, rising from \$3.0 billion to \$6.7 billion (figure 2.23).



Figure 2.23 R&D services: U.S. affiliate sales and purchases, 2018–22

In billions of dollars. Underlying data for this figure can be found in appendix B, table B.34.

Services supplied by U.S. firms' foreign affiliates Services supplied by U.S. affiliates of foreign firms

Source: USDOC, BEA, table 4.1, "Services Supplied to Foreign Persons by U.S. MNEs," October 8, 2024; USDOC, BEA, table 5.1, "Services Supplied to U.S. Persons by Foreign MNEs," October 8, 2024.

Note: MNEs = multinational enterprises.

¹⁵³ USDOC, BEA, table 4.1, "Services Supplied to Foreign Persons by U.S. MNEs," October 8, 2024; USDOC, BEA, table 5.1, "Services Supplied to U.S. Persons by Foreign MNEs," October 8, 2024.

Tax Incentives and R&D Trade and Investment

Tax incentives, both in the United States and in foreign countries, are meant to encourage innovation and attract foreign firms to invest in specific markets.¹⁵⁴ Conversely, high tax burdens or restrictive fiscal policies can deter R&D activities.¹⁵⁵ For example, intellectual property (IP) regimes, referred to as "IP boxes" or "patent boxes," are tax policy tools designed to incentivize R&D by offering preferential tax rates on income generated from IP.¹⁵⁶ These regimes allow companies to benefit from significantly lower tax rates on qualifying income from IP, such as patents, trademarks, copyrights, software, or designs, compared to the standard corporate tax rate.

The recipients of the highest total U.S. R&D services exports are not necessarily the countries that rank highest in a variety of other R&D measures. For example, in 2021 South Korea received only 1.0 percent of U.S. R&D services exports but ranked first globally in both domestic spending in R&D as a percentage of GDP and researchers in R&D per million population. On the other hand, Ireland, the largest destination of U.S. R&D services exports in 2021, with a 28.2 percent share, spent only 1.1 percent of its GDP on R&D and ranked 17th in the number of researchers in R&D per capita (table 2.6). Varying taxation levels for R&D and IP assets among the major destinations for U.S. exports, as well as the location of R&D-intensive business activities (such as pharmaceuticals), likely explain part of this pattern of trade.

	Percentage of	Domestic	Percentage of total	Global ranking				
	total U.S. R&D	spending on R&D	U.S. foreign	for business	Global ranking			
	services	as a percentage	affiliate spending	spending on	for researchers in			
Country	exports ^a	of GDP ^b	on R&D	R&D ^c	R&D per capita			
Ireland	28.2	1.1	31.6	22	17			
Switzerland	13.4	3.3	15.0	13	10			
Japan	7.0	3.3	7.9	3	12			
Sweden	4.3	3.4	4.8	15	2			
Netherlands	3.5	2.3	3.9	12	11			
Germany	3.2	3.1	3.6	5	13			
France	1.9	2.2	2.1	(^d)	19			
South Korea	1.0	4.9	1.1	4	1			
China ^e	0.5	2.4	0.6	2	48			

 Table 2.6 Research and Development (R&D) statistics, 2021, selected countries

 In percentages and global rank

Sources: OECD, "Main Science and Technology Indicators," March 2024; World Bank Group, "Research and Development Expenditure (% of GDP)," September 30, 2024; "Researchers in R&D (per Million People)," September 30, 2024; USDOC, BEA, table 2.2, "U.S. Trade in Services, by Type of Service and by Country or Affiliation," July 3, 2024; "U.S. Direct Investment Abroad," August 23, 2024.

^a This measure provides a breakdown of the destination of U.S. R&D services exports and includes both cross-border R&D services, as reported by the BEA under other business services, and R&D-related categories classified under charges for the use of intellectual property.

 $^{\rm b}$ Domestic spending includes spending by governments as well as the private sector.

^c Business spending excludes spending by governments.

^d Data not available.

^e Excluding Hong Kong and Macao.

¹⁵⁴ Asen, "Patent Box Regimes in Europe, 2021," September 9, 2021.

¹⁵⁵ Neubig et al., "Fiscal Incentives for R&D and Innovation in a Diverse World," September 13, 2016, 11.

¹⁵⁶ Neubig et al., "Fiscal Incentives for R&D and Innovation in a Diverse World," September 13, 2016, 9, 13.

The countries that are the destination of the highest amount of U.S. exports in R&D services have notably favorable IP boxes or similar incentives. The top five importers of U.S. R&D services—Ireland, Switzerland, Japan, Sweden, and the Netherlands—have implemented various IP regimes to promote innovation and attract R&D-intensive businesses.¹⁵⁷ Ireland's Knowledge Development Box stands out with a favorable effective tax rate of 6.25 percent on income from qualifying IP. It focuses heavily on patents and software copyrights developed through R&D activities performed within Ireland, ensuring compliance with OECD nexus rules.¹⁵⁸ Similarly, Switzerland offers competitive tax treatment through IP boxes, with effective rates as low as 8–13 percent, depending on the canton, covering patents and other qualifying IP.¹⁵⁹ Swiss cantons often complement this with R&D super deductions,¹⁶⁰ making the cantons an attractive jurisdiction for multinational companies.¹⁶¹ Japan does not offer an IP box but incentivizes R&D with a tax credit system, which allows companies to deduct a portion of their R&D expenses from corporate taxes.¹⁶² This approach targets local innovation without requiring preferential tax treatment for IP-derived income.¹⁶³ Sweden also has no formal IP box but incentivizes R&D by reduced employer social security contributions for R&D staff and deductions for R&D expenses.¹⁶⁴ The Netherlands' Innovation Box offers an effective tax rate of 9 percent for income derived from patents, software, and certain other self-developed IP.¹⁶⁵

Effects of Profit Shifting on Services Trade

Policies that offer tax incentives for R&D may lead to the shifting of profits within multinational enterprises (MNEs). This section focuses on how profit shifting, often motivated by tax strategies, can influence the pattern and measurement of services trade statistics.¹⁶⁶ Profit shifting is defined as the transfer of profits within an MNE from high- to low-tax jurisdictions, as in offshore profit shifting where

¹⁶¹ Grant Thornton, "Tax Breaks for R&D, Switzerland/Liechtenstein," June 17, 2020.

 ¹⁵⁷ USDOC, BEA, table 2.2, "U.S. Trade in Services, by Type of Service and by Country or Affiliation," July 3, 2024; Neubig et al., "Fiscal Incentives for R&D and Innovation in a Diverse World," September 13, 2016, 12.
 ¹⁵⁸ Government of Ireland, Irish Tax and Customs, "Knowledge Development Box," November 2022, 9. The OECD/G20 Inclusive Framework's nexus rules try to address tax issues by setting the fixed returns within a market jurisdiction for commercial activities of multinational enterprises.

¹⁵⁹ Beginning in 2020, Switzerland's patent box regime functions at the cantonal level, which provides a maximum tax base reduction of 90 percent on income from patents and similar rights developed in Switzerland (i.e., at least 10 percent of the "box profits" are subject to ordinary taxation). Cantons can opt for a lower reduction. IPrime Rentsch Kaelin and Balmer-Etienne, "The Swiss Patent-Box," November 3, 2021; "Tax Law Aspects of the Patent Box," November 3, 2021.

¹⁶⁰ R&D super deductions are available in addition to the normal tax deduction. They allow for a maximum of 50 percent of qualified expenses, which include up to 35 percent of the actual expenditure or 80 percent of the invoiced cost of outsourced R&D. Scalemetrics, "Maximise Swiss R&D Tax Benefits," August 2, 2023.

¹⁶² Government of Japan, Ministry of Economy, Trade and Industry, "Overview of Industrial Technology Policy/Innovation Policy," February 6, 2025.

¹⁶³ JETRO, "Incentive Programs," 2025.

¹⁶⁴ Boyanova, "Maximising R&D Incentives," August 21, 2023.

¹⁶⁵ Government of the Netherlands, Netherlands Tax Administration, Belastingdienst, "How to Use the Innovation Box in Your VPB," accessed February 13, 2025.

¹⁶⁶ This topic was also covered in a prior publication of this report; see USITC, *Recent Trends in U.S. Services Trade:* 2021 Annual Report, April 2021, 54–55.

profits that would have accrued in the headquarter country are instead accrued by foreign affiliates.¹⁶⁷ The most common methods of profit shifting are through intra-firm transfer pricing: (1) of goods and services when higher-tax entities export at lower prices to and import at higher prices from lower-tax entities; (2) of loans where higher tax entities borrow money from lower-tax entities at relatively high rates; and (3) of intangible asset movements at prices below true value from high-tax entities to lower-tax entities where profits are eventually earned.¹⁶⁸

Several examples illustrate these various profit shifting mechanisms across industries. In the pharmaceutical industry, it has been reported that although Pfizer's foreign affiliates paid the U.S. parent for licenses for the use of patents, the parent imported manufactured drugs from its affiliates at higher relative prices.¹⁶⁹ In this scenario, the profits were accrued to foreign affiliates and likely overstated manufacturing imports while understating services exports (charges for the use of IP from the parent to the affiliates).¹⁷⁰ Relatedly, through Apple's documented cost-sharing agreement, its Irish subsidiaries held the rights to Apple's IP for the manufacture and sale of products outside of the Americas.¹⁷¹ This structure likely underestimated R&D services exports related to activities aimed at expanding knowledge (if payments made to Apple to help fund R&D as part of the agreement were undervalued) and licenses for the use of outcomes of R&D (given profits earned from IP assets were not realized as U.S. exports and instead were accrued as profits by the foreign affiliate).¹⁷²

A 2022 paper analyzed patterns of such offshore profit shifting by U.S. MNEs and how it has affected key economic measurements, including trade flows.¹⁷³ To illustrate the potential extent and impacts of profit shifting by U.S. MNEs, the authors used firm-level data from the BEA, covering MNE transactions between 1982 and 2016, and a model of profit-maximizing firms to redistribute reported profits among

¹⁷⁰ On the other hand, if the elimination of this profit shifting were to drive drug production back to the United States, then, with profit shifting, R&D exports in the form of charges for the use of IP were overestimated (or, rather, generated by the profit shifting). Setser, "When the Services Trade Data Tells You," April 22, 2020.

¹⁶⁷ Guvenen et al., "Offshore Profit Shifting and Aggregate Measurement," 2022, 1848; Wier and Zucman, "Global Profit Shifting, 1975–2019," 2022, 3.

¹⁶⁸ Guvenen et al., "Offshore Profit Shifting and Aggregate Measurement," 2022, 1857; Wier and Zucman, "Global Profit Shifting, 1975–2019," 2022, 3.

¹⁶⁹ Bergin and Drawbaugh, "How Pfizer Has Shifted U.S. Profits Overseas for Years," November 16, 2015. Also see Wyden, "Pfizer's Colossal Tax Avoidance," May 12, 2025; Hannon, "This Country Won the Global Tax Game," October 10, 2023; Golle and Niquette, "US-Ireland Trade Deficit at Record on Pre-Tariff Pharma Imports," May 6, 2025.

¹⁷¹ Clifford Chance, *Apple Loses the 13 Billion Tax Ruling State Aid Case*, September 24, 2024, 1; Tax Notes, "Levin, McCain Examine Apple's Use of Irish Subsidiaries for Tax Avoidance," May 21, 2013; Levin and McCain, "Offshore Profit Shifting and the U.S. Tax Code - Part 2 (Apple Inc.)," May 21, 2013.

¹⁷² Guvenen et al. provide a hypothetical example using Apple's iPhone. Assuming that intangible assets are created in the United States and materials and labor costs of production, with production occurring abroad, are \$250 per phone and the average selling price is \$750, then profits are \$500 per phone. The authors argue that the entire profit per phone should be attributed to U.S. GDP and included as a net export under "charges for the use of intellectual property." However, this may not be the result when intangible assets are transferred to foreign affiliates. Guvenen et al., "Offshore Profit Shifting and Aggregate Measurement," 2022, 1850, 1857. Also see Jenniges et al., "Strategic Movement of Intellectual Property within US Multinational Enterprises," 2023, 215–16.

U.S. parent companies and their foreign affiliates. The authors found that profit shifting appeared to be highly concentrated in certain low-tax jurisdictions and certain industries:

- Applying its model to geography, the study identified that most reattributed earnings originated from low-tax jurisdictions, including the Netherlands, Bermuda, Ireland, Luxembourg, Switzerland, and Singapore.¹⁷⁴
- Additionally, profit shifting was likely highly concentrated in R&D-intensive industries, particularly in the electronics and computer manufacturing, chemical manufacturing (pharmaceuticals and petroleum), and information technology sectors. In 2016, these industries accounted for approximately 79 percent of the total adjustment estimated in the model.¹⁷⁵

The pattern of concentration of profit shifting in R&D intensive industries forms the basis of the article's assumption that profit shifting had occurred through intangible asset allocation, and the authors provide estimates of how movement of IP affects key economic measures. Between 1982 and 2016, an average of 38 percent of income reported as U.S. direct investment abroad was likely related to intangible assets created in the United States and, the authors argue, should be reattributed to U.S. GDP.¹⁷⁶ Assuming the form of profit shifting in the model estimates, adjusting for profit shifting also reduces the U.S. trade deficit by increasing the services trade surplus (but does not change the goods trade deficit): In 2016, the unadjusted services trade balance was 1.4 percent of GDP, whereas the adjusted balance reached 2.2 percent. The model outputs also illustrate that correcting for profit shifting lowers the return on U.S. foreign direct investment abroad, influences labor productivity, and reduces the labor share of income.¹⁷⁷

Outlook

Industry analysts expect U.S. R&D spending to continue to accelerate in advanced technology sectors over the next five years, driven by an anticipated combination of rising corporate profits and lower interest rates.¹⁷⁸ Maintaining strong IP protections is a key success factor for the United States to hold its global leadership in the R&D industry.¹⁷⁹ Industry analysts expect that foreign competition, especially from China, could intensify as foreign economies expand their own R&D investments.¹⁸⁰

¹⁷⁴ Guvenen et al., "Offshore Profit Shifting and Aggregate Measurement," 2022, 1864, table 2.

¹⁷⁵ Guvenen et al., "Offshore Profit Shifting and Aggregate Measurement," 2022, 1867–68, table 4.

¹⁷⁶ Guvenen et al., "Offshore Profit Shifting and Aggregate Measurement," 2022, 1848–49.

¹⁷⁷ Guvenen et al., "Offshore Profit Shifting and Aggregate Measurement," 2022, 1868–78, figure 5.

¹⁷⁸ IBISWorld, *Scientific Research & Development in the US*, October 2024, 9.

¹⁷⁹ IBISWorld, *Scientific Research & Development in the US*, October 2024, 21.

¹⁸⁰ Boroush and Guci, "Research and Development," April 28, 2022, 67.

Bibliography

- Adolphus, Emell, and Jonathan Keller. "The Top 225 International Design Firms: Firms Step Up Value in Engineering." *Engineering News Record*, September 2, 2024. <u>https://www.enr.com/</u>.
- Adolphus, Emell, Peter Reina, and Jonathan Keller. "The Top 225 International Design Firms: Resilient Design Demands Grow." *Engineering News Record*, August 7/14, 2024. <u>https://www.enr.com/</u>.
- American Lawyer. "The 2024 Global 200 Ranked by Head Count." September 16, 2024. <u>https://www.law.com/international-edition/2024/09/16/the-2024-global-200-ranked-by-head-count/</u>.
- American Lawyer. "The 2024 Global 200 Ranked by Revenue." September 17, 2024. <u>https://www.law.com/international-edition/2024/09/16/the-2024-global-200-ranked-by-revenue/</u>.
- American Marketing Association. "Marketing vs. Advertising." Accessed January 15, 2025. https://www.ama.org/marketing-vs-advertising/.
- Armstrong, Madeline. "Why Fewer International Are Coming to the US." *The Nation*, December 13, 2023. <u>https://www.thenation.com/article/society/fewer-international-students-us-h1b-visa/</u>.
- Asen, Elke. "Patent Box Regimes in Europe, 2021." Tax Foundation, September 9, 2021. https://taxfoundation.org/data/all/eu/patent-box-regimes-europe-2021/.
- Barnes Reports. *Colleges & Universities Industry (NAICS 61131)*. 2023. https://www.barnesreports.com/education.
- BDO. "2024 Global Financial Results." December 18, 2024. <u>https://www.bdo.global/en-gb/news/2024/2024-global-financial-results</u>.
- Bergin, Tom, and Kevin Drawbaugh. "How Pfizer Has Shifted U.S. Profits Overseas for Years." Reuters, November 16, 2015. <u>https://www.reuters.com/article/business/how-pfizer-has-shifted-us-profits-overseas-for-years-idUSKCN0T51ZS/</u>.
- Blanco, Maurice, Stephen A. Byeff, Hillary A. Coleman, et al. "Financial Statements Required for Securities Offerings by U.S. Companies." Davis Polk & Wardwell, July 1, 2024. <u>https://www.davispolk.com/insights/client-update/financial-statements-required-securities-offerings-us-companies</u>.
- Blau, Gavin. Engineering Services in the US—Market Size (2004–2031). IBISWorld Industry Report No. 54133. July 2016. <u>https://www.ibisworld.com/industry-statistics/market-size/engineering-services-united-states/</u>.
- Boroush, Mark, and Ledia Guci. *Research and Development: U.S. Trends and International Comparisons*. National Science Foundation, May 21, 2024. <u>https://ncses.nsf.gov/pubs/nsb20225</u>.

- Bouchrika, Imed. "60 University Statistics: 2025 Data, Trends & Predictions." Research.com, January 8, 2025. <u>https://research.com/universities-colleges/university-statistics</u>.
- Boulay Group. "Guide to Public Company Auditing." February 2021. <u>https://boulaygroup.com/wp-content/uploads/2021/02/Guide-to-Public-Company-Auditing.pdf</u>.
- Boyanova, Milena. "Maximising R&D Incentives: Swedish Government Raises Deduction Threshold." *Aspia Group (blog).* August 21, 2023. <u>https://kunskap.aspia.se/blogg/maximising-rd-incentives-</u> <u>swedish-government-raises-deduction-threshold</u>.
- Bruner, Jennifer, and Alexis Grimm. "A Profile of U.S. Services Traders, 2006–2022." U.S. Department of Commerce (USDOC). Bureau of Economic Analysis (BEA). Survey of Current Business, May 31, 2024. <u>https://apps.bea.gov/scb/issues/2024/05-may/0524-profile-services-traders.htm</u>.
- Caines, Daniel. "Robust Growth for KPMG as Global Revenues Rise 5.1% to US\$ 38.4 Billion." KPMG. Press release, December 17, 2024. <u>https://kpmg.com/xx/en/media/press-</u> <u>releases/2024/12/robust-growth-for-kpmg-as-global-revenues-rise-5-percent-to-us-dollar-38-point-4-billion.html</u>.
- Candy, Matthew, Blaine Dolph, Salima Lin, and Jacob Dencik. "Consulting Reimagined, Powered by AI." IBM, October 2024. <u>https://www.ibm.com/thought-leadership/institute-business-value/en-us/report/consulting-ai</u>.
- Citi Wealth and Hildebrandt Consulting. 2025 Citi Hildebrandt Client Advisory Reports Law Firm Industry Revenue Growth Driven by Increasing Demand. Press release, December 5, 2024. <u>https://www.citigroup.com/global/news/press-release/2024/2025-citi-hildebrandt-client-advisory-reports-growth-driven-by-increasing-demand</u>.
- Clifford Chance. Apple Loses the 13 Billion Tax Ruling State Aid Case. September 24, 2024. <u>https://www.cliffordchance.com/content/cliffordchance/briefings/2024/09/apple-loses-the-13-billion-tax-ruling-state-aid-case.html</u>.
- Deloitte. "Deloitte 2024 Global Impact Report." September 12, 2024. <u>https://www.deloitte.com/global/en/about/governance/global-impact-report/global-report-our-people.html</u>.
- Fischer, Karin, and Sasha Aslanian. "Fading Beacon." *The Chronicle of Higher Education*, August 2, 2021. https://www.chronicle.com/article/fading-beacon.
- Foley, Stephen. "US Accounting Firms Rethink Global Networks." *The Financial Times*, July 29, 2024. <u>https://www.ft.com/content/64447d4e-96d4-4b5b-84ca-efb4ea18c05c</u>.
- Foley, Stephen, and Michael O'Dwyer. "Deloitte Revenues Hit Record on Back of Tech Consulting Boom." *The Financial Times*, September 8, 2022. <u>https://www.ft.com/content/71d455df-09a6-4cd0b190-fbe22271a072</u>.
- Golle, Vince, and Mark Niquette. "US-Ireland Trade Deficit at Record on Pre-Tariff Pharma Imports." Bloomberg, May 6, 2025. <u>https://www.bloomberg.com/news/articles/2025-05-06/us-ireland-trade-deficit-at-record-on-pre-tariff-pharma-imports</u>.

- Govdysh, Alexander. *Professional, Scientific and Technical Services in the US*. IBISWorld, September 2024. <u>https://www.ibisworld.com/united-states/industry/professional-scientific-and-technical-services/1386/</u>.
- Government of Ireland. Irish Tax and Customs. "Knowledge Development Box (KDB)." Revenue Knowledge Development Box (KDB), November 18, 2024. <u>https://www.revenue.ie/en/companies-and-charities/reliefs-and-exemptions/knowledgedevelopment-box-kdb/index.aspx.</u>
- Government of Japan. Ministry of Economy, Trade and Industry. "Overview of Industrial Technology Policy/Innovation Policy." February 6, 2025. <u>https://www.meti.go.jp/english/policy/economy/industrial_technology/</u>.
- Government of the Netherlands. Netherlands Tax Administration, Belastingdienst. "How to Use the Innovation Box in Your VPB." Accessed February 13, 2025. <u>https://business.gov.nl/finance-and-taxes/deductibles-and-schemes/how-to-use-the-innovation-box/</u>.
- Grant Thornton. "Tax Breaks for R&D." June 17, 2020. <u>https://www.grantthornton.ch/en/insights/tax-breaks-for-r-d/</u>.
- Grosso, Massimo Geloso, Hildegunn Kyvik Nordås, Frédéric Gonzales, et al. *Services Trade Restrictiveness Index (STRI): Legal and Accounting Services*. Organisation for Economic Co-operation and Development. OECD Trade Policy Papers No. 171. November 4, 2014. <u>https://www.oecd.org/en/publications/services-trade-restrictiveness-index-stri-legal-and-accounting-services_5jxt4nkg9g24-en.html</u>.
- Guvenen, Fatih, Raymond J. Mataloni, Dylan G. Rassier, and Kim J. Ruhl. "Offshore Profit Shifting and Aggregate Measurement: Balance of Payments, Foreign Investment, Productivity, and the Labor Share." *The American Economic Review* 112, no. 6 (2022): 1848–84. https://www.aeaweb.org/articles?id=10.1257/aer.20190285.
- Hannon, Paul. "This Country Won the Global Tax Game, and Is Swimming in Money." *The Wall Street Journal*, October 10, 2023. <u>https://www.wsj.com/economy/global/this-country-won-the-global-tax-game-and-is-swimming-in-money-57c3c70</u>.
- IBISWorld. Architects in the US. November 2024. <u>https://my.ibisworld.com/us/en/industry/54131/at-a-glance</u>.
- IBISWorld. *Colleges & Universities in the US*. June 2024. <u>https://www.ibisworld.com/united-states/industry/colleges-universities/1970/</u>.
- IBISWorld. Engineering Services in the US. August 2024. https://my.ibisworld.com/us/en/industry/54133/at-a-glance.
- IBISWorld. *Global Accounting Services*. May 2024. <u>https://www.ibisworld.com/global/industry/global-accounting-services/1960/</u>.
- IBISWorld. *Global Architectural Services*. June 2024. <u>https://my.ibisworld.com/gl/en/industry/I6721-gl/at-a-glance</u>.

- IBISWorld. *Global Engineering Services*. May 2024. <u>https://my.ibisworld.com/gl/en/industry/l6722-gl/at-a-glance</u>.
- IBISWorld. *Global Management Consultants*. December 2024. https://www.ibisworld.com/global/industry/global-management-consultants/1950/.
- IBISWorld. *Management Consulting in the US*. August 2024. <u>https://www.ibisworld.com/united-states/market-research-reports/management-consulting-industry/</u>.
- IBISWorld. *Scientific Research & Development in the US*. October 2024. <u>https://www.ibisworld.com/united-states/industry/scientific-research-development/1430/</u>.
- IPrime Rentsch Kaelin and Balmer-Etienne. "Tax Law Aspects of the Patent Box." Patent Box, November 3, 2021. <u>https://www.patent-box.ch/en/index.php?tid=276</u>.
- IPrime Rentsch Kaelin and Balmer-Etienne. "The Swiss Patent-Box." Patent Box, November 3, 2021. https://www.patent-box.ch/en/.
- Irwin-Hunt, Alex. "The Great Competition for International Students." *fDiIntelligence*, March 27, 2024. <u>https://www.fdiintelligence.com/content/data-trends/the-great-competition-for-international-</u> students-83628.
- Jenniges, Derrick, Raymond Mataloni Jr, Sarah Atkinson, and Erin (Yiran) Xin. "Strategic Movement of Intellectual Property within U.S. Multinational Enterprises." I In *Challenges of Globalization in the Measurement of National Accounts*, edited by Nadim Ahmad, Brent Moulton, J. David Richardson, and Peter van de Ven. National Bureau of Economic Research and University of Chicago Press, May 2023. <u>https://www.nber.org/system/files/chapters/c14140/c14140.pdf</u>.
- Levin, Carl, and John McCain. "Offshore Profit Shifting and the U.S. Tax Code—Part 2 (Apple Inc.)." Memorandum to Members of the Permanent Subcommittee on Investigations. May 21, 2013. <u>https://www.taxnotes.com/research/federal/legislative-documents/congressional-news-releases/levin-mccain-examine-apples-use-of-irish-subsidiaries-for-tax/f3fc.</u>
- Lloyd, Rachel. "EY Reports Global Revenue of US\$51.2b for Fiscal Year 2024." EY, October 17, 2024. <u>https://www.ey.com/en_gl/newsroom/2024/10/ey-reports-global-revenue-of-51-point-2-billion-us-dollars-for-fiscal-year-2024</u>.
- Japan External Trade Organization (JETRO). "Incentive Programs." ジェトロ, 2025. <u>https://www.jetro.go.jp/en/invest/support_programs/incentive/</u>.
- Mann, Michael A. "Measuring Trade in Services by Mode of Supply." U.S. Department of Commerce (USDOC). Bureau of Economic Analysis (BEA). Working Paper Series WP2019-7, August 2019. <u>https://www.bea.gov/sites/default/files/papers/WP2019-7_2.pdf</u>.
- MarketLine. Asia-Pacific (APAC) Legal Services Market Summary, Competitive Analysis and Forecast to 2028. February 2024. <u>https://store.marketline.com/report/apac-legal-services-market-analysis/</u>.
- MarketLine. *Europe Legal Services Market Summary, Competitive Analysis and Forecast to 2028*. February 2024. <u>https://store.marketline.com/report/europe-legal-services-market-analysis/</u>.

- MarketLine. *Global Accountancy*. February 2024. <u>https://www.marketresearch.com/MarketLine-v3883/Global-Accountancy-Summary-Competitive-Forecast-36877093/</u>.
- MarketLine. *Global Advertising Market Summary, Competitive Analysis and Forecast to 2028*. April 2024. <u>https://store.marketline.com/report/global-overview-of-advertising-market-analysis/</u>.
- MarketLine. *Global Legal Services Market Summary, Competitive Analysis and Forecast to 2028*. February 2024. <u>https://store.marketline.com/report/global-overview-of-legal-services-market-analysis/</u>.
- MarketLine. North America (NAFTA) Advertising Market Summary, Competitive Analysis and Forecast to 2028. May 2024. <u>https://store.marketline.com/report/nafta-countries-advertising-market-analysis/</u>.
- MarketLine. United States (US) Legal Services Market Summary, Competitive Analysis and Forecast to 2028. February 2024. <u>https://store.marketline.com/report/usa-legal-services-market-analysis/</u>.
- MarketLine. United States (US) Management and Marketing Consultancy Market Summary, Competitive Analysis, and Forecast to 2029. April 2025. <u>https://store.marketline.com/report/usa-</u> management-and-marketing-consultancy-market-analysis/.
- McKinsey & Company. "Social Commerce: The Future of How Consumers Interact with Brands." October 19, 2022. <u>https://www.mckinsey.com/capabilities/growth-marketing-and-sales/ourinsights/social-commerce-the-future-of-how-consumers-interact-with-brands</u>.
- McNally, Victoria. "Global Ad Revenue Cracks The \$1 Trillion Dollar Mark In 2024." AdExchanger, December 8, 2024. <u>https://www.adexchanger.com/marketers/global-ad-revenue-cracks-the-1-trillion-dollar-mark-in-2024/</u>.
- Morea, Stephen. Architects in the US—Market Research Report (2015–2030). IBISWorld Industry Report No. 54131. June 2016. <u>https://www.ibisworld.com/united-states/market-research-</u> <u>reports/architects-industry/</u>.
- Morgan, Glenn, Andrew Sturdy, and Sigrid Quack. *The Globalization of Management Consultancy Firms: Constraints and Limitations*. University of Warwick. Centre for the Study of Globalisation and Regionalisation. CSGR Working Paper No. 168/05, May 2006. <u>https://wrap.warwick.ac.uk/id/eprint/1951/1/WRAP_Morgan_wp16805.pdf</u>.
- Neubig, Thomas, Fernando Galindo-Rueda, Silvia Appelt, Chiara Criscuolo, and Matej Bajgar. *Fiscal Incentives for R&D and Innovation in a Diverse World*. No. 27. Organisation for Economic Cooperation and Development. OECD Working Paper No. 27, September 13, 2016. <u>https://www.oecd.org/en/publications/fiscal-incentives-for-r-d-and-innovation-in-a-diverseworld_5jlr9stckfs0-en.html</u>.
- O'Malley, Matty. "Shifts in Digital Advertising: The Impact of Technology and Evolving Consumer Behavior." IBISWorld, April 16, 2024. <u>https://www.ibisworld.com/blog/shifts-in-digital-advertising/99/1126/</u>.
- Open Doors. "Enrollment Trends." Institute of International Education. Accessed February 3, 2025. https://opendoorsdata.org/data/international-students/enrollment-trends/.

- Organisation for Economic Co-operation and Development (OECD). *Frascati Manual 2015: Guidelines for Collecting and Reporting Data on Research and Experimental Development*. October 8, 2015. <u>https://www.oecd.org/en/publications/2015/10/frascati-manual-2015_g1g57dcb.html</u>.
- Organisation for Economic Co-operation and Development (OECD). "Main Science and Technology Indicators—Business Enterprise Expenditure on R&D (BERD)." March 2024. <u>https://www.oecd.org/en/data/datasets/main-science-and-technology-indicators.html</u>.
- Organisation for Economic Co-operation and Development (OECD). *OECD Services Trade Restrictiveness Index (STRI)—Legal Services 2023*. January 2024. <u>https://www.oecd.org/en/publications/the-costs-of-regulatory-barriers-to-trade-in-services_bae97f98-en.html</u>.
- Organisation for Economic Co-operation and Development (OECD). OECD Services Trade Restrictiveness Index Simulator. Accessed January 27, 2025. <u>https://sim.oecd.org/?lang=En&ds=STRI&d1c=psleg&cs=psleg</u>.
- PKC Management Consulting. "Auditing for International Businesses: Cross Border Challenges." May 29, 2024. <u>https://pkcindia.com/blogs/auditing-for-international-businesses-cross-border-challenges/</u>.
- PricewaterhouseCoopers (PwC). "Digital Ad Spending Moves Closer to the Consumer." Accessed January 13, 2025. <u>https://www.pwc.com/gx/en/industries/tmt/media/outlook/changes-in-digital-advertising-spend.html</u>.
- PricewaterhouseCoopers (PwC). "Financials." October 2024. https://www.pwc.com/gx/en/about/transparency-report/2024/financials.html.
- PricewaterhouseCoopers (PwC). "Perspectives from the Global Entertainment & Media Outlook 2024– 2028." July 16, 2024. <u>https://www.pwc.com/gx/en/issues/business-model-</u> <u>reinvention/outlook/insights-and-perspectives.html</u>.
- Raghunathan, Krishnan. "Five Trends That Will Redefine Finance and Accounting in 2025." Accounting Today, January 7, 2025. <u>https://www.accountingtoday.com/list/five-trends-that-will-redefine-finance-and-accounting-in-2025</u>.
- Scalemetrics. "Maximise Swiss R&D Tax Benefits." August 2, 2023. https://www.startupmetrics.ch/maximise-swiss-rd-tax-benefits/.
- Scottish Financial News. "PwC Announces Record Global Revenues of \$50bn." October 4, 2022. <u>https://www.scottishfinancialnews.com/articles/pwc-announces-record-global-revenues-of-50bn</u>.
- Setser, Brad W. "When the Services Trade Data Tells You More About Tax Avoidance Than About Actual Trade." Council on Foreign Relations. *Follow the Money* (blog). April 22, 2020. <u>https://www.cfr.org/blog/when-services-trade-data-tells-you-more-about-tax-avoidance-about-actual-trade</u>.

- Shorelight. "Global Student Demand Picks Up, US Loses International Student Market Share to Competitors." *Shorelight*, May 22, 2023. <u>https://shorelight.com/news-and-insights/global-</u> student-demand-picks-up-us-loses-international-student-market-share/.
- Skyscrapers World. "The Rise of Thin Skyscrapers: Redefining the New York City Skyline." February 6, 2025. <u>https://skyscrapersworld.com/thin-skyscraper-new-york</u>.
- Statista. Advertising: Market Data & Analysis. November 2024. https://www.statista.com/study/166789/advertising-market-data-and-analysis/.
- Statista. "Advertising: United States." Market Insights, October 2024. https://www.statista.com/outlook/amo/advertising/united-states#ad-spending.
- Statista. "Companies with Highest R&D Spending Worldwide 2022." May 2023. <u>https://www.statista.com/statistics/265645/ranking-of-the-20-companies-with-the-highest-spending-on-research-and-development/</u>.
- Statista. "Research and Development (R&D) Worldwide." 2024. https://www.statista.com/study/70627/research-and-development-worldwide/.
- Statista. "Share of Digital in Advertising Revenue Worldwide from 2023 to 2027." January 10, 2025. https://www.statista.com/statistics/375008/share-digital-ad-spend-worldwide/.
- Sussman, Anna Louie. "Separating Auditing from Consulting: More Complex than It Seems." University of California, Los Angeles. Anderson School of Management. UCLA Anderson Review, July 28, 2021. https://anderson-review.ucla.edu/separating-auditing-from-consulting-more-complex-than-it-seems/.
- Tax Notes. "Levin, McCain Examine Apple's Use of Irish Subsidiaries for Tax Avoidance." May 21, 2013. <u>https://www.taxnotes.com/research/federal/legislative-documents/congressional-news-</u> <u>releases/levin-mccain-examine-apples-use-of-irish-subsidiaries-for-tax/f3fc</u>.
- Thomson Reuters. "Auditing in Accounting: What It Is and How It Works." January 12, 2023. <u>https://tax.thomsonreuters.com/blog/auditing-in-accounting-what-it-is-and-how-it-works/</u>.
- Turner, Lynn E., and Hemming Morse. "Reforms of the Auditing Profession: Improving Quality Transparency, Governance and Accountability." *The Harvard Law School Forum on Corporate Governance*, December 28, 2020. <u>https://corpgov.law.harvard.edu/2020/12/28/reforms-of-the-</u> <u>auditing-profession-improving-quality-transparency-governance-and-accountability/</u>.
- United Nations Conference on Trade and Development (UNCTAD). *Trade and Development Aspects of Professional Services and Regulatory Frameworks*. Trade and Development Board, November 25, 2004. <u>https://unctad.org/system/files/official-document/c1em25d2_en.pdf</u>.
- U.S. Census Bureau (Census). "North American Industry Classification System (NAICS)." 2022. https://www.census.gov/naics/.

- U.S. Census Bureau (Census). "2012 NAICS Definition: 541310 Architectural Services." Accessed January 22, 2025. <u>https://www.census.gov/cgi-bin/sssd/naics/naicsrch?code=541310&search=2012</u> NAICS Search.
- U.S. Department of Commerce (USDOC). Bureau of Economic Analysis (BEA). Form BE-120, 2022 Benchmark Survey of Transactions in Selected Services and Intellectual Property with Foreign Persons. August 2022. https://www.bea.gov/sites/default/files/2023-01/BE-120.pdf.
- U.S. Department of Commerce (USDOC). Bureau of Economic Analysis (BEA). Form BE-125, Quarterly Survey of Transactions in Selected Services and Intellectual Property with Foreign Persons. 2022. https://www.bea.gov/system/files/2022-03/BE-125%20FINAL%202022.pdf.
- U.S. Department of Commerce (USDOC). Bureau of Economic Analysis (BEA). *Guide to Industry Classifications for International Surveys 2022*. BE–799 (REV. 04–2022). 2022. <u>https://www.bea.gov/system/files/2022-04/NAICS-Guide-2022.pdf</u>.
- U.S. Department of Commerce (USDOC). Bureau of Economic Analysis (BEA). "U.S. Direct Investment Abroad, All Majority-Owned Foreign Affiliates (Data for 2009 and Forward), Research and Development Expenditures." August 23, 2024. <u>https://www.bea.gov/international/di1usdop</u>.
- U.S. Department of Commerce (USDOC). Bureau of Economic Analysis (BEA). U.S. International Economic Accounts: Concepts and Methods. June 2024. https://www.bea.gov/resources/methodologies/international/pdf/iea-concepts-methods.pdf.
- U.S. Department of Commerce (USDOC). Bureau of Economic Analysis (BEA). Table 2.1. "U.S. Trade in Services, by Type of Service." July 3, 2024. <u>https://apps.bea.gov/iTable/?ReqID=62&step=1#eyJhcHBpZCI6NjIsInN0ZXBzIjpbMSw5LDZdLCJkY</u> <u>XRhIjpbWyJQcm9kdWN0liwiNCJdLFsiVGFibGVMaXN0liwiMjQ1II1dfQ==</u>.
- U.S. Department of Commerce (USDOC). Bureau of Economic Analysis (BEA). Table 2.2. "U.S. Trade in Services, by Type of Service and by Country or Affiliation." July 3, 2024. <u>https://apps.bea.gov/iTable/?ReqID=62&step=1#eyJhcHBpZCI6NjIsInN0ZXBzIjpbMSw5LDEwLDd</u> <u>dLCJkYXRhIjpbWyJQcm9kdWN0IiwiNCJdLFsiVGFibGVMaXN0IiwiMzA1NjgiXSxbIIRhYmxITGIzdFNIY</u> <u>29uZGFyeSIsIjMwNTY5II1dfQ==</u>.
- U.S. Department of Commerce (USDOC). Bureau of Economic Analysis (BEA). Table 4.1. "Services Supplied to Foreign Persons by U.S. MNEs Through Their MOFAs, by Industry of Affiliate and by Country of Affiliate." October 8, 2024. <u>https://apps.bea.gov/iTable/?ReqID=62&step=1#eyJhcHBpZCI6NjIsInN0ZXBzIjpbMSw5LDZdLCJkY</u> <u>XRhIjpbWyJQcm9kdWN0liwiNCJdLFsiVGFibGVMaXN0liwiMzYzIl1dfQ==</u>.
- U.S. Department of Commerce (USDOC). Bureau of Economic Analysis (BEA). Table 5.1. "Services Supplied to U.S. Persons by Foreign MNEs Through Their MOUSAs, by Industry of Affiliate and by Country of UBO." October 8, 2024. <u>https://apps.bea.gov/iTable/?ReqID=62&step=1#eyJhcHBpZCI6NjIsInN0ZXBzIjpbMSw5LDZdLCJkY</u> <u>XRhIjpbWyJQcm9kdWN0liwiNCJdLFsiVGFibGVMaXN0liwiMzY5II1dfQ==</u>.
- U.S. Department of Labor (USDOL). Bureau of Labor Statistics (BLS). "Management Analysts." Accessed January 27, 2025. <u>https://www.bls.gov/ooh/business-and-financial/management-analysts.htm</u>.

- U.S. International Trade Commission (USITC). *Recent Trends in U.S. Services Trade: 2021 Annual Report*. Publication number 5192. April 2021. <u>https://www.usitc.gov/publications/332/pub5192.pdf</u>.
- Wier, Ludvig, and Gabriel Zucman. Global Profit Shifting, 1975–2019. United Nations University World Institute for Development Economics Research Working Paper (UNU-WIDER). UNU-WIDER, November 2022. https://doi.org/10.35188/UNU-WIDER/2022/254-6.
- Williams, Christopher, and Sander van Triest. "Researching Practice Areas of Consultancy Firms." *Management Consulting Journal* 6, no. 2 (May 24, 2023): 104–10. <u>https://doi.org/10.2478/mcj-2023-0012</u>.
- Wilson, Despina. "Largest Universities in the World by Number of Students Enrollment, 2024." *CEOWORLD Magazine*, January 28, 2024. <u>https://ceoworld.biz/2024/01/28/revealed-largest-universities-in-the-world-by-number-of-students-enrollment-2024/</u>.
- World Bank Group. "Research and Development Expenditure (% of GDP)." World Bank Open Data, September 30, 2024. <u>https://data.worldbank.org</u>.
- World Bank Group. "Researchers in R&D (per Million People)." World Bank Open Data, September 30, 2024. <u>https://data.worldbank.org</u>.
- Wyden, Ron. "Pfizer's Colossal Tax Avoidance: How Pfizer Used "Round Tripping" Scheme to Avoid Billions in Taxes on U.S. Drug Sales." The United States Senate Committee on Finance, March 2025.
 <u>https://www.finance.senate.gov/imo/media/doc/wyden_pfizer_investigation_report_final_marc</u> h 2025pdf.
- Zewe, Adam. "Explained: Generative AI." Massachusetts Institute of Technology. *MIT News*, November 9, 2023. <u>https://news.mit.edu/2023/explained-generative-ai-1109</u>.

Recent Trends in U.S. Services Trade: 2025 Annual Report

This chapter highlights how skills gaps and the introduction of artificial intelligence (AI) technologies are affecting the competitiveness of the suppliers of professional services, specifically in the fields of legal services, architectural and engineering services, and accounting and auditing services.

Professional services firms have traditionally relied heavily on a knowledgeable and highly skilled and credentialed labor force to provide specialized products within various industries. The professional services labor force remains highly skilled, but firms that supply professional business services have increasingly integrated technological advancements such as machine learning (ML) and Al into their product offerings to improve the efficiency and breadth of their products.¹⁸¹ However, an increasing reliance on these technological tools has in part led to a skills gap within the labor force of several professional services industries, many of which are also experiencing labor shortages.¹⁸²

In the field of legal services, lower-priced and higher-tech alternative legal service providers (ALSPs) have proliferated, and traditional law firms have integrated legal technology (legal tech), including AI, into their own offerings.¹⁸³ While the adoption of legal tech improves efficiency and provides competitive advantages (especially to developers of the technology), it may also alter legal services business models and hiring practices.¹⁸⁴ Although U.S. law firms are at the forefront of technology integration, their ability to effectively deliver legal services or maintain their competitive positions is constrained by complex regulatory environments they face in certain foreign markets. Due to a variety of factors, including regulatory issues, U.S. law firms have reduced their commercial presence in

¹⁸¹ Machine learning (ML) refers to statistical algorithms that process and learn from sets of data to generalize unseen data and, thus, can perform tasks without explicit instructions. References to AI in this report refer to the application of advanced algorithms such as ML in professional services products and processes. Generative AI technologies use sophisticated large content machine learning models to generate high quality original output such as text and images. Martineau, "What Is Generative AI?," April 20, 2023. Columbia University, "Artificial Intelligence (AI) vs. Machine Learning," accessed April 16, 2024; Zewe, "Explained: Generative AI," November 9, 2023.
¹⁸² SAPRO, "The Accounting Skills Shortage," May 6, 2024; Kennedy, "The Accounting Shortage Is Hurting Your Bottom Line," August 12, 2024; Adolphus and Keller, "The Top 500 Design Firms," April 29, 2024, 64–65, 68.
¹⁸³ Statista, *Legal Tech*, 2023, 11, 15, 17; International Bar Association and the Center for AI and Digital Policy, *The Future Is Now*, September 2024, 8.

¹⁸⁴ Wolters Kluwer, *The 2024 Wolters Kluwer Future Ready Lawyer Report*, 2024, 16–17; International Bar Association and the Center for AI and Digital Policy, *The Future Is Now*, September 2024, 14, 16; Wirth, "Law Firm COO & CFO Forum," November 7, 2024; Davis, "The Future of Law Firms (and Lawyers)," October 2, 2020, 8.

China.¹⁸⁵ Relatedly, the share of U.S. cross-border exports and foreign affiliate sales to the Asia-Pacific region declined in the latest 10-year period for which data are available.¹⁸⁶

In architectural and engineering services, rising labor and input costs have strained firm performance and led to the adoption of high-tech solutions in the traditionally conservative industry.¹⁸⁷ Many architectural and engineering firms have begun to digitalize their operations and devote resources to implementing AI to automate tasks, optimize designs, and improve decision-making. Architectural and engineering firms are also incorporating tools such as building information modeling and digital twins, which create detailed 3D models of projects.

In accounting and auditing services, firms continue to digitalize and use robotic process automation (RPA) and cloud services automation to improve their product offerings and the efficiency of their operations.¹⁸⁸ These firms are also increasing their adoption of AI to enhance automated processes and better organize and analyze their clients' information. The integration of AI and continuing digitalization of this industry has led accounting and auditing firms to update their recruiting efforts to target employees that possess AI expertise and skills such as critical thinking and adaptability to identify errors in AI model output.¹⁸⁹

Legal Services: U.S. Law Firms Remain Competitive Globally Although Market Conditions and Regulatory Barriers Impede Presence in China

As discussed in chapter 2, the United States maintains a surplus in legal services trade, and U.S. crossborder exports and sales of legal services by U.S.-owned foreign affiliates have shown strong growth in recent years. U.S. firms supplying legal services in foreign markets are at the forefront of integrating new technologies—including AI technologies—into their business practices to increase efficiency and maintain or enhance their competitive positions. At the same time, they are facing increasingly complex regulatory environments in certain countries, which makes it difficult to be competitive in those markets.

AI Adoption and Impacts

Reports indicate that growth of legal technology (legal tech), which generally refers to the legal industry's use of software and other tools, processes, and solutions to improve the provision of legal services, has

- ¹⁸⁶ USDOC, BEA, table 4.1, "U.S. Trade in Services, by Country or Affiliation and by Type of Service," July 3, 2024; USDOC, BEA, table 4.1, "Services Supplied to Foreign Persons by U.S. MNEs," October 8, 2024.
- ¹⁸⁷ Clarke, "The Effects of Inflation on the AEC Industry," May 16, 2024; IBISWorld, *Architects in the US*, November 2024, 63; IBISWorld, *Engineering Services in the US*, August 2024, 7.
- ¹⁸⁸ Ayinla et al., "The Role of Robotic Process Automation (RPA)," February 17, 2024, 431–32; Abogadie, "How Cloud Computing Accounting Is Changing the Industry," May 14, 2024.
- ¹⁸⁹ Sahota, "The Dawn of a New Era," April 22, 2024.

¹⁸⁵ Mok, "Flight or Fight?," October 7, 2024.

been significant in recent years.¹⁹⁰ The proliferation of legal tech has altered the legal services industry landscape through the growth of lower priced and higher tech alternative legal services providers, legal tech companies, and client use of legal technology.¹⁹¹ In turn, as a result of competitive pressure and client demands, law firms have increasingly adopted legal tech and, in particular, surveys show that the use of Al tools by law firms has gained momentum.¹⁹² Al adoption provides competitive advantages to law firms integrating the technology, allowing firms to supply legal services at a greater scale and with enhanced efficiency.¹⁹³ Al adoption also has the potential to alter business models and related practices, including changes to pricing structures and greater demand for hiring attorneys with Al training and non-lawyer tech staff.¹⁹⁴

Growth of Legal Tech Companies and Alternative Legal Services Providers

Within the legal services industry, legal technology is typically provided directly by companies to law firms and corporate legal departments. Legal tech companies are considered a subset of these companies, but a broader group referred to as alternative legal service providers (ALSPs) offers technology-based services as well as certain types of legal services as an alternative to traditional law firms.¹⁹⁵ Reported value, growth, and service offerings of legal tech companies and ALSPs are summarized below; however, their distinctions and boundaries with law firms are increasingly unclear.¹⁹⁶

Although estimates of the legal tech market value vary depending on the methodology employed and years covered, sources agree on strong growth potential going forward.¹⁹⁷ Relatedly, throughout 2024, reporting indicates notable growth of legal tech startups and investment in legal tech, including by law firms, with demand for generative AI (GenAI) underpinning such investments.¹⁹⁸ Legal tech companies primarily operate in the business-to-business sphere and provide a range of software, including for client

¹⁹¹ Langemo, "Transforming the Legal Profession Through Technology and Entrepreneurship," May 23, 2024, 4–5.
 ¹⁹² GenAl is also discussed in the accounting and auditing section below. Statista, *Legal Tech*, 2023, 11–15;
 International Par Association and the Conter for AL and Digital Policy. *The Enture Ic New*. Sentember 2024. 8

¹⁹⁴ International Bar Association and the Center for AI and Digital Policy, *The Future Is Now*, September 2024, 8.
 ¹⁹⁵ Thomson Reuters Institute, the Center on Ethics and the Legal Profession at Georgetown Law, and the Saïd Business School at the University of Oxford, *Alternative Legal Services Providers 2023*, 2023, 2; Statista, *Legal Tech*, 2023, 5, 8; Business Research Company, *Legal Technology Global Market Report 2025*, January 2025; Shaybak, "Legal Tech v. Legal Outsourcing and ALSP," May 10, 2022.

¹⁹⁰ Statista, *Legal Tech*, 2023, 3; Business Research Company, *Legal Technology Global Market Report 2025*, January 2025.

International Bar Association and the Center for AI and Digital Policy, *The Future Is Now*, September 2024, 8. ¹⁹³ *Thomson Reuters Law Blog*, "Impact of AI on Law Firms of Every Size," August 15, 2023.

¹⁹⁶ Thomson Reuters Institute, the Center on Ethics and the Legal Profession at Georgetown Law, and the Saïd Business School at the University of Oxford, *Alternative Legal Services Providers 2023*, 2023, 2.

¹⁹⁷ For example, one source estimates the global market value at \$27.6 billion in 2021 and projects the market value will reach \$35.6 billion in 2027 while another source estimates global value at \$30.3 billion in 2024 and projects a market value of \$47.6 billion in 2029; also, the methodology for market value is unclear in relation to the inclusion of ALSPs. See Statista, *Legal Tech*, 2023, 6, 40; Business Research Company, *Legal Technology Global Market Report 2025*, January 2025.

¹⁹⁸ Sherman, "Legal Tech Startups That Law Firms Invested in Throughout 2024," December 18, 2024; Black, "The Legal Tech Trends That Defined 2024," December 17, 2024.

management, case management, time/calendar/expense/workflow tracking and analysis, document automation, billing, and finances.¹⁹⁹ The business-to-business segment is expected to account for around 90 percent of the North American and European legal tech market in 2027.²⁰⁰

In the United States, the United Kingdom (UK) and other European countries, Canada, and Australia, a report of the ALSP market shows it has grown from an estimated \$8.5 billion in 2015 to \$28.5 billion in 2023.²⁰¹ ALSPs in the report include independent companies such as Axiom, those created by and operating within law firms—for example, Re:link (Linklaters)—and the Big Four accounting firms.²⁰² Throughout 2015–23, the largest category was independent ALSPs and the fastest-growing category was ALSPs operating within law firms. Both law firms and corporate law departments report using ALSPs for various services and support, such as electronic discovery services and legal research.²⁰³ Notably, in 2022, consulting on legal technology (including outsourcing technology support and training in technology) emerged as a top service ALSPs supplied to global law firms.²⁰⁴ Furthermore, among the top services supplied by ALSPs to law firms, the need for access to technology not available internally was of increasing importance, while access to specialized expertise and meeting demand without staff increases continued in importance.²⁰⁵

AI Adoption by Law Firms and Its Effects

In addition to working with ALSPs and legal tech companies, law firms are also adopting AI and GenAI tools internally. AI improves productivity and the ability to advise clients more quickly. These applications facilitate faster and more efficient legal research and analysis of legal documents and data, identifying additional and relevant content, predicting case outcomes, and understanding legal questions.²⁰⁶ Firms that develop and integrate AI technologies are considered to be in stronger competitive positions going forward.²⁰⁷ At the same time, many have outlined the challenges and risks posed by AI use and

¹⁹⁹ Statista, *Legal Tech*, 2023, 5,8. For examples of legal tech companies and service offerings, see Clio, "Products," accessed February 3, 2025; Mycase, "Industry-Leading Legal Case Management Software; and Solutions," accessed February 3, 2025.

²⁰⁰ Statista, Legal Tech, 2023, 8.

²⁰¹ Thomson Reuters Institute, the Center on Ethics and the Legal Profession at Georgetown Law, and the Saïd Business School at the University of Oxford, *Alternative Legal Services Providers 2025*, 2025, 5, 29. These estimates appear to be based on survey responses from law firms and corporate law departments in the United States, the UK and other European countries, Canada, and Australia.

²⁰² Thomson Reuters Institute, the Center on Ethics and the Legal Profession at Georgetown Law, and the Saïd Business School at the University of Oxford, *Alternative Legal Services Providers 2025*, 2025, 5, 29.

²⁰³ Thomson Reuters Institute, the Center on Ethics and the Legal Profession at Georgetown Law, and the Saïd Business School at the University of Oxford, *Alternative Legal Services Providers 2023*, 2023, 5–11.

²⁰⁴ Thomson Reuters Institute, the Center on Ethics and the Legal Profession at Georgetown Law, and the Saïd Business School at the University of Oxford, *Alternative Legal Services Providers 2023*, 2023, 5–6.

²⁰⁵ Thomson Reuters Institute, the Center on Ethics and the Legal Profession at Georgetown Law, and the Saïd Business School at the University of Oxford, *Alternative Legal Services Providers 2023*, 2023, 12.

²⁰⁶ Bloomberg Law, "The Real Impact of AI in Legal Research," November 21, 2023.

²⁰⁷ Smaller firms may be a competitive disadvantage going forward, owing to a gap in technological capabilities with larger firms. International Bar Association and the Center for AI and Digital Policy, *The Future Is Now*, September 2024, 16. However, for smaller and mid-size firms with less staff, resulting efficiencies from adopting AI can help

integration.²⁰⁸ Ethical issues include obligations to provide competent representation (e.g., reviewing GenAl output for inaccuracy) and to protect client confidentiality (e.g., evaluating risks of unauthorized access to client-related information when using GenAl tools).²⁰⁹

Recent surveys report on AI usage in the legal services industry and at least one which differentiates results shows that AI adoption rates differ among firms, for example across geographies and size of firm. According to an International Bar Association (IBA) survey conducted in April and May 2024, adoption of AI by law firms is widespread, with 63 percent of respondents reporting using AI.²¹⁰ Questionnaire findings point to a greater and more advanced integration of AI in firms based in the United States and Europe (including the UK) that have a presence in other countries.²¹¹ This result correlates with a distinction in AI implementation by firms of different sizes, with a majority of smaller firms not yet using AI.²¹²

Surveys also report on specific uses and types of AI. According to the 2024 IBA survey, the most prevalent use of AI within firms is for tasks that include business development and marketing, although a higher share of larger law firms report client-facing uses of AI (such as legal research and contract drafting).²¹³ Such uses appear to be associated with GenAI, which is reported to be in the "testing and development" stage at most law firms.²¹⁴ According to another 2024 survey, law firms and corporate law departments in the United States using or planning to use GenAI responded that top uses of the technology included legal research and document review.²¹⁵

compete with larger staffed firms, see *Thomson Reuters Law Blog*, "Impact of AI on Law Firms of Every Size," August 15, 2023.

 ²⁰⁸ International Bar Association and the Center for AI and Digital Policy, *The Future Is Now*, September 2024, 8, 13, 17; Bloomberg Law, "The Real Impact of AI in Legal Research," November 21, 2023; Davis, "The Future of Law Firms (and Lawyers)," October 2, 2020, 9–12.

²⁰⁹ American Bar Association, Formal Opinion 512: Generative Artificial Intelligence Tools, July 29, 2024, 2–7.

²¹⁰ International Bar Association and the Center for AI and Digital Policy, *The Future Is Now*, September 2024, 11– 12, 34, 43. According to a different survey of law firms and corporate legal departments in the United States and select countries in the beginning of 2024, about a quarter indicated currently using or planning to use GenAI, see Thomson Reuters, *2024 Generative AI in Professional Services*, February 27, 2024, 13. Another survey of law firm and corporate legal departments in the United States and nine European countries reports daily and weekly GenAI usage (with about a third reporting daily use). Wolters Kluwer, *The 2024 Wolters Kluwer Future Ready Lawyer Report*, 2024, 4, 18. Also see Bloomberg Law, *Legal Ops and Tech Survey*, 2024.

²¹¹ International Bar Association and the Center for AI and Digital Policy, *The Future Is Now*, September 2024, 11– 12. For information on challenges in adopting AI that global firms' offices in Asia experience relative to offices in the United States and Europe, see Teja, "Global Firms' Asia Offices Lag Behind US and Europe in AI Uptake," August 26, 2024.

²¹² International Bar Association and the Center for AI and Digital Policy, *The Future Is Now*, September 2024, 11–
12. See this report for definitions of law firm size and AI implementation by firm size.

²¹³ International Bar Association and the Center for AI and Digital Policy, *The Future Is Now*, September 2024, 8, 12. ²¹⁴ International Bar Association and the Center for AI and Digital Policy, *The Future Is Now*, September 2024, 8, 13. For more information on AI applications designed for contract review, legal research, and the prediction of legal outcomes, see Handa and Papineau-Wolff, "Law & AI," November 7, 2019. Also see Davis, "The Future of Law Firms (and Lawyers)," October 2, 2020, 3–5.

²¹⁵ Thomson Reuters, *2024 Generative AI in Professional Services*, February 27, 2024, 16.

According to many sources, AI adoption will likely result in changes to law firm structures, business models, and hiring.²¹⁶ Analysis suggests the model of billing will be defined less by time and more by the value of services provided by attorneys as they work in complementary ways with new technologies.²¹⁷ In turn, survey responses of the legal industry point to expectations that AI adoption will result in shifting away from the billable hour model and moving toward fixed or value-added fees.²¹⁸

Additionally, law firm survey responses indicate that certain roles will be replaced by AI, with younger lawyers and junior associates expected to be most significantly affected.²¹⁹ Many reports suggest that firms have or are likely to shift their staffing demand away from staff who handle more routine matters toward specialists in certain areas (including AI, legal tech, and data analysis) and attorneys with AI-training.²²⁰ Because of these expected changes, the emphasis is on training to help manage risks and integrate AI into law firms, as well as to develop the legal expertise of junior attorneys.²²¹

Declining U.S. Law Firm Market Presence and Exports of Legal Services to China Reflect Difficult Market and Regulatory Conditions

Although U.S. law firms have adapted their business models using new technologies to be more costeffective and responsive to client needs, as discussed above, their ability to supply legal services competitively in a foreign market also depends on the regulatory and other market conditions they encounter. In recent years, foreign-owned law firms have reduced or withdrawn their commercial presence in China; such activity accelerated for U.S. law firms in 2024.²²² This decline in U.S. law firm

²¹⁶ International Bar Association and the Center for AI and Digital Policy, *The Future Is Now*, September 2024, 8–9. For effects of AI adoption on firms of different sizes, see *Thomson Reuters Law Blog*, "Impact of AI on Law Firms of Every Size," August 15, 2023.

²¹⁷ Davis, "The Future of Law Firms (and Lawyers)," October 2, 2020, 8–9.

²¹⁸ Wolters Kluwer, *The 2024 Wolters Kluwer Future Ready Lawyer Report*, 2024, 4, 16; International Bar Association and the Center for AI and Digital Policy, *The Future Is Now*, September 2024, 14. Also see Thomson Reuters, *2024 Generative AI in Professional Services*, February 27, 2024, 21.

²¹⁹ International Bar Association and the Center for AI and Digital Policy, *The Future Is Now*, September 2024, 15. For more information on how technology is changing roles and recruitment in law firms, see Bullows, "How Technology Is Changing the Legal Sector," April 3, 2021.

²²⁰ Wolters Kluwer, *The 2024 Wolters Kluwer Future Ready Lawyer Report*, 2024, 16–17; International Bar Association and the Center for AI and Digital Policy, *The Future Is Now*, September 2024, 14; Wirth, "Law Firm COO & CFO Forum," November 7, 2024; Davis, "The Future of Law Firms (and Lawyers)," October 2, 2020, 8. Other reporting indicates that AI applications have the potential for replacing entry-level paralegal work, see Bloomberg Law, "How Is AI Changing the Legal Profession?," May 23, 2024. For information on firms past efforts to add technology staff and data scientists, see Citi Wealth and Hildebrandt Consulting, *2024 Citi Hildebrandt Client Advisory*, December 6, 2023, 21. Also see Thomson Reuters, *2024 Generative AI in Professional Services*, February 27, 2024, 25–26.

²²¹ International Bar Association and the Center for AI and Digital Policy, *The Future Is Now*, September 2024, 8–9, 15–16.

²²² Mok, "Flight or Fight?," October 7, 2024; Merken, "US Legal Industry's China Retreat Gained Steam in 2024," December 26, 2024.

market presence—due to a variety of factors, including regulatory issues—is starting to result in declining trade or regional trade shares.

Throughout 2013–23 (for cross-border exports) and 2012–22 (for U.S. affiliate sales), the top regional destinations for legal services trade were Europe and the Asia-Pacific region.²²³ Although legal services exports to Europe as a share of such exports to all countries increased from 47.5 percent in 2013 to 50.1 percent in 2023, the share to the Asia-Pacific region declined from 31.7 percent to 24.8 percent over the same years. Similarly, although affiliate sales to Europe as a share of such sales to all countries increased from 73.8 percent in 2012 to 75.0 percent in 2022, the share to the Asia-Pacific region declined from 22.9 percent to 21.0 percent over the same years.

Notably, cross-border exports of legal services to certain markets in the Asia-Pacific region decreased from 2022 to 2023, including to China. U.S. legal services exports to China had increased since 2013 but dropped from a peak in 2021.²²⁴ Potentially because of lags in reporting data, this recent change is not evident in the data for legal services affiliate sales to China, although U.S. legal services affiliate sales to China as a share of U.S. legal services affiliate sales to the Asia-Pacific region has declined from 20.9 percent in 2012 to 17.3 percent in 2022.²²⁵

During 2024, 35 of 60 U.S. law firms either reduced the number of attorneys in or closed their Beijing or Shanghai offices as well as reduced Hong Kong operations.²²⁶ Several factors were cited as underlying this trend, including the complex business and regulatory environment in China, weaker demand as China's economy has softened, and growing local competition.²²⁷ Data for 2023 also showed declining profits for more than half of China's largest domestic law firms in that year.²²⁸

Going forward, China is expected to continue to be a challenging market for U.S. law firms.²²⁹ The UK, Singapore, and the United Arab Emirates are likely to expand investment and commercial presence.²³⁰ As noted in chapter 2, however, many countries maintain barriers to trade in legal services, and the sector has a relatively high level of restrictiveness.²³¹ For example, foreign lawyers or foreign law firms have historically not been permitted to open offices in India. In 2023, the Bar Council of India announced that foreign law firms and foreign lawyers would be permitted to establish a presence and practice certain areas of law on the basis of reciprocity, although such liberalization is uncertain given subsequent

²²³ For all source data for this paragraph, see USDOC, BEA, table 2.3, "U.S. Trade in Services, by Country or Affiliation and by Type of Service," July 3, 2024; USDOC, BEA, table 4.1, "Services Supplied to Foreign Persons by U.S. MNEs," October 8, 2024.

 ²²⁴ USDOC, BEA, table 2.3, "U.S. Trade in Services, by Country or Affiliation and by Type of Service," July 3, 2024.
 ²²⁵ USDOC, BEA, table 4.1, "Services Supplied to Foreign Persons by U.S. MNEs," October 8, 2024.

²²⁶ The reported period was between December 2023 and December 2024 and included the count of largest U.S. law firms with a presence in China but not announced closures; see Merken, "US Legal Industry's China Retreat Gained Steam in 2024," December 26, 2024.

²²⁷ Roemer, "US Law Firms Are Closing Their China Offices," October 1, 2024; Seah, "Chinese Law Firms Are in Trouble as Downward Spiral Show No Signs of Easing," September 16, 2024.

 ²²⁸ Seah, "Chinese Law Firms Are in Trouble as Downward Spiral Show No Signs of Easing," September 16, 2024.
 ²²⁹ Citi Wealth and Hildebrandt Consulting, *2025 Citi Hildebrandt Client Advisory*, December 5, 2024, 10.

²³⁰ Citi Wealth and Hildebrandt Consulting, *2025 Citi Hildebrandt Client Advisory*, December 5, 2024, 10; *2024 Citi Hildebrandt Client Advisory*, December 6, 2023, 9.

²³¹ OECD, OECD Services Trade Restrictiveness Index (STRI)—Legal Services 2023, January 2024, 1.

litigation against the Bar Council and actions by the Indian government.²³² Additionally, the U.S.-Korea Free Trade Agreement includes commitments specific to the legal services sector, including obligations allowing U.S. law firms to establish joint ventures with Korean law firms and employ Korean-licensed lawyers.²³³ One industry representative has noted the difficulty related to hiring local attorneys and the challenges that poses for U.S. law firms operating in Korea to deliver legal services in the market.²³⁴

Outlook

U.S. law firms are expected to continue adopting technology, including GenAl tools, to enhance the delivery of their services and streamline business practices.²³⁵ Such integration is expected to increase the efficiency, service quality, and competitive position of law firms globally, including for smaller firms competing with larger firms.²³⁶ In turn, failure to adopt legal technologies is increasingly seen as a "risk of falling behind," especially given the projected expansion of higher-tech ALSPs, including the Big Four accounting firms.²³⁷ U.S. law firms are also likely to continue scaling back their presence in certain practice areas or foreign markets, depending on their growth potential and the underlying regulatory environment.²³⁸

Architectural and Engineering Services: Rising Costs and Technology Are Affecting Suppliers

In recent years, U.S. providers of architectural and engineering services have faced several supply-side challenges—including rising input costs and labor issues—that have had a substantial impact on firm performance. Global inflationary pressures, issues affecting the global supply chain, and resource demands from megaprojects have contributed to higher prices for project materials as well as office and travel expenses.²³⁹ Elevated interest rates have also affected the cost of project financing and project starts, impacting revenues for architectural and engineering services.²⁴⁰ However, steady wage growth and difficulties attracting talent have been the primary challenges facing firms. Wage increases have

 ²³² Majmudar & Partners, "Current State of Play of the Liberalization of India's Legal Market," March 4, 2025;
 OECD, "OECD Services Trade Restrictiveness Index Simulator," accessed January 27, 2025.
 ²³³ USTR 1, USTR 1,

²³³ USTR, Annex II, 2007, 44–45.

²³⁴ U.S. industry representative, interview by USITC staff, August 21, 2024.

²³⁵ Citi Wealth and Hildebrandt Consulting, 2025 Citi Hildebrandt Client Advisory, December 5, 2024, 8.

²³⁶ Thomson Reuters, "Impact of AI on Law Firms of Every Size," August 15, 2023; International Bar Association and the Center for AI and Digital Policy, *The Future Is Now*, September 2024, 16.

²³⁷ MarketLine, *United States (US) Legal Services Market Summary*, February 2024, 17; *Global Legal Services Market Summary*, February 2024, 16.

 ²³⁸ Citi Wealth and Hildebrandt Consulting, *2025 Citi Hildebrandt Client Advisory*, December 5, 2024, 8.
 ²³⁹ Adolphus and Keller, "The Top 500 Design Firms," April 29, 2024, 66; Clarke, "The Effects of Inflation on the AEC Industry," May 16, 2024; IBISWorld, *Architects in the US*, November 2024; *Engineering Services in the US*, August 2024. For example, one source reports that from February 2020 to August 2023, the cost of inputs used in construction projects grew by over 40 percent. Meisels et al., "2024 Engineering and Construction Industry Outlook," November 6, 2023.

²⁴⁰ IBISWorld, *Engineering Services in the US*, August 2024; *Architects in the US*, November 2024; Clarke, "The Effects of Inflation on the AEC Industry," May 16, 2024.

been insufficient to attract talent in part because science, technology, engineering, and mathematics (STEM) graduates have pursued more lucrative opportunities in other fields, and firms find it difficult to raise wages further because of constrained revenue growth. The adoption of new technologies offers a possible means of addressing these challenges but may entail costs related to software and equipment purchases, the incorporation of technology into existing processes, and training.

Labor Costs and Attracting Talent Challenge Suppliers

In recent years, steadily rising labor costs and difficulties attracting talent have challenged the competitiveness of architectural and engineering services suppliers.²⁴¹ This has been a particular issue for the U.S. architectural services industry, where revenue growth has been constrained. Architectural and engineering services is a labor-intensive industry that relies heavily on skilled workers with specialized expertise and complex problem-solving responsibilities.²⁴² Wage rates among certain U.S. and Canadian architectural and engineering service providers reportedly increased by 5 percent in 2022 before moderating to a less than 1 percent increase in 2023.²⁴³ Sources indicate that recent increases in labor costs are squeezing industry profits and straining project delivery as well as the ability to pursue new opportunities.²⁴⁴

A decline in the number of STEM graduates has affected labor supply in the architectural and engineering services industry.²⁴⁵ In the U.S. architectural services industry in particular, despite recent wage increases, wages remain lower than in other STEM fields.²⁴⁶ One industry representative reports that relatively low wages have contributed to declining interest in architecture degrees and that visa issues affect the ability of educational institutions to sustain enrollment levels with international students.²⁴⁷ Industry sources also suggest that barriers to providing services in many architectural and engineering industry markets are relatively high because architects and engineers are generally subject to stringent licensing requirements.²⁴⁸

²⁴¹ IBISWorld, Architects in the US, November 2024; Engineering Services in the US, August 2024.

²⁴² IBISWorld, *Global Engineering Services*, May 2024; *Global Architectural Services*, June 2024; *Engineering Services in the US*, August 2024.

²⁴³ Deltek, *Clarity: Architecture & Engineering Industry Study*, May 2024, 102. These figures are based on a survey of architecture and engineering firms, in which 90 percent of respondents were based in the United States, and 10 percent were based in Canada.

 ²⁴⁴ Deltek, *Clarity: Architecture & Engineering Industry Study*, May 2024, 102; IBISWorld, *Architects in the US*, November 2024. Adolphus and Reina, "The Top 225 International Design Firms," August 7, 2023, 35–36, 40.
 ²⁴⁵ Adolphus and Keller, "The Top 500 Design Firms," April 29, 2024, 64.

²⁴⁶ U.S. Department of Labor data suggest that wages paid to architects are lower than those for many other STEM occupations. For example, mean wages for architects (other than naval and landscape architects) in the United States stood at \$48.27 per hour in May 2023 and mean wages for engineers, computer occupations, and mathematical science occupations stood at \$53.79, \$54.39, and \$53.53, respectively. USDOL, BLS, "May 2023 National Occupational Employment and Wage Estimates," accessed March 4, 2025.

²⁴⁷ Industry representative, interview by USITC staff, January 31, 2025.

²⁴⁸ IBISWorld, *Global Architectural Services*, June 2024; *Engineering Services in the US*, August 2024; *Architects in the US*, November 2024.

These industries continue to confront challenges related to hiring and retaining a skilled workforce that have reportedly impeded firms' ability to take on work and grow their businesses.²⁴⁹ One recent survey identified hiring and retention of qualified employees as the top challenge affecting architectural and engineering firms' finances and found that hiring workers is complicated by difficulties in finding skilled candidates and the ability to offer competitive wages.²⁵⁰ Another survey of engineering firms found that talent shortages have led more than 50 percent of respondents to decline work opportunities in recent months and many of these firms have also increased their selectivity when accepting projects. Additionally, this survey reports that more than 25 percent of firms had declined opportunities characterized as "good, profitable."²⁵¹ Some firms have reportedly addressed skills needs by devoting resources to employee retention and initiating programs, such as internships, that aim to facilitate knowledge transfer from experienced workers to new employees.²⁵² Also, as discussed below, firms are adopting new technologies to increase worker efficiency and take on simple tasks.

In their recruiting efforts, some companies are offering nonmonetary benefits such as mentorship opportunities and flexible schedules to alleviate growing competition in compensation.²⁵³ For example, Tetra Tech offers employee training programs that focus on project management, leadership, and client relationship skills,²⁵⁴ and Operations Management and Facilities Services at Jacobs Solutions offers a one-year program to mentor future leaders.²⁵⁵ One industry source adds that subcontracting may mitigate labor costs.²⁵⁶

Labor supply issues remain a key concern in the short term, but some evidence shows this may change soon. One recent survey reported a decrease in open positions and an increase in job acceptance rates among architectural and engineering firms, suggesting that labor challenges may be abating.²⁵⁷ Another survey found that 68 percent of responding engineering firms expect increased hiring during the coming year, and it reported a decrease in the share of open positions.²⁵⁸ Furthermore, an industry representative suggested that labor supply pressure may ease if a recent decline in contracts and inquiries for design work leads to a decreased number of projects.²⁵⁹

Architectural and Engineering Services Suppliers Increasingly Adopt High-Tech Solutions

Industry surveys indicate that many architectural and engineering services providers continue to use manual and paper-based techniques for work tasks. One recent survey indicated that at a large number of firms, data entry tasks are still performed manually, particularly for business operations such as

²⁴⁹ Adolphus and Keller, "The Top 500 Design Firms," April 29, 2024, 64–65, 68.

²⁵⁰ Deltek, Clarity: Architecture & Engineering Industry Study, May 2024, 62, 88.

²⁵¹ ACEC, Engineering Business Sentiment, October 2024, 14–15.

²⁵² Adolphus and Keller, "The Top 500 Design Firms," April 29, 2024, 68.

²⁵³ Deltek, *Clarity: Architecture & Engineering Industry Study*, May 2024, 62.

²⁵⁴ Tetra Tech, "Life at Tetra Tech," accessed February 11, 2025.

²⁵⁵ Jacobs, "Opportunity Awaits," accessed February 11, 2025.

²⁵⁶ IBISWorld, Architects in the US, November 2024; Engineering Services in the US, August 2024.

²⁵⁷ Deltek, *Clarity: Architecture & Engineering Industry Study*, May 2024, 60.

²⁵⁸ ACEC, Engineering Business Sentiment, October 2024, 6, 12.

²⁵⁹ Industry representative, interview by USITC staff, January 31, 2025.

finance, accounting, management, and administration.²⁶⁰ In another survey, 28 percent of responding firms indicated that their operations are paper free and at least 30 percent of respondents reported using paper-based approaches for tasks such as collecting signatures, archiving, and historical documentation, or because paper-based approaches are perceived as easier or preferred.²⁶¹ However, one survey suggests that industry participants generally anticipate increasing digital maturity in the near future,²⁶² and the adoption of technologies has accelerated as firms aim to boost quality and efficiency, reduce costs, and offset talent gaps.²⁶³

The adoption of new technologies offers a possible means of addressing these supply-side challenges but may entail costs related to software and equipment purchases, the incorporation of technology into existing processes, and training. The use of AI in the architectural and engineering services industry is reportedly nascent, with firms proceeding cautiously.²⁶⁴ However, several sources indicate that the technology is already transforming some work processes in the industry,²⁶⁵ and many firms report that they have devoted financial resources to AI implementation.²⁶⁶ Furthermore, prior experience with the implementation of other technologies, such as Building Information Modeling (BIM) (discussed below), has reportedly eased industry concerns regarding the adoption of more recent technologies, such as AI.²⁶⁷

A recent survey indicated that almost 50 percent of responding architecture and engineering firms were using AI in some capacity and identified some of the most common applications as the development of content for marketing and proposals, summarizing data, and creating renderings and models.²⁶⁸ Other sources point to design; project planning, management, and archiving; risk assessment; and self-assessment of firm styles as among the top applications of AI by architectural and engineering services providers.²⁶⁹ For example, Gensler is reportedly testing the use of AI in design tasks, and ZGF Architects has created in-house AI tools for tasks such as photo descriptions and code review.²⁷⁰ Industry observers suggest that the use of AI has the potential to reduce labor and materials costs, increase productivity and efficiency by automating certain routine tasks, boost creativity, optimize designs, and improve decision-making through increased information processing.²⁷¹ At the same time, firms are balancing the

²⁶⁰ Deltek, *Clarity: Architecture & Engineering Industry Study*, May 2024, 17.

²⁶¹ Bluebeam, *Building the Future*, November 2024.

²⁶² Deltek, *Clarity: Architecture & Engineering Industry Study*, May 2024, 11.

²⁶³ IBISWorld, Architects in the US, November 2024; Engineering Services in the US, August 2024; Bluebeam,

Building the Future, November 2024; Adolphus and Keller, "The Top 500 Firms," April 29, 2024, 64, 71–72.

²⁶⁴ IBISWorld, *Architects in the US*, November 2024; industry representative, interview with USITC staff, January 31, 2025; Bluebeam, *Building the Future*, November 2024.

 ²⁶⁵ Soto, "Understanding the AEC Industry," April 20, 2024; Deltek, *Clarity: Architecture & Engineering Industry Study*, May 2024, 12; IBISWorld, *Architects in the US*, November 2024; industry representative, interview with USITC staff, January 31, 2025; Adolphus and Keller, "The Top 500 Design Firms," April 29, 2024, 75.
 ²⁶⁶ Bluebeam, *Building the Future*, November 2024.

²⁶⁷ Industry representative, interview by USITC staff, January 31, 2025.

²⁶⁸ Deltek, *Clarity: Architecture & Engineering Industry Study*, May 2024, 12.

²⁶⁹ Industry representative, interview by USITC staff, January 31, 2025; Bluebeam, *Building the Future*, November 2024; Rajan, "7 Technological Innovations," February 26, 2024.

²⁷⁰ Dorsey, "Gensler Among Several Design Firms," February 26, 2024.

²⁷¹ IBISWorld, *Architects in the US*, November 2024; Rajan, "7 Technological Innovations," February 26, 2024; Soto, "AI in Architecture," October 25, 2024.

use of AI as a tool with the continued need for human engagement in design.²⁷² It is anticipated that AI will affect the architectural and engineering industry more quickly than preceding technologies.²⁷³

Although AI is the focus of much current attention, less recently developed technologies are continuing to have a substantial impact. The use of BIM software is increasingly common in the global architectural and engineering industry.²⁷⁴ BIM software uses data on indicators such as project measurements and materials to create a detailed 3D model, enabling greater productivity, improved decision-making and quality, fewer errors, lower costs, and better communications, among other benefits.²⁷⁵ Increasing use of BIM software is being prompted by firms' need to respond to government directives and initiatives, and the ongoing development and revision of global BIM standards.²⁷⁶ For example, several U.S. cities, states, and federal government entities have developed BIM standards and objectives,²⁷⁷ and since 2016 the UK has mandated BIM use in public construction projects.²⁷⁸ According to a recent survey of managers and higher-level leaders in the architectural, engineering, and construction industry, firms that design, plan, build, maintain, and operate buildings employ BIM more frequently than any other technology, with 69 percent of respondents reporting its use at their firms.²⁷⁹ Another survey of architectural firms—which reports that 62 percent of respondents use BIM—found that BIM is used most often for tasks such as visualizing designs, renderings and presentations, document coordination, and model sharing.²⁸⁰

Digital twins have also become a valuable tool in the industry.²⁸¹ Like BIM models, digital twins are 3D models of a physical structure or space; however, a digital twin is distinct in that it is a dynamic virtual copy of a structure or space, which is continuously updated with real-time information from the physical asset.²⁸² Digital twins can be used in both the development and operation of a project and reportedly may improve collaboration, increase efficiency, and reduce costs and waste.²⁸³ Digital twins can also be used in conjunction with BIM models,²⁸⁴ a practice that one source indicates will become common.²⁸⁵

²⁷² Industry representative, interview by USITC staff, January 31, 2025; Soto, "AI in Architecture," October 25, 2024.

²⁷³ Adolphus and Keller, "The Top 500 Design Firms," April 29, 2024, 73, 76.

²⁷⁴ Pinnacle Infotech, *Global BIM Adoption*, January 1, 2024.

²⁷⁵ IBISWorld, *Architects in the US*, November 2024; Soto, "Understanding the AEC Industry," April 20, 2024.

²⁷⁶ Pinnacle Infotech, *Global BIM Adoption*, January 1, 2024.

²⁷⁷ DBM Vircon, "How Government Mandated BIM Standards Are Being Applied," May 23, 2023.

²⁷⁸ Freeman, "The Rise of Building Information Modelling," July 29, 2024.

²⁷⁹ Bluebeam, *Building the Future*, November 2024.

²⁸⁰ AIA, *The Business of Architecture 2024*, October 2024, 43–44.

²⁸¹ Lane, "Digital Twins in Construction and Architecture," accessed January 27, 2025; Matterport, "What Is Digital Twin & BIM?," accessed January 27, 2025.

²⁸² Matterport, "What Is Digital Twin & BIM?," accessed January 27, 2025; Turney, "What Is a Digital Twin?," December 11, 2025.

²⁸³ Lane, "Digital Twins in Construction and Architecture," accessed January 27, 2025; Matterport, "What Is Digital Twin & BIM?," accessed January 27, 2025.

²⁸⁴ Matterport, "What Is Digital Twin & BIM?," accessed January 27, 2025.

²⁸⁵ Turney, "What Is a Digital Twin?," December 11, 2025.

Other tools used by architectural and engineering services firms to facilitate design and decision-making tasks include cloud technology and virtual and augmented reality.²⁸⁶ The storage of large volumes of information on cloud platforms enables access to up-to-date data from various locations and facilitates decision-making.²⁸⁷ Augmented and virtual reality allow for enhanced and immersive visualizations of a project, which reportedly can contribute to increased design accuracy, and can be used as a tool in construction rehearsals, training, and marketing, among other tasks.²⁸⁸ In a recent survey, more than 80 percent of respondents from large architectural firms reported the use of cloud computing and over 60 percent of such firms reported the use of virtual reality. Among midsize architectural firms, the use of cloud computing and virtual reality was reported by 69 percent and 39 percent of respondents, respectively; the use of these technologies stood at 51 percent and 17 percent among small firm respondents.²⁸⁹

High-technology hardware is also being adopted by the architectural and engineering services industry, as firms are employing robots and drones to facilitate several tasks. For example, architects use robots to manufacture customized parts for building projects and to construct precise models that can be used to test materials for particular structures.²⁹⁰ Among other tasks, drones collect images, measurements, and other information that enable architectural and engineering services providers to survey, map, inspect, or monitor progress at a site.²⁹¹ These tools reportedly benefit architectural and engineering services providers by boosting the speed and efficiency of certain tasks, increasing consistency and accuracy, lowering labor costs, enabling firms to conduct inspections while machinery is in operation, and performing tasks in dangerous or hard-to-access areas.²⁹² Industry observers expect that architectural and engineering services firms' use of robots and drones will expand in the near future, encompassing a growing number of tasks and complementing or enhanced by other technologies (such as AI and BIM).²⁹³

Industry sources indicate that architectural and engineering services firms face several challenges to technology adoption and operation. These include difficulties in determining technology priorities, technology implementation costs (which may be particularly burdensome for small firms), lack of available training time, the unpredictability of returns, and cybersecurity and liability issues among others.²⁹⁴ For example, one source noted that the skilled use of AI in architecture can require extensive upskilling and reports that more than one-third of architects perceive the unavailability of training as an

²⁸⁶ Adolphus and Keller, "The Top 500 Design Firms," April 29, 2024, 75; AIA, *The Business of Architecture 2024*, October 2024, 47; Rajan, "7 Technological Innovations," February 26, 2024; Soto, "Understanding the AEC Industry," April 20, 2024; IBISWorld, *Architects in the US*, November 2024.

²⁸⁷ Soto, "Understanding the AEC Industry," April 20, 2024.

 ²⁸⁸ Rajan, "7 Technological Innovations," February 26, 2024; Soto, "Understanding the AEC Industry," April 20, 2024.
 ²⁸⁹ AIA, *The Business of Architecture 2024*, October 2024, 47.

²⁹⁰ HMC Architects, "Robotics in Architecture and Construction," October 23, 2019.

²⁹¹ Nasir, "The Role of Drones," October 8, 2024; PE Impact, "12 Uses of Drones," May 2, 2022.

 ²⁹² Adolphus and Keller, "The Top 500 Design Firms," April 29, 2024, 75; PE Impact, "12 Uses of Drones," May 2, 2022; HMC Architects, "Robotics in Architecture and Construction," October 23, 2019; IBISWorld, *Engineering Services in the US*, August 2024.

²⁹³ HMC Architects, "Robotics in Architecture and Construction," October 23, 2019; Nasir, "The Role of Drones," October 8, 2024.

²⁹⁴ Deltek, *Clarity: Architecture & Engineering Industry Study*, May 2024, 10, 14; IBISWorld, *Architects in the US*, November 2024; Soto, "AI in Architecture," October 25, 2024.

obstacle to AI adoption.²⁹⁵ Another source noted that the use of AI in architecture may raise issues related to attribution, plagiarism, and other ethical concerns.²⁹⁶

Outlook

Over the next five years, the architectural and engineering workforce is expected to grow. More specifically, IBISWorld data suggest that employment in the U.S. and global architectural services industries will increase at average annual rates of over 2 percent during 2025–30 and 2025–29, respectively.²⁹⁷ IBISWorld also projects that employment in the U.S. and global engineering services industries will increase at rates of 1.3 percent and 1.0 percent, respectively, from 2025–30.²⁹⁸ At the same time, talent gaps and rising labor costs remain a concern that firms are expected to address through training, the introduction of technology, and outsourcing.²⁹⁹ The use of technologies such as AI, BIM, and cloud computing is expected to increase in the near future and to broaden to an increasing number of tasks.³⁰⁰ According to a 2024 survey of architecture and engineering firms, 76 percent of respondents expected that their organizations' levels of digital transformation would be mature or advanced within three years, while 35 percent of respondents considered their organizations' current levels to be mature or advanced.³⁰¹ The adoption of new technologies will reportedly lead to efficiency gains as well as heightened training and cybersecurity needs.³⁰² Technology adoption is also expected to be a determinant of future industry competitiveness.³⁰³

Accounting and Audit Services: Global Firms Are Leveraging New Technologies and Outsourcing to Maintain Competitiveness

Global accounting firms provide a variety of services to their clients, including tax, assurance, and advisory services, among others. Generally, tax and assurance services are industry terms that roughly correlate to the definitions of accounting and auditing, respectively, outlined in chapter 2. Fundamentally, auditors provide assurance services when they verify financial statements and tax documents prepared by accountants (who are providing accounting and tax services). The major

²⁹⁵ Soto, "AI in Architecture," October 25, 2024.

²⁹⁶ RIBA, *RIBA AI Report 2024*, February 29, 2024, 25.

²⁹⁷ IBISWorld did not provide projections for the global architectural services industry for 2030.

²⁹⁸ USITC staff calculations based on IBISWorld, *Architects in the US*, November 2024; *Engineering Services in the US*, August 2024; *Global Architectural Services*, June 2024; *Global Engineering Services*, May 2024.

²⁹⁹ IBISWorld, *Architects in the US*, November 2024; *Engineering Services in the US*, August 2024; Soto, "Understanding the AEC Industry," April 20, 2024.

³⁰⁰ IBISWorld, *Architects in the US*, November 2024; *Engineering Services in the US*, August 2024; *Global Engineering Services*, May 2024.

³⁰¹ Deltek, *Clarity: Architecture & Engineering Industry Study*, May 2024, 11.

³⁰² IBISWorld, *Engineering Services in the US*, August 2024; *Architects in the US*, November 2024; Adolphus and Keller, "The Top 500 Design Firms," April 29, 2024, 73, 75; Deltek, *Clarity: Architecture & Engineering Industry Study*, May 2024, 10–11, 14.

³⁰³ IBISWorld, *Engineering Services in the US*, August 2024; *Architects in the US*, November 2024.

accounting and audit firms also frequently provide advisory services, including management consulting services (discussed in chapter 4) through their advisory services arms.³⁰⁴ The large international accounting firms are leveraging digital technologies and AI—as well as turning to outsourcing and offshoring when more cost effective—to supply the full range of tax and audit services more competitively in U.S. and global markets.

In the last 20 years, demand from large multinational companies for tax and auditing services has been naturally capped.³⁰⁵ Companies require tax and auditing services on a regular annual schedule. In response, the Big Four accounting firms have driven revenue growth by steadily increasing their supply of higher-value advisory services, ³⁰⁶ as well as by making their tax and auditing businesses more cost effective.³⁰⁷ Technological advances, including robot process automation, data analytics, and AI, have allowed firms to automate certain low-skill tax and auditing functions. Outsourcing and offshoring strategies help reduce labor and other input costs, although they also entail significant coordination costs. Firms also see the supply of both advisory services and traditional tax and audit services as a competitive advantage. In addition to higher-value output, advisory services can facilitate deeper connections with clients compared to "once-a-year" tax and assurance services.³⁰⁸

Global Accounting and Audit Firms Seek Ways to Increase Labor Productivity and Lower Costs

U.S. accounting firms continue to increase total revenues (even as there are less accountants employed in the United States), but the outlook for revenue growth from providing tax and auditing services is constrained (figure 3.1).³⁰⁹ Relatively slow business growth has limited remuneration levels for accountants and auditors, which in turn has made the profession less attractive to new entrants.³¹⁰ According to Bureau of Labor Statistics data, total accountants in the United States decreased by 300,000 from 2019–21.³¹¹ Many of these separations may have been retirements, as 75 percent of certified public accountants (CPAs) reached retirement age by 2019.³¹² Furthermore, after the "Great Resignation" and the COVID-19 pandemic, the number of employed accountants dropped sharply in 2020–21 and has

³⁰⁴ This section will refer to firms that perform accounting and auditing services, among others, as accounting firms. References to tasks will be specified as tax services (e.g., tax preparation), audit services (e.g., financial document verification), and advisory services (e.g., management consulting).

³⁰⁵ Cogar, "Big 4 Audit Clients," September 19, 2024.

³⁰⁶ Partially because of the reasons for increased demand discussed in the Management Consulting section of chapter 4. Dey and Quamina, "Surveying a Shifting Landscape," July 24, 2024.

³⁰⁷ The Big Four accounting firms refers to Deloitte, PricewaterhouseCoopers, Ernst & Young, and KPMG, which collectively account for nearly 34 percent of global accounting revenue; see table 2.1.

³⁰⁸ Kenney, "Small Firms Find Success with Advisory Services," August 1, 2024.

³⁰⁹ AIM Research, "Deloitte's Slowdown Signals the Need for Big 4 Transformation," September 17, 2024.

³¹⁰ AIM Research, "Deloitte's Slowdown Signals the Need for Big 4 Transformation," September 17, 2024; Cruz-Martínez, "The Big CPA Shortage Problem in Accounting," September 19, 2024.

³¹¹ Maurer, "Job Security Isn't Enough to Keep Many Accountants from Quitting," September 22, 2023.

³¹² Pardo, "US Companies Turn to Global Talent Pool as Accountant Shortage Hits Crisis Point," August 26, 2024.

not recovered.³¹³ In addition, fewer people are seeking accounting degrees, and, of those who do, fewer are sitting for their CPA exam.³¹⁴ According to industry press, many recent accounting graduates do not believe that becoming an accountant has enough return on investment.³¹⁵ Reportedly, the education and testing requirements do not offer enough salary to entice recent graduates, who may perceive the profession as boring.³¹⁶ These two trends have contributed to the increased amount of open accounting positions and to those positions staying open longer.³¹⁷

Figure 3.1 Total number of accountants and auditors in the United States and total revenue for accounting services in the United States, 2019–23





Source: Census, "Accounting, Tax Preparation, Bookkeeping, and Payroll Services: U.S. Total," accessed January 8, 2025; USDOL, BLS, "Labor Force Statistics-Employed: Accountants and Auditors," accessed January 8, 2025.

Labor demand is also relevant. Reportedly, the Big Four accounting firms hired excess staff during the COVID-19 pandemic—partially as a result of overestimating staff attrition.³¹⁸ Subsequently, overstaffing in the Big Four accounting firms (and other large accounting firms) led to staff reductions in several

³¹³ The "Great Resignation" refers to the period when 4.3 million U.S. workers quit between February 2020 and January 2022. Amanor-Boadu, "Empirical Evidence for the 'Great Resignation,'" November 2022; USDOL, BLS, "Labor Force Statistics-Employed: Accountants and Auditors," accessed January 8, 2025.

³¹⁴ Wood et al., "2024 CFO Talent Pulse Survey," 2024; Kasztelnik and Campbell, "The Future of Business Data Analytics and Accounting Automation," January 31, 2024; Cruz-Martínez, "The Big CPA Shortage Problem in Accounting," September 19, 2024.

³¹⁵ Cruz-Martínez, "The Big CPA Shortage Problem in Accounting," September 19, 2024.

³¹⁶ Cruz-Martínez, "The Big CPA Shortage Problem in Accounting," September 19, 2024.

³¹⁷ USDOL, BLS, "Labor Force Statistics-Employed: Accountants and Auditors," accessed January 8, 2025.

³¹⁸ Maurer, "Big Four Accounting Firms Pare Their Consultant Ranks in Postpandemic Reversal," July 18, 2023.
service lines; multiple rounds of layoffs affected thousands of workers in 2023 and 2024.³¹⁹ According to an industry executive, accounting firms needed to recalibrate their staffing models to address slowing growth and overcapacity.³²⁰

U.S. Accounting Firms Are Outsourcing and Offshoring

Because accounting firms are staffing fewer accountants in the United States, they are outsourcing more tax services. Outsourcing can improve cost-effectiveness while also helping accounting firms expand their international network.³²¹ Tax preparation services are often outsourced because they are easily digitalized and can be executed remotely.³²² In addition to restricted labor supply in the United States, lower labor costs—even after considering the higher coordination costs associated with managing an international supply chain—provide a powerful incentive to outsource. U.S. accountants averaged \$90,780 in annual salary compared to \$3,072 for Indian accountants in 2023.³²³ Non-Big Four firms are also outsourcing or offshoring operations in India. For example, Brussels-based BDO began offshoring to India in 2015.³²⁴ Furthermore, the Big Four accounting firms are expanding their operations in India, moving away from large Indian cities like Mumbai, Delhi, and Bengaluru to smaller cities with lower costs like Jaipur or Vadodara.³²⁵ Industry reports suggest that global accounting firms expanding in these "tiertwo" cities in India can benefit from favorable local regulations, improved infrastructure, and lower costs of labor and office space.³²⁶

Global Accounting Firms Continue to Digitalize Their Operations

Accounting firms have sought to increase productivity and lower costs by continuing to digitalize their operations and adopting new technologies, including robot process automation (RPA), cloud computing, and data analytics.

³¹⁹ Maurer, "Big Four Accounting Firms Pare Their Consultant Ranks in Postpandemic Reversal," July 18, 2023; Bramwell, "KPMG to Cut Audit Staff by About 4%," November 5, 2024; Big 4 Accounting Firms, "Big 4 Accounting Firm Layoffs 2025," February 8, 2025.

³²⁰ Maurer and Saeedy, "Accounting Firms Overhired," December 15, 2023.

³²¹ IBISWorld, *Global Accounting Services*, May 2024, 25; Invedus, "How Outsourcing Accounting Drives Growth in CPA Firms?," May 12, 2023.

³²² IBISWorld, *Global Accounting Services*, May 2024, 25.

³²³ USDOL, BLS, "Accountants and Auditors," April 3, 2024; Sharma, "What Is the Salary of an Accountant," April 1, 2024. USITC calculations of exchange rate from 263,069 Indian rupees based on 1 rupee = \$0.012 as of April 1, 2020.

³²⁴ *Finance Story*, "This Ex-BDO Manager Led Offshore Operations," October 16, 2023.

³²⁵ Kumar et al., "Global Accounting Firms Set up Shop in India's Smaller Cities," July 12, 2023; Sen, "How India Is Gearing up for a US\$110b GCC Industry by 2030," June 8, 2023.

³²⁶ Kumar et al., "Global Accounting Firms Set up Shop in India's Smaller Cities," July 12, 2023; Sen, "How India Is Gearing up for a US\$110b GCC Industry by 2030," June 8, 2023.

Robotic Process Automation

Historically, tax and auditing functions have used a series of computerized processes that were connected and advanced by human interventions and keystrokes, like downloading data or moving data to a spreadsheet.³²⁷ These tax and auditing tasks include data capture and entry, reconciliation, balance sheet management, and financial reporting, among others.³²⁸ RPA refers to software programs or robots that automate these manual actions and keystrokes. RPA is designed to let accountants focus on other tasks, including higher-value advisory tasks that require human judgment and expertise, such as interpretation and analysis.³²⁹ For example, one report estimates that 77 percent of general tax functions can be fully automated.³³⁰ RPA software can standardize procedures by improving efficiency, reducing error, and enhancing regulatory compliance.³³¹ Furthermore, RPA streamlines workflow processes. For example, RPA generates real-time data that allow accounting firms to continuously perform tax and auditing tasks throughout a reporting period and to adapt quickly to changing conditions.³³² This allows global accounting firms to streamline their financial closing tasks throughout reporting periods to avoid sharp increases in workloads.³³³

Advances in RPA have led to concerns about RPA robots replacing humans. Reportedly, major RPA initiatives have been followed by layoffs because corporations are aiming to cut costs.³³⁴ Practitioners often stress that humans still must participate in critical decisions. Humans may retain knowledge and expertise that RPAs do not, likely have different ethical standards, and may have a more comprehensive and targeted strategic vision.³³⁵

Cloud Computing

Global accounting firms have continued to leverage cloud technologies to deliver services to their clients. Cloud computing refers to the delivery of technological services via the internet. These services include data storage, data analytics, and software provision, among others.³³⁶ Many firms use cloud computing to lower operational costs, improve collaboration across offices, and allow for constant access to information.³³⁷ Furthermore, cloud services providers are likely to have better security protocols and

³²⁷ Vasarhelyi and Rozario, "How Robotic Process Automation Is Transforming Accounting and Auditing," July 2, 2018.

³²⁸ Vasarhelyi and Rozario, "How Robotic Process Automation Is Transforming Accounting and Auditing," July 2, 2018.

³²⁹ Vasarhelyi and Rozario, "How Robotic Process Automation Is Transforming Accounting and Auditing," July 2, 2018; Ayinla et al., "The Role of Robotic Process Automation (RPA)," February 17, 2024, 431–32.

³³⁰ Plaschke et al., "Bots, Algorithms, and the Future of the Finance Function," January 9, 2018.

³³¹ Ayinla et al., "The Role of Robotic Process Automation (RPA)," February 17, 2024, 428, 432.

³³² Singh, "Robotic Process Automation in Accounting," September 20, 2024.

³³³ The closing process in accounting refers to end-of period processes in which accounting firms update the financial accounts of their clients in preparation for producing financial statements, among other tasks. Principles of Accounting, "The Accounting Cycle and Closing Process," accessed March 4, 2025; Singh, "Robotic Process Automation in Accounting," September 20, 2024.

³³⁴ Roose, "The Robots Are Coming for Phil in Accounting," March 6, 2021.

³³⁵ Singh, "Robotic Process Automation in Accounting," September 20, 2024.

³³⁶ Microsoft, "What Is Cloud Computing?," accessed December 30, 2024.

³³⁷ Abogadie, "How Cloud Computing Accounting Is Changing the Industry," May 14, 2024.

data storage infrastructure than in-house operations. Cloud computing can improve a firm's performance; for example, a global accounting team can react instantly to new information and the entire team can access the same document simultaneously.³³⁸ Cloud-based accounting is virtually ubiquitous in the accounting industry, and reportedly firms that use only cloud accounting technologies generate five times the client growth annually compared to those that do not.³³⁹

Data Analytics

Global accounting businesses use data analytics to give financial data-based accounting advice to their clients. Because of the global proliferation of "big data" and increased computing power, accounting and auditing firms have more information available to analyze the financial positions of their clients.³⁴⁰ Global accounting firms are also using unstructured data, including video, audio, and text that can be generated from social media.³⁴¹ Reportedly, data analytics are standard and used by 90 percent of audit departments.³⁴² Data analytics can be broadly categorized as descriptive, diagnostic, predictive, and prescriptive.

Global accounting firms use descriptive analytics to understand the current status of their client's financial situation.³⁴³ Modern data analytics techniques can sift through historical data to identify patterns and better understand the client's historical trend.³⁴⁴ For example, accounting firms can analyze historical customer preferences, purchasing patterns, and demographics.³⁴⁵ Diagnostic analytics refer to assessing the reasons for the client's current financial position.³⁴⁶ Accounting firms sift through data to understand why patterns (e.g., client spending habits) have emerged historically.³⁴⁷ Predictive analytics refer to accounting firms, providing their clients business and economic forecasts—for example, predicting future customer purchasing patterns using client data and external economic indicators.³⁴⁸

³³⁸ Abogadie, "How Cloud Computing Accounting Is Changing the Industry," May 14, 2024.

³³⁹ Gaetano, "2023's Major Trends in Accounting Tech," November 29, 2023; Raza, "Cloud Accounting in 2024," April 30, 2024.

³⁴⁰ Big data refers to the availability of large amounts and types of data that has been facilitated by advancements in computer technologies, such as cloud computing and storage. These technologies allow for large amounts of data to be analyzed more quickly and at a lower cost. Herath and Woods, "Impacts of Big Data on Accounting," December 2021, 196.

³⁴¹ Unstructured data refers to data that do not have a defined organizational structure, including text, audio, photo, and video formats Herath and Woods, "Impacts of Big Data on Accounting," December 2021, 188–89, 191, 198.

³⁴² The survey was conducted among 127 global chief audit executives. Gartner, "Top 5 Audit Trends and Priorities in 2025," 2024, 5.

³⁴³ Franklin University, "Accounting & Data Analytics: What You Need to Know," accessed December 30, 2024.

³⁴⁴ Herath and Woods, "Impacts of Big Data on Accounting," December 2021, 200.

³⁴⁵ OneClick IT Consultancy, "Data Analytics in Accounting," June 19, 2024.

³⁴⁶ Franklin University, "Accounting & Data Analytics: What You Need to Know," accessed December 30, 2024.

³⁴⁷ Herath and Woods, "Impacts of Big Data on Accounting," December 2021, 200.

³⁴⁸ Franklin University, "Accounting & Data Analytics: What You Need to Know," accessed December 30, 2024; Herath and Woods, "Impacts of Big Data on Accounting," December 2021, 200; OneClick IT Consultancy, "Data Analytics in Accounting," June 19, 2024.

Accounting firms offer prescriptive analysis by analyzing data to produce objective reports that can identify the optimal actions a client may take to improve internal operations and better manage risk.³⁴⁹

Skill Considerations

Technological advances that have facilitated digitalization are likely changing the accounting profession, as firms now require technical skills in addition to traditional accounting skills. Global accounting firms are increasingly looking to attract workers with the skills to leverage these technological advances. However, recent accounting graduates reportedly do not have the requisite skills in data analytics and automation.³⁵⁰ Although industry associations and accrediting agencies have recommended including these skills in accounting programs, evidence suggests that faculty in accounting schools do not have the necessary skills to prepare future accountants with data analytics skills.³⁵¹ This has led firms to hire fewer CPAs because they do not possess these skills in data analytics that can help accounting firms (including small firms) digitalize, lower costs, and offer more advisory services.³⁵² Firms are still seeking accountants with a firm grasp on accounting fundamentals but are increasingly requiring new hires to possess these additional skills. Furthermore, accounting firms are educating their accountants and auditors (both new and incumbent) in-house, not only in data analytics but also in cloud computing and cybersecurity.³⁵³

Global Accounting Firms Continue to Implement AI While Exploring GenAI Use Cases

In the past several years, global accounting firms have digitalized their operations and used AI techniques to improve their performance. For example, RPA automates repetitive tasks with robots that execute the same way in every instance, but AI systems use algorithms that learn and can improve processes as they operate.³⁵⁴ Furthermore, although AI can analyze and process large amounts of information and data, GenAI technologies like ChatGPT can generate new output, including forecasts, financial reports, and financial advice.³⁵⁵ GenAI technologies use sophisticated large-content machine learning (ML) models to generate high-quality original output such as text and images (as opposed to predicting outcomes based on a data set).³⁵⁶ As they continue employing AI technologies, accounting and auditing firms are exploring test cases for GenAI in their operations.³⁵⁷

³⁵² Kasztelnik and Campbell, "The Future of Business Data Analytics and Accounting Automation," January 31, 2024.

"Addressing the Accountancy Skills Gap," January 24, 2022.

³⁴⁹ Franklin University, "Accounting & Data Analytics: What You Need to Know," accessed December 30, 2024; Herath and Woods, "Impacts of Big Data on Accounting," December 2021; Kenan-Flagler, "Why Data Analytics Matters to Accountants," March 23, 2021.

³⁵⁰ SAPRO, "The Accounting Skills Shortage," May 6, 2024; Kennedy, "The Accounting Shortage Is Hurting Your Bottom Line," August 12, 2024.

³⁵¹ Salimi, "Degree of Inclusion of Data Analytics," March 1, 2024.

³⁵³ XBRL is a global digital reporting standard that facilitates the exchange of business information. INAA,

³⁵⁴ Centreviews, "RPA vs. AI What's the Difference?," July 14, 2021.

³⁵⁵ Perkins, "Beyond Spreadsheets," September 26, 2024.

³⁵⁶ Zewe, "Explained: Generative AI," November 9, 2023.

³⁵⁷ CPA.com, "GenAl Practical Use," accessed March 26, 2025.

Artificial Intelligence

Accounting firms are increasing their use of AI to improve the digitalization of their operations by further streamlining internal processes, improving customer service, and enhancing fraud detection and regulatory compliance. According to a global survey of individuals in the accounting sector, 83 percent of accounting firms are employing AI.³⁵⁸ Furthermore, another global survey of accountants indicated that 67 percent of accountants see the implementation of AI as a potential competitive advantage.³⁵⁹

Al can help international accounting firms make their operations more efficient by enhancing automated processes, improving data analytics, and allowing firms to better organize their clients' information. Similar to RPA, AI can automate many low-skill tasks, like tax preparation and bookkeeping.³⁶⁰ An AI algorithm may be able to use big data to identify patterns in a decision-making process, which cannot be performed by an RPA because, as noted above, an RPA simply executes the same way every time.³⁶¹ For example, Deloitte's Cognitive Advantage initiative improves automation, which allows Deloitte's employees to provide more advisory services.³⁶² Also, AI can improve data analytics by quickly reading and identifying patterns in financial statements to streamline the auditing process.³⁶³ For example, KPMG has developed KPMG Ignite, an AI platform that improves data analysis to offer clients more advisory services, including predictions, trend identification, and strategic consulting.³⁶⁴ Similarly, AI technologies may be used to quickly review a client's invoices to avoid processing duplicate charges.³⁶⁵ Furthermore, AI algorithms can be used to organize a client's data more efficiently.³⁶⁶ For example, Ernst & Young is employing an AI tool that reviews documents at a speed and accuracy that human accountants cannot match.³⁶⁷ These benefits of AI can allow accounting firms to shift their focus to higher-value tasks like strategic planning and custom client support that are more associated with advisory services.³⁶⁸ Reportedly, small and medium-sized firms are also employing AI tools to expand their provision of advisory services—improving their competitiveness with larger firms by offering financial insights that were previously mostly produced by larger companies.³⁶⁹

In addition, AI is allowing global accounting firms to shift and improve their customer service models. For example, AI can allow CPAs and other accounting professionals to devote more time to individualized

³⁵⁸ The survey comprised 539 individuals from 6 continents, with the majority coming from the United States. Karbon, "The State of AI in Accounting Report 2025," 2025, 12.

³⁵⁹ The survey comprised 595 individuals from 6 continents, with the majority coming from the United States. Karbon, "The State of AI in Accounting Report 2024," 2024, 11.

³⁶⁰ Gleason, "The Role of Technology and AI in Advisory Services," December 10, 2024.

³⁶¹ Gleason, "The Role of Technology and AI in Advisory Services," December 10, 2024; Centreviews, "RPA vs. AI What's the Difference?," July 14, 2021.

³⁶² Sahota, "The Dawn of a New Era," April 22, 2024.

³⁶³ BILL Operations, "How AI in Accounting Is Transforming the Industry," accessed December 30, 2024; IBISWorld, *Global Accounting Services*, May 2024, 25.

³⁶⁴ Sahota, "The Dawn of a New Era," April 22, 2024.

³⁶⁵ BILL Operations, "How AI in Accounting Is Transforming the Industry," accessed December 30, 2024.

³⁶⁶ BILL Operations, "How AI in Accounting Is Transforming the Industry," accessed December 30, 2024; IBISWorld, *Global Accounting Services*, May 2024, 25.

³⁶⁷ Sahota, "The Dawn of a New Era," April 22, 2024.

³⁶⁸ Gleason, "The Role of Technology and AI in Advisory Services," December 10, 2024.

³⁶⁹ Sahota, "The Dawn of a New Era," April 22, 2024.

customer support. The customer can perceive the role of accountants shifting from "transactional service provider" to "trusted advisor."³⁷⁰ This has also shifted the billing model for accounting firms. Advisory services allow accounting firms to transition from a transactional hourly billing model to a value-based model where firms can price services according to the value provided to their client.³⁷¹

Global accounting firms also use AI to detect fraud and improve compliance and risk management. Accounting firms can employ AI techniques such as ML and natural language processing to detect fraud.³⁷² AI models improve efficiency in fraud detection because they can analyze much more data than humans.³⁷³ AI systems can also allow global accounting firms to manage and monitor the changing regulations of the jurisdictions where their clients are active to help multinational firms remain compliant.³⁷⁴

The use of AI in accounting has privacy considerations when AI algorithms are based on the personal or proprietary financial data of clients. Handling and analyzing client data using AI increases risk for data breaches or unauthorized access.³⁷⁵

Generative Artificial Intelligence

As firms begin to incorporate AI technologies, they are also exploring test cases for GenAI. In accounting, GenAI has several applications including document processing, improved customer service, and report generation.³⁷⁶ GenAI can quickly sort through large amounts of financial documents leading to faster decision-making. GenAI can improve customer service through enhancing chatbots to be more humanlike and to better understand context. Also, GenAI can allow accounting firms to offer better forecasting to their clients. For example, GenAI can analyze historical data, current prices, and various economic indicators to better estimate future demand.³⁷⁷ GenAI can produce financial reports more quickly and systematically, which can allow accounting firms to generate detailed reports more often.³⁷⁸ According to a Thomson Reuters report, 49 percent of tax and accounting firms are not currently planning on implementing GenAI tools, while 30 percent of these firms do have plans to employ GenAI technologies.³⁷⁹

³⁷⁰ Gleason, "The Role of Technology and AI in Advisory Services," December 10, 2024.

³⁷¹ Gleason, "The Role of Technology and AI in Advisory Services," December 10, 2024.

³⁷² Kasztelnik and Jermakowicz, "Financial Statement Fraud Detection in the Digital Age," June 24, 2024; Sang and Kniepmann, "AI and Fraud," May 1, 2024; BILL Operations, "How AI in Accounting Is Transforming the Industry," accessed December 30, 2024.

³⁷³ BILL Operations, "How AI in Accounting Is Transforming the Industry," accessed December 30, 2024.

³⁷⁴ BILL Operations, "How AI in Accounting Is Transforming the Industry," accessed December 30, 2024.

³⁷⁵ Schweitzer, "Artificial Intelligence (AI) Ethics in Accounting," March 15, 2024, 67, 72–73.

³⁷⁶ Perkins, "Beyond Spreadsheets," September 26, 2024.

³⁷⁷ Perkins, "Beyond Spreadsheets," September 26, 2024.

³⁷⁸ Perkins, "Beyond Spreadsheets," September 26, 2024.

³⁷⁹ The survey consisted of 1,128 respondents in professional services industries (legal, tax, government, and corporate risk) across the United States, the UK, Canada, Australia, and New Zealand. Thomson Reuters, "How Do Different Accounting Firms Use AI?," June 3, 2024; Colson, "How Are Different Accounting Firms Using AI?," June 3, 2024.

The Big Four are already implementing GenAl solutions. For example, Deloitte has launched its "DARTbot," which is a GenAl chatbot that its staff can ask complex accounting questions, and Ernst & Young (EY) has announced a chatbot for internal payroll questions.³⁸⁰ Furthermore, PricewaterhouseCoopers (PwC) is the largest user of ChatGPT Enterprise and KPMG has announced at least \$2 billion in investment over three years starting in November 2023 to implement GenAl technologies.³⁸¹ GenAl may increase the competitiveness of smaller firms, too, because it can automate labor-intensive tasks, including financial statement analysis, content summarization, and sales projections.³⁸²

GenAl use has considerations related to biased output, transparency, and accountability.³⁸³ GenAl systems can produce biased output if the data they are trained on are biased (e.g., data based on discrimination).³⁸⁴ Also, GenAl technologies are often described as a "black box" because what occurs between the input of data and the output is uncertain, resulting in a lack of transparency.³⁸⁵ This uncertainty raises concerns for accountability because the source of biased outcomes may be unclear.³⁸⁶ These considerations likely necessitate human judgment when implementing GenAl.³⁸⁷

Skill Considerations

Similar to the technical skills discussed above, global accounting companies are seeking employees who possess AI skills that can improve firms' implementation of data analytics and automation. Furthermore, it is important for firms to ensure employees are proficient in both accounting fundamentals and AI.³⁸⁸ The introduction of AI in accounting may require that accountants have skills like discernment, ingenuity, and flexibility because AI and GenAI models may identify patterns that are not obvious or correct.³⁸⁹ Also, skilled human intervention is needed to build sound GenAI models.³⁹⁰ In addition, auditors must understand the algorithms and data structures to ensure their ethical and effective use and regulatory compliance—particularly because regulatory compliance is complicated and evolves differently in various jurisdictions.³⁹¹

³⁸⁰ Noto, "Deloitte Brings GenAI to Audit, Accounting Professionals with New Bot," November 9, 2023.

³⁸¹ PwC, "How PwC Is Using Generative AI to Deliver Business Value," January 2025; Cushman, "KPMG's \$2 Billion GenAI Plan Targets Transformation with Trust," November 2, 2023; Thomson Reuters, "How Do Different Accounting Firms Use AI?," June 3, 2024.

³⁸² Dethe, "Generative AI Can Boost Small Accountancy Firms," May 31, 2024; Stuart, "Could Gen AI End Incumbent Firms' Competitive Advantage?," November 21, 2024.

³⁸³ Lawton, "Generative AI Ethics," July 23, 2024.

³⁸⁴ Schweitzer, "Artificial Intelligence (AI) Ethics in Accounting," March 15, 2024, 74–75.

³⁸⁵ Schweitzer, "Artificial Intelligence (AI) Ethics in Accounting," March 15, 2024, 76–77.

³⁸⁶ Schweitzer, "Artificial Intelligence (AI) Ethics in Accounting," March 15, 2024, 78–79.

³⁸⁷ Sottolano, "Beyond the Numbers," September 11, 2024.

³⁸⁸ Sahota, "The Dawn of a New Era," April 22, 2024.

³⁸⁹ Choudhary, "The Crucial Role of Soft Skills in Accounting in the AI Era," May 23, 2024.

³⁹⁰ ICAEW Insights, "The Most Desired Skills for AI-Assisted Accountants," November 20, 2024; Thomson Reuters,

[&]quot;How Is GenAI Reshaping the Auditor's Skill Set?," September 24, 2024.

³⁹¹ Thomson Reuters, "How Is GenAl Reshaping the Auditor's Skill Set?," September 24, 2024.

Outlook

Global accounting firms are expected to continue increasing labor productivity and lowering costs by leveraging digitalization, AI, and GenAI to automate low-skill tax and auditing functions.³⁹² Some industry observers anticipate that leveraging these technological advancements to offer more advisory services will remain a competitiveness factor for global accounting firms.³⁹³ According to a survey of 291 accounting professionals in the United States, labor constraints are expected to persist. About 50 percent of firms that expect to hire new staff in 2025 are aiming to add only half the number hired in 2024.³⁹⁴ However, industry representatives have indicated that accounting firms will need to increase stagnant wages to attract recent graduates who may prefer professions with higher starting and career salaries, such as finance or marketing.³⁹⁵ Also, firms are expected to continue outsourcing and offshoring lower-skilled functions.³⁹⁶ Furthermore, increasing internet connectivity in the developing world will continue the digitalization of business practices globally and contribute to the continued automation of routine tax and auditing functions.³⁹⁷

³⁹² Raghunathan, "Five Trends That Will Redefine Finance and Accounting," January 7, 2025; IBISWorld, *Global Accounting Services*, May 2024, 24–25.

³⁹³ Spotlight Reporting, *Global Advisory Trends Report*, 2023, 6; Gleason, "The Role of Technology and AI in Advisory Services," December 10, 2024.

³⁹⁴ Hood, "The Year Ahead: 2025 in Accounting," December 17, 2024, 6–7.

³⁹⁵ Ellis and Overberg, "Why No One's Going Into Accounting," October 6, 2023; Cayto Group, "Declining CPA Candidates and Accounting Graduates," February 9, 2025.

³⁹⁶ IBISWorld, *Global Accounting Services*, May 2024, 25.

³⁹⁷ IBISWorld, *Global Accounting Services*, May 2024, 24.

Bibliography

- Abogadie, Avon. "How Cloud Computing Accounting Is Changing the Industry." *ClientHub*, May 14, 2024. <u>https://www.clienthub.app/blog/cloud-computing-accounting</u>.
- Adolphus, Emell, and Jonathan Keller. "The Top 500 Design Firms: Artificial Intelligence Ramps Up." Engineering News Record, April 29, 2024. <u>https://www.enr.com/</u>
- Adolphus, Emell, and Peter Reina. "The Top 225 International Design Firms: Resilient Design Demands Grow." *Engineering News Record*, August 7, 2023. <u>https://www.enr.com/</u>
- AIM Research. "Deloitte's Slowdown Signals the Need for Big 4 Transformation." September 17, 2024. <u>https://aimresearch.co/market-industry/deloittes-slowdown-signals-the-need-for-big-4-</u> <u>transformation</u>.
- Amanor-Boadu, Vincent. "Empirical Evidence for the 'Great Resignation.'" U.S. Department of Labor (USDOL). Bureau of Labor Statistics (BLS). *Monthly Labor Review*, November 2022. <u>https://www.bls.gov/opub/mlr/2022/article/empirical-evidence-for-the-great-resignation.htm</u>.
- American Bar Association (ABA). Formal Opinion 512: Generative Artificial Intelligence Tools. July 29, 2024.

https://www.americanbar.org/content/dam/aba/administrative/professional_responsibility/ethi cs-opinions/aba-formal-opinion-512.pdf.

- American Council of Engineering Companies (ACEC) Research Institute. *Engineering Business Sentiment* 2024 Q4. October 2024. <u>https://www.acec.org/resource/engineering-business-sentiment-q4-</u> 2024-2/.
- American Institute of Architects (AIA). *The Business of Architecture 2024*. October 2024. <u>https://www.aia.org/resource-center/aia-firm-survey-report-2024</u>.
- Ayinla, Benjamin Samson, Akoh Atadoga, Chinedu Ugochukwu Ike, Ndubuisi Leonard Ndubuisi, Onyeka Franca Asuzu, and Rhoda Adura Adeleye. "The Role of Robotic Process Automation (RPA) in Modern Accounting: A Review—Investigating How Automation Tools Are Transforming Traditional Accounting Practices." *Engineering Science & Technology Journal* 5, no. 2 (February 17, 2024): 427–47. <u>https://doi.org/10.51594/estj.v5i2.804</u>.
- Big 4 Accounting Firms. "Big 4 Accounting Firm Layoffs 2025." February 8, 2025. https://big4accountingfirms.com/the-blog/big-4-accounting-firm-layoffs-2025/.
- BILL Operations. "AI in Accounting: How Artificial Intelligence (AI) in Accounting Is Transforming the Industry." *BILL (blog)*. Accessed December 30, 2024. <u>https://www.bill.com/blog/ai-in-accounting</u>.
- Black, Nicole. "The Legal Tech Trends That Defined 2024." American Bar Association. *ABA Journal*, December 17, 2024. <u>https://www.abajournal.com/columns/article/the-legal-tech-trends-that-defined-2024</u>.

- Bloomberg Law. "Al in Legal Research: 8 Real Ways Al Can Impact Your Practice." November 21, 2023. https://pro.bloomberglaw.com/insights/technology/ai-impact-on-legal-research/.
- Bloomberg Law. "How Is AI Changing the Legal Profession?" May 23, 2024. https://pro.bloomberglaw.com/insights/technology/how-is-ai-changing-the-legal-profession/.
- Bloomberg Law. *Legal Ops and Tech Survey*. 2024. <u>https://pro.bloomberglaw.com/insights/legal-operations/legal-ops-and-tech-survey/</u>.
- Bloomberg Law. "The Real Impact of AI in Legal Research." Bloomberg Law. November 21, 2023. https://pro.bloomberglaw.com/insights/technology/ai-impact-on-legal-research/.
- Bluebeam. Building the Future: Bluebeam AEC Technology Outlook 2025. November 2024. https://www.bluebeam.com/resources/ebooks/2025-aec-industry-outlook/.
- Bramwell, Jason. "KPMG to Cut Audit Staff by About 4%." CPA Practice Advisor, November 5, 2024. <u>https://www.cpapracticeadvisor.com/2024/11/04/kpmg-to-lay-off-4-percent-of-audit-staff/152096/</u>.
- Bullows, Amy. "How Technology Is Changing the Legal Sector." *The Law Teacher* 55, no. 2 (April 3, 2021): 258–64. <u>https://doi.org/10.1080/03069400.2021.1915591</u>.
- Business Research Company. Legal Technology Global Market Report 2025—Legal Technology Market Research and Overview. January 2025. https://www.thebusinessresearchcompany.com/report/legal-technology-global-market-report.
- Cayto Group. "Declining CPA Candidates and Accounting Graduates." February 9, 2025. <u>https://caytogroup.com/why-the-us-accounting-talent-pipeline-is-broken-2025/</u>.
- Centreviews. "RPA vs. AI What's the Difference? How Can They Work Together?" Accessed December 28, 2024. <u>https://centreviews.com/automation/rpa-vs-ai</u>.
- Chandra, Kapil, Frank Plaschke, and Ishaan Seth. "Memo to the CFO: Get in Front of Digital Finance—or Get Left Back." McKinsey & Company, July 5, 2018. <u>https://www.mckinsey.com/capabilities/strategy-and-corporate-finance/our-insights/memo-to-the-cfo-get-in-front-of-digital-finance-or-get-left-back</u>.
- Choudhary, Amrita. "The Crucial Role of Soft Skills in Accounting in the Al Era." Kiplinger, May 23, 2024. <u>https://www.kiplinger.com/kiplinger-advisor-collective/crucial-role-of-soft-skills-in-accounting-in-the-ai-era</u>.
- Citi Wealth and Hildebrandt Consulting. 2024 Citi Hildebrandt Client Advisory. December 6, 2023. https://www.privatebank.citibank.com/insights/citi-hildebrandt-client-advisory.
- Citi Wealth and Hildebrandt Consulting. 2025 Citi Hildebrandt Client Advisory Reports Law Firm Industry Revenue Growth Driven by Increasing Demand. Press release, December 5, 2024. <u>https://www.citigroup.com/global/news/press-release/2024/2025-citi-hildebrandt-client-advisory-reports-growth-driven-by-increasing-demand</u>.

- Clarke, Jeremy. "The Effects of Inflation on the AEC Industry: Challenges and Adaptations in 2024." Emissary. May 16, 2024. <u>https://www.emissarysearch.com/post/the-effects-of-inflation-on-the-aec-industry-challenges-and-adaptations-in-2024</u>.
- Clio. "Products." Accessed February 3, 2025. https://www.clio.com/.
- Cogar, Joshua. "Big 4 Audit Clients." Management Consulted. September 19, 2024. https://managementconsulted.com/big-4-audit-clients/.
- Columbia University. Fu Foundation School of Engineering and Applied Science. "Artificial Intelligence (AI) vs. Machine Learning." Accessed April 16, 2024. <u>https://ai.engineering.columbia.edu/ai-vs-machine-learning/</u>.
- CPA.com. "Demystifying AI for the Accounting and Finance Profession." Accessed February 11, 2025. https://www.cpa.com/gen-ai.
- Cruz-Martínez, Gabriella. "The Big CPA Shortage Problem in Accounting." Kiplinger, September 19, 2024. <u>https://www.kiplinger.com/taxes/the-cpa-shortage-problem</u>.
- Cushman, David. "KPMG's \$2 Billion GenAI Plan Targets Transformation with Trust." HFS Research, November 2, 2023. <u>https://www.hfsresearch.com/research/kpmg-genai/</u>.
- Davis, Anthony. "The Future of Law Firms (and Lawyers) in the Age of Artificial Intelligence." *The Professional Lawyer* 27, no. 1 (October 2, 2020): 3–12. <u>https://www.americanbar.org/groups/professional_responsibility/publications/professional_law</u> <u>yer/27/1/the-future-law-firms-and-lawyers-the-age-artificial-intelligence/</u>.
- DBM Vircon. "How Government Mandated BIM Standards Are Being Applied to Projects." *Insights* (blog). May 23, 2023. <u>https://www.dbmvircon.com/how-government-mandated-bim-standards-are-being-applied-to-projects/</u>.
- Deltek. *Clarity: Architecture & Engineering Industry Study, 45th Annual Comprehensive Report.* May 2024. <u>https://www.deltek.com/en/architecture-and-engineering/clarity</u>.
- Dethe, Amol. "Generative AI Can Boost Small Accountancy Firms." *The Economic Times. ETCFO*, May 31, 2024. <u>https://cfo.economictimes.indiatimes.com/news/cfo-tech/generative-ai-can-boost-small-accountancy-firms/110580045</u>.
- Dey, Mithu, and Lesia Quamina. "Surveying a Shifting Landscape." *The CPA Journal*, July 24, 2024. <u>https://www.cpajournal.com/2024/07/24/surveying-a-shifting-landscape/</u>.
- Dorsey, Hilary. "Gensler Among Several Design Firms Experimenting with Al." *Finance & Commerce*, February 26, 2024. <u>https://finance-commerce.com/2024/02/gensler-among-several-design-firms-experimenting-with-ai/</u>.
- Ellis, Lindsay, and Paul Overberg. "Why No One's Going Into Accounting." *The Wall Street Journal*, October 6, 2023. <u>https://www.wsj.com/lifestyle/careers/accounting-salary-cpa-shortage-dec2caa2</u>.

- *Finance Story.* "This Ex-BDO Manager Led Offshore Operations for UK & US Markets. Here Are the Steps to Establish an Offshore Practice." October 16, 2023. <u>https://thefinancestory.com/how-to-build-an-accounting-offshore-firm-in-india</u>.
- Franklin University. "Accounting & Data Analytics: What You Need to Know." Accounting MVP Blog. Accessed December 30, 2024. <u>https://www.franklin.edu/blog/accounting-mvp/accounting-data-analytics</u>.
- Freeman, Becky. "The Rise of Building Information Modelling (BIM) in the UK Construction Industry." Lanview, July 29, 2024. <u>https://www.lanview.co.uk/2024/07/29/the-rise-of-building-information-modelling-bim-in-the-uk-construction-industry/</u>.
- Gaetano, Chris. "2023's Major Trends in Accounting Tech: Cloud, Automation, AI and More." Accounting Today, November 29, 2023. <u>https://www.accountingtoday.com/list/this-year-in-vendor-</u> <u>spotlights-cloud-computing-automation-solutions-artificial-intelligence-and-other-trends</u>.
- Gartner. "Top 5 Audit Trends and Priorities in 2025." Accessed January 25, 2025. https://www.gartner.com/en/audit-risk/trends/audit-trends.
- Gleason, Adam. "How AI Transformed the Tax and Accounting Profession in 2023." *Thomson Reuters*, January 8, 2024. <u>https://tax.thomsonreuters.com/blog/how-ai-transformed-the-tax-and-accounting-profession-in-2023/</u>.
- Handa, Rish, and Sophie Papineau-Wolff. "Law & Al." *Ivey Business Journal*, November/December 2019. <u>https://iveybusinessjournal.com/law-ai/</u>.
- Herath, Siriyama Kanthi, and Destiny Woods. "Impacts of Big Data on Accounting." Centre for Business and Economic Research. *The Business and Management Review* 12, no. 2 (December 2021): 186–93. <u>https://cberuk.com/cdn/conference_proceedings/2022-01-20-18-24-55-PM.pdf</u>
- HMC Architects. "Robotics in Architecture and Construction: An Industry Shift." October 23, 2019. <u>https://hmcarchitects.com/news/robotics-in-architecture-and-construction-an-industry-shift-</u> <u>2019-10-23/</u>.
- Hood, Dan. "The Year Ahead: 2025 in Accounting: A Balancing Act between Confidence and Resistance." *Accounting Today*, December 17, 2024. <u>https://www.accountingtoday.com/research-report/a-balancing-act-between-confidence-and-resistance</u>.
- IBISWorld. Architects in the US. November 2024. <u>https://my.ibisworld.com/us/en/industry/54131/at-a-glance</u>.
- IBISWorld. Engineering Services in the US. August 2024. https://my.ibisworld.com/us/en/industry/54133/at-a-glance.
- IBISWorld. *Global Accounting Services*. May 2024. <u>https://www.ibisworld.com/global/industry/global-accounting-services/1960/</u>.
- IBISWorld. *Global Architectural Services*. June 2024. <u>https://my.ibisworld.com/gl/en/industry/I6721-gl/at-a-glance</u>.

- IBISWorld. *Global Engineering Services*. May 2024. <u>https://my.ibisworld.com/gl/en/industry/l6722-gl/at-a-glance</u>.
- ICAEW Insights. "The Most Desired Skills for AI-Assisted Accountants." ICAEW, November 20, 2024. <u>https://www.icaew.com/insights/viewpoints-on-the-news/2024/nov-2024/the-most-desired-skills-for-aiassisted-accountants</u>.
- International Association of Independent Accounting Firms (INAA). "Addressing the Accountancy Skills Gap." January 24, 2022. https://www.inaa.org/addressing-the-accountancy-skills-gap/.
- International Bar Association and the Center for AI and Digital Policy. *The Future Is Now: Artificial Intelligence and the Legal Profession*. September 2024. <u>https://www.ibanet.org/document?id=The-future-is%20now-AI-and-the-legal-profession-report</u>.
- Invedus. "How Outsourcing Accounting Drives Growth in CPA Firms?" May 12, 2023. https://invedus.com/blog/how-outsourcing-accounting-drives-growth-in-cpa-firms/.
- Jacobs. "Opportunity Awaits within Operations Management and Facilities Services." Accessed February 11, 2025. <u>https://www.jacobs.com/solutions/services/operations-facilities-management/careers</u>.
- Karbon. "The State of AI in Accounting 2024." 2024. <u>https://karbonhq.com/resources/state-of-ai-accounting-report-2024/</u>.
- Karbon. "The State of AI in Accounting 2025." 2025. <u>https://karbonhq.com/resources/state-of-ai-accounting-report-2025/</u>.
- Kasztelnik, Karina, and Stephen Campbell. "The Future of Business Data Analytics and Accounting Automation." *The CPA Journal*, November/December 2023. <u>https://www.cpajournal.com/2024/01/31/the-future-of-business-data-analytics-and-accounting-automation/</u>.
- Kasztelnik, Karina, and Eva Jermakowicz. "Financial Statement Fraud Detection in the Digital Age." *The CPA Journal*, March/April 2024. <u>https://www.cpajournal.com/2024/06/24/financial-statement-fraud-detection-in-the-digital-age/</u>.
- Kenan-Flagler. "Why Data Analytics Matters to Accountants." University of North Carolina, March 23, 2021. <u>https://www.kenan-flagler.unc.edu/perspectives/why-data-analytics-matter-to-accountants/</u>.
- Kennedy, Rachael. "The Accounting Shortage Is Hurting Your Bottom Line." The CFO, August 12, 2024. <u>https://the-cfo.io/2024/08/12/the-accounting-shortage-is-hurting-your-bottom-line/</u>.
- Kenney, Andrew. "Small Firms Find Success with Advisory Services." Journal of Accountancy, August 1, 2024. <u>https://www.journalofaccountancy.com/issues/2024/aug/small-firms-find-success-with-advisory-services.html</u>.
- Kumar, Manoj, Jatindra Dash, and Sumit Khanna. "Global Accounting Firms Set up Shop in India's Smaller Cities." Reuters, July 12, 2023. <u>https://www.reuters.com/business/finance/global-accounting-firms-set-up-shop-indias-smaller-cities-2023-07-12/</u>.

- Lane, Danielle. "Digital Twins in Construction and Architecture." *Freedes Studio*. Accessed January 27, 2025. <u>https://freedesstudio.com/blog/digital-twins-in-construction-and-architecture/</u>.
- Langemo, Bree. "Transforming the Legal Profession Through Technology and Entrepreneurship." American Bar Association. *Law Technology Today*, May 23, 2024. <u>https://www.americanbar.org/groups/law_practice/resources/law-technology-today/2024/transforming-the-legal-profession-through-technology-and-enterpreneurship/</u>.
- Lawton, George. "Generative AI Ethics: 8 Biggest Concerns and Risks." Search Enterprise AI, July 23, 2024. https://www.techtarget.com/searchenterpriseai/tip/Generative-AI-ethics-8-biggest-concerns.
- Majmudar & Partners. "Current State of Play of the Liberalization of India's Legal Market." March 4, 2025. <u>https://www.majmudarindia.com/liberalization-of-indias-legal-market/</u>.
- MarketLine. *Global Legal Services Market Summary, Competitive Analysis and Forecast to 2028*. February 2024. <u>https://store.marketline.com/report/global-overview-of-legal-services-market-analysis/</u>.
- MarketLine. United States (US) Legal Services Market Summary, Competitive Analysis and Forecast to 2028. February 2024. <u>https://store.marketline.com/report/usa-legal-services-market-analysis/</u>.
- Martineau, Kim. "What Is Generative AI?" *IBM Research Blog*. April 20, 2023. <u>https://research.ibm.com/blog/what-is-generative-AI</u>.
- Matterport. "What Is Digital Twin & BIM? Use Cases Explained." Accessed January 27, 2025. https://matterport.com/learn/digital-twin/bim.
- Maurer, Mark. "Big Four Accounting Firms Pare Their Consultant Ranks in Postpandemic Reversal." *The Wall Street Journal*, July 18, 2023. <u>https://www.wsj.com/articles/big-four-accounting-firms-pare-their-consultant-ranks-in-postpandemic-reversal-dc50f9da</u>.
- Maurer, Mark. "Job Security Isn't Enough to Keep Many Accountants from Quitting." *Wall Street Journal*, September 22, 2023. <u>https://www.wsj.com/articles/accounting-quit-job-security-675fc28f</u>.
- Maurer, Mark, and Alexander Saeedy. "Accounting Firms Overhired. Now Some Are Starting to Lay Off Partners." *The Wall Street Journal*, December 15, 2023. <u>https://www.wsj.com/articles/accounting-firms-overhired-now-some-are-starting-to-lay-off-partners-d0c6ee6b.</u>
- Meisels, Michelle, Misha Nikulin, Kate Hardin, Matt Sloane, and Kruttika Dwivedu. "2024 Engineering and Construction Industry Outlook." Deloitte, November 6, 2023. <u>https://www2.deloitte.com/us/en/insights/industry/engineering-and-construction/engineering-and-construction-industry-outlook-2024.html</u>.
- Merken, Sara. "US Legal Industry's China Retreat Gained Steam in 2024." Reuters, December 26, 2024. <u>https://www.reuters.com/legal/legalindustry/us-legal-industrys-china-retreat-gained-steam-2024-2024-12-26/</u>.
- Microsoft. "What Is Cloud Computing?" Accessed December 30, 2024. <u>https://azure.microsoft.com/en-us/resources/cloud-computing-dictionary/what-is-cloud-computing</u>.

- Mok, Maisy. "Flight or Fight?" China Business Law Journal, October 7, 2024. <u>https://law.asia/us-law-firms-exit-china-causes-trends/</u>.
- Mycase. "Industry-Leading Legal Case Management Software and Solutions." Accessed February 3, 2025. <u>https://www.mycase.com/</u>.
- Nasir, Osama. "The Role of Drones in Architecture." *Parametric Architecture*, October 8, 2024. <u>https://parametric-architecture.com/the-role-of-drones-in-</u> <u>architecture/?srsltid=AfmBOoowYQhOeaNEv7zZlsScS8fasgllwkQKS9GF3e_gsOG2KEJ7b1lb</u>.
- Noto, Grace. "Deloitte Brings GenAl to Audit, Accounting Professionals with New Bot." CFO Dive, November 9, 2023. <u>https://www.cfodive.com/news/deloitte-brings-genai-internal-audit-accounting-professionals/699354/</u>.
- Office of the U.S. Trade Representative (USTR). Annex II: Non-Conforming Measures for Services and Investment, Korea Annex II. 2007. <u>https://ustr.gov/sites/default/files/uploads/Countries%20Regions/africa/agreements/korus/Kore</u> <u>a%20Annex%20II.pdf</u>.
- OneClick IT Consultancy. "Data Analytics in Accounting: Complete Guide for 2025." June 19, 2024. https://www.oneclickitsolution.com/blog/data-analytics-in-accounting.
- Organisation for Economic Co-operation and Development (OECD). *OECD Services Trade Restrictiveness* Index (STRI) - Legal Services 2023. January 2024. <u>https://doi.org/10.1787/bae97f98-en</u>.
- Organisation for Economic Co-operation and Development (OECD). OECD Services Trade Restrictiveness Index Simulator. Accessed January 27, 2025. https://sim.oecd.org/Simulator.ashx?lang=En&ds=STRI&d1c=psleg&d2c=kor.
- Pardo, Santiago Bedoya. "US Companies Turn to Global Talent Pool as Accountant Shortage Hits Crisis Point." The Accountant, August 26, 2024. <u>https://www.theaccountant-online.com/news/uscompanies-turn-to-global-talent-pool-as-accountant-shortage-hits-crisis-point/</u>.
- PE Impact. "12 Uses of Drones in Engineering and Construction." *Tuesday Tidbits*. May 2, 2022. <u>https://peimpact.com/12-uses-of-drones-in-engineering/</u>.
- Perkins, Emily. "Beyond Spreadsheets: Generative Al's Role in Reshaping Modern Accounting." Vic.ai, September 26, 2024. <u>https://www.vic.ai/resources/www.vic.ai/resources/beyond-spreadsheets-generative-ais-role-in-reshaping-modern-accounting</u>.
- Pinnacle Infotech. *Global BIM Adoption: A 2024 Outlook*. January 1, 2024. <u>https://pinnacleinfotech.com/global-bim-adoption/</u>.
- Plaschke, Frank, Ishaan Seth, and Rob Whiteman. "Bots, Algorithms, and the Future of the Finance Function." McKinsey & Company, January 9, 2018. <u>https://www.mckinsey.com/capabilities/strategy-and-corporate-finance/our-insights/bots-algorithms-and-the-future-of-the-finance-function#/</u>.

- PricewaterhouseCoopers (PwC). "How PwC Is Using Generative AI to Deliver Business Value." January 2025. <u>https://www.pwc.com/us/en/tech-effect/ai-analytics/generative-ai-impact-on-business.html</u>.
- Principles of Accounting. "The Accounting Cycle and Closing Process." Accessed March 4, 2025. <u>https://www.principlesofaccounting.com/chapter-4/the-accounting-cycle-and-closing-process/</u>.
- Raghunathan, Krishnan. "Five Trends That Will Redefine Finance and Accounting in 2025." Accounting Today, January 7, 2025. <u>https://www.accountingtoday.com/list/five-trends-that-will-redefine-finance-and-accounting-in-2025</u>.
- Rajan, Shanthi. "7 Technological Innovations Shaping the Future of Architecture, Engineering and Construction in 2024." *Buildings*, February 26, 2024. <u>https://www.buildings.com/print/content/33037352</u>.
- Raza, Hasan. "Cloud Accounting in 2024: Pros and Cons." *Expertise Accelerated*. April 30, 2024. <u>https://expertiseaccelerated.com/cloud-accounting-pros-and-cons/</u>.
- Roemer, John. "US Law Firms Are Closing Their China Offices." American Bar Association. ABA Journal (October 1, 2024). <u>https://www.abajournal.com/magazine/article/us-law-firms-are-closing-their-china-offices.</u>
- Roose, Kevin. "The Robots Are Coming for Phil in Accounting." *The New York Times*, March 6, 2021. <u>https://www.nytimes.com/2021/03/06/business/the-robots-are-coming-for-phil-in-accounting.html</u>.
- Royal Institute of British Architects (RIBA). *RIBA AI Report 2024*. February 29, 2024. <u>https://riba-prd-assets.azureedge.net/-/media/GatherContent/Business-Benchmarking/Additional-Documents/RIBA-2024-AI-Report.pdf?rev=7fc210b741f3452b9052b701c243b7dd</u>.
- Sahota, Neil. "The Dawn of a New Era: Ai's Revolutionary Role in Accounting." *Forbes*, April 22, 2024. <u>https://www.forbes.com/sites/neilsahota/2024/04/22/the-dawn-of-a-new-era-ais-revolutionary-role-in-accounting/</u>.
- Salimi, Anwar Y. "Degree of Inclusion of Data Analytics in Different Courses in the Accounting Curriculum and Differences in the Knowledge and Preparation of Faculty Teaching Those Courses." *Journal of Business & Educational Leadership* 14, no. 1 (March 1, 2024): 40–48. <u>http://asbbs.org/files/2023-</u> 24/JBEL_14.1_Spring_2024.pdf.
- Sang, Ray, and Clay Kniepmann. "Al and Fraud: What CPAs Should Know." Journal of Accountancy, May 1, 2024. <u>https://www.journalofaccountancy.com/issues/2024/may/ai-and-fraud-what-cpas-should-know.html</u>.
- SAPRO. "The Accounting Skills Shortage: Innovatively Navigating the Gap." SAPRO, May 6, 2024. https://www.sapro.com/blog/the-accounting-skills-shortage-innovatively-navigating-the-gap.
- Schweitzer, Brandon. "Artificial Intelligence (AI) Ethics in Accounting." *Journal of Accounting, Ethics & Public Policy* 25, no. 1 (March 15, 2024): 67–103. https://jaepp.org/index.php/jaepp/article/view/349/329.

- Seah, Jessica. "Chinese Law Firms Are in Trouble as Downward Spiral Show No Signs of Easing." Law.com International (September 16, 2024). <u>https://www.law.com/international-</u> edition/2024/09/16/the-2024-global-200-chinese-law-firms-are-in-trouble-as-downward-spiralshow-no-signs-of-easing/.
- Sen, Arindam. "How India Is Gearing up for a US\$110b GCC Industry by 2030." EY, June 8, 2023. <u>https://www.ey.com/en_in/insights/consulting/global-capability-centers/how-india-is-gearing-up-for-a-us-110b-dollars-gcc-industry-by-2030</u>.
- Sharma, Kudeep. "What Is the Salary of an Accountant in India in 2023." *Jobaaj* (blog), April 1, 2024. <u>https://www.jobaaj.com/blog/what-is-the-salary-of-an-accountant-in-india-in-2023</u>.
- Shaybak, Natali. "Legal Tech v. Legal Outsourcing and ALSP." Lawflex, May 10, 2022. https://lawflex.com/legal-tech-v-legal-outsourcing-and-alsp/.
- Sherman, Ella. "Legal Tech Startups That Law Firms Invested in Throughout 2024." Law.com Legaltech News, December 18, 2024. <u>https://www.law.com/legaltechnews/2024/12/18/legal-tech-</u> <u>startups-that-law-firms-invested-in-throughout-2024/</u>.
- Singh, Shweta. "Robotic Process Automation in Accounting: A Comprehensive Guide." Savant, September 20, 2024. <u>https://www.savantlabs.io/blog/robotic-accounting</u>.
- Soto, Cinthia. "Al in Architecture: Benefits, Use Cases, and Examples." *OpenAsset*, October 25, 2024. <u>https://openasset.com/blog/ai-and-architecture/</u>.
- Soto, Cinthia. "Understanding the AEC Industry + Top 17 AEC Trends for 2024." *OpenAsset*, April 20, 2024. <u>https://openasset.com/blog/aec-industry-trends/</u>.
- Sottolano, Becky. "Beyond the Numbers: Why AI Can't Replace Human Judgment in Accounting." *Trullion*, September 11, 2024. <u>https://trullion.com/blog/beyond-the-numbers-why-ai-cant-replace-human-judgment-in-accounting/</u>.
- Spotlight Reporting. *Global Advisory Trends Report: Realise Success in the Changing World of Accounting*. 2023. <u>https://www.spotlightreporting.com/whitepaper/global-advisory-trends-report-2023-</u>2024.
- Statista. Legal Tech. 2023. https://www.statista.com/study/84775/legal-tech/.
- Stuart, Toby E. "Could Gen Al End Incumbent Firms' Competitive Advantage?" *Harvard Business Review*, November 21, 2024. <u>https://hbr.org/2024/11/could-gen-ai-end-incumbent-firms-competitive-advantage</u>.
- Teja, Natasha. "Global Firms' Asia Offices Lag Behind US and Europe in Al Uptake." *Law.com Legaltech News*, August 26, 2024. <u>https://www.law.com/international-edition/2024/08/26/global-firms-</u> <u>asia-offices-lag-behind-us-and-europe-in-ai-uptake/</u>.
- Tetra Tech. "Life at Tetra Tech." Accessed February 11, 2025. <u>https://www.tetratech.com/careers/life-at-tetra-tech/#professional-development</u>.

- Thomson Reuters. 2024 Generative AI in Professional Services. February 27, 2024. https://www.thomsonreuters.com/en/reports/2024-generative-ai-in-professional-services.html.
- Thomson Reuters. "How Do Different Accounting Firms Use AI?" *Tax & Accounting*, June 3, 2024. <u>https://tax.thomsonreuters.com/blog/how-do-different-accounting-firms-use-ai/</u>.
- Thomson Reuters Institute, the Center on Ethics and the Legal Profession at Georgetown Law, and the Saïd Business School at the University of Oxford. *Alternative Legal Services Providers 2023*. 2023. <u>https://legal.thomsonreuters.com/en/insights/reports/alternative-legal-services-providers-report-2023/form</u>.
- Thomson Reuters Institute, the Center on Ethics and the Legal Profession at Georgetown Law, and the Saïd Business School at the University of Oxford. *Alternative Legal Services Providers 2025*. 2025. <u>https://www.thomsonreuters.com/en/press-releases/2025/january/alternative-legal-services-providers-2025-report-shows-segment-comprises-28-billion-of-the-legal-market</u>.
- Thomson Reuters Law Blog. "Impact of AI on Law Firms of Every Size." August 15, 2023. https://legal.thomsonreuters.com/blog/ais-impact-on-law-firms-of-every-size/.
- Turney, Drew. "What Is a Digital Twin? How Intelligent Data Models Can Shape the Built World." Autodesk. December 11, 2025. <u>https://www.autodesk.com/design-make/articles/what-is-a-digital-twin?msockid=0847337cf6866ac521f8267ff7366b9e</u>.
- U.S. Census Bureau (Census). "Accounting, Tax Preparation, Bookkeeping, and Payroll Services: U.S. Total." Accessed January 8, 2025. <u>https://www.census.gov/econ/currentdata/?programCode=QSS&startYear=2003&endYear=2025</u> <u>&categories[]=5412T&dataType=QREV&geoLevel=US&adjusted=1¬Adjusted=1&errorData=0</u> <u>#table-results.Business and Industry, accessed January 8, 2025</u>.
- U.S. Department of Commerce (USDOC). Bureau of Economic Analysis (BEA). Table 2.3. "U.S. Trade in Services, by Country or Affiliation and by Type of Service." July 3, 2024. <u>https://apps.bea.gov/iTable/?ReqID=62&step=1& gl=1*i1kcnx*_ga*MjM5MTY3NzYuMTcyNzEw</u> <u>MDc5MA..*_ga_J4698JNNFT*MTczODY4ODkwMS44LjEuMTczODY4ODkxNi40NS4wLjA.#eyJhcHB</u> <u>pZCI6NjIsInN0ZXBzIjpbMSw5LDEwXSwiZGF0YSI6W1siUHJvZHVjdCIsIjQiXSxbIIRhYmxITGIzdCIsIjM</u> wNTgzIl1dfQ==.
- U.S. Department of Commerce (USDOC). Bureau of Economic Analysis (BEA). Table 4.1. "Services Supplied to Foreign Persons by U.S. MNEs Through Their MOFAs, by Industry of Affiliate and by Country of Affiliate." October 8, 2024. <u>https://apps.bea.gov/iTable/?ReqID=62&step=1#eyJhcHBpZCI6NjIsInN0ZXBzIjpbMSw5LDZdLCJkY</u> <u>XRhIjpbWyJQcm9kdWN0liwiNCJdLFsiVGFibGVMaXN0liwiMzYzII1dfQ==</u>.
- U.S. Department of Labor (USDOL). Bureau of Labor Statistics (BLS). "Accountants and Auditors." Occupational Employment and Wage Statistics, April 3, 2024. https://www.bls.gov/oes/2023/may/oes132011.htm.
- U.S. Department of Labor (USDOL). Bureau of Labor Statistics (BLS). "Employment, Hours, and Earnings from the Current Employment Statistics Survey (National)." Accessed May 12, 2025. <u>https://www.bls.gov/help/one_screen/ce.htm</u>.

- U.S. Department of Labor (USDOL). Bureau of Labor Statistics (BLS). "Labor Force Statistics-Employed -Accountants and Auditors." Current Population Survey. Accessed January 8, 2025. <u>https://data.bls.gov/dataViewer/view/timeseries/LNU02038217</u>.
- U.S. Department of Labor (USDOL). Bureau of Labor Statistics (BLS). "May 2023 National Occupational Employment and Wage Estimates." Accessed March 4, 2025. <u>https://www.bls.gov/oes/current/oes_nat.htm#17-0000</u>.
- Vasarhelyi, Miklos, and Andrea Rozario. "How Robotic Process Automation Is Transforming Accounting and Auditing." The CPA Journal, July 2, 2018. <u>https://www.cpajournal.com/2018/07/02/howrobotic-process-automation-is-transforming-accounting-and-auditing/</u>.
- Wirth, Gregg. "Law Firm COO & CFO Forum: Al's Potential Impacts on the Legal Industry." Thomson Reuters Institute, November 7, 2024. <u>https://www.thomsonreuters.com/en-us/posts/legal/law-firm-coo-cfo-forum-ai-potential-impacts/</u>.
- Wolters Kluwer. *The 2024 Wolters Kluwer Future Ready Lawyer Report*. 2024. <u>https://www.wolterskluwer.com/en/know/future-ready-lawyer-2024</u>.
- Wood, Matt, Megan Weis, Steve Casey, Jennifer Gray, Joe Pistacchio, and Neil Stewart. "2024 CFO Talent Pulse Survey." Personiv, 2024. https://2399470/CFO%20Talent%20Survey%20-%202024/Report%20Files/2024%20CFO%20Talent%20Pulse%20Survey.pdf? <a href="https://steentwice.nstruetwi
- Zewe, Adam. "Explained: Generative AI." Massachusetts Institute of Technology. MIT News, November 9, 2023. <u>https://news.mit.edu/2023/explained-generative-ai-1109</u>.

Recent Trends in U.S. Services Trade: 2025 Annual Report

Chapter 4 Demand for Professional Services: Changing Demographics, Income Growth, and Continued Digitalization of Businesses

This chapter highlights broad trends that are driving global demand for professional services, including changing demographics in various markets, rising incomes and spending power in China and other emerging markets, businesses' ongoing digitalization, and the move by many manufacturing firms to reorganize their supply chains in the wake of the global pandemic. The focuses are on recent developments in three major professional services sectors: education services, management consulting services, and advertising services.

In recent years, aging populations and declining birth rates in some countries have profoundly affected demand for professional services. These circumstances may reduce overall demand for education services and may prompt more advertising services aimed at the elderly.³⁹⁸ Globally, people aged 65 and above were 10.0 percent of the total population in 2023, up from 7.1 percent in 2003.³⁹⁹ In the United States, people aged 65 or older were less than 5.0 percent of the total population in 1920 but increased to 16.7 percent by 2020.⁴⁰⁰ China currently has the most people aged 65 or older among all countries, with 216.8 million in 2024; one source projects that this number may increase to 400 million by 2050.⁴⁰¹ Aging is expected to lower economic growth and labor capacities and increase government spending and demand for senior housing, among other consequences. However, chronological age does not necessarily correspond with physiological functioning, retirement rates, or other important factors that are affected by increased longevity.⁴⁰²

Additionally, stronger demand for professional services—from both businesses and individuals—is typically associated with substantial income growth in an economy. Some estimates suggest that from 2019 to 2023, professional services revenue grew by 39.2 percent in China and by 45.9 percent in India (in comparison, GDP per capita grew by 24.4 percent in China and 21.5 percent in India).⁴⁰³ By one

³⁹⁸ A "demographic cliff" refers to a decrease in college enrollment due to declining birth rates. Gradstein and Kaganovich, "Aging Population and Education Finance," December 2004; Eisend, "Older People in Advertising," July 15, 2022; Marcus, "A Looming 'Demographic Cliff," January 8, 2025.

³⁹⁹ World Bank, "World Development Indicators," accessed January 6, 2025.

⁴⁰⁰ Caplan, "U.S. Older Population Grew From 2010 to 2020," May 25, 2023.

 ⁴⁰¹ From 2022 to 2023, the number of kindergartens in China fell by an estimated 14,800. Coulson, "Ensuring All Older People Can Benefit," February 2, 2024; Hawkins, "China's Kindergarten Numbers Shrink," October 28, 2024.
⁴⁰² Kotschy et al., "On the Limits of Chronological Age," November 2024.

⁴⁰³ Statista, "Professional Services—China," June 2024; "Professional Services—India," June 2024; World Bank, "World Development Indicators," accessed January 6, 2025. China and India were among the countries categorized as emerging markets in USITC, *Recent Trends in U.S. Services Trade: 2021 Annual Report*, April 2021, 155–56.

estimate, the annual revenue growth of professional services globally was 10.4 percent in 2022, a slight increase from 9.7 percent in 2018.⁴⁰⁴

Digitalization (the adoption and integration of digital technologies into activities across the economy) is significantly changing professional services business models. For example, it especially drives demand for management consultants and online education services. Digitalization is also spurring a shift in advertising services from print to social media and other online channels (discussed below). Professional services firms, along with information technology, media, and financial services firms, started adopting digital assets relatively early, in contrast to construction and health-care services, which lagged.⁴⁰⁵ The shift toward delivery of services through online channels was accelerated by the COVID-19 pandemic, which motivated remote work; digitalization (especially by small firms) increased sharply in 2020 compared to 2015–19.⁴⁰⁶ One analysis published in 2023 found that 92 percent of online U.S. job postings now require digital skills.⁴⁰⁷

In the education sector, foreign student enrollments remain strong at U.S. colleges and universities, although schools are seeing a significant decline in domestic enrollments, as well as a sharp increase in online education after the COVID-19 pandemic.⁴⁰⁸ Demand for management consulting services has been supported by companies' increased need for information technology consulting, as firms pursue accelerated digital transformations, adopt artificial intelligence (AI) and generative AI (GenAI) technologies, and reorganize their supply chains.⁴⁰⁹ Firms in the advertising sector have seen large changes in advertising channels, with a decline in traditional broadcast television and a rise in streaming, significant growth in social media, and increased adoption of AI.⁴¹⁰

Education Services: U.S. Exports Are Delivered Through Multiple Modes

Education services, as defined by the Bureau of Economic Analysis, include formal academic instruction at primary, secondary, and postsecondary programs at educational institutions in the United States. This section focuses on education services provided by postsecondary institutions, primarily because international students studying at colleges and universities represent the vast majority of international trade in education services (see chapter 2). Such trade, termed mode 2 or consumption abroad, occurs when a student from one country travels to another country for university-level study; it consists of expenditures by students and their dependents, if any, for tuition, fees, and living expenses while studying in a foreign country. As an example, expenditures incurred by an Italian student studying in the

⁴⁰⁶ Jaumotte et al., "How Pandemic Accelerated Digital Transformation in Advanced Economies," March 21, 2023.

⁴⁰⁴ Bohne, "Revenue Growth of Professional Services," July 12, 2023.

⁴⁰⁵ Gandhi et al., "Which Industries Are the Most Digital (and Why)?," April 1, 2016.

⁴⁰⁷ Bergson-Shilcock and Taylor, "Closing the Digital Skill Divide," February 2023, 4.

⁴⁰⁸ Fischer, "The Shrinking of Higher Ed," August 12, 2022; McMurtrie, "Classroom Walls Are Shifting," October 18, 2023.

 ⁴⁰⁹ Laffitte, "2024 In Review," November 25, 2024; IBISWorld, *Global Management Consultants*, December 2024.
⁴¹⁰ Feger, "Linear TV: What It Is," September 6, 2024; Harlow, "How Effective Are Ads on Social Media?," June 5, 2023.

Chapter 4: Demand for Professional Services: Changing Demographics, Income Growth, and Continued Digitalization of Businesses

United States are considered U.S. exports of education services, whereas expenditures incurred by a U.S. student studying in Italy are considered U.S. imports.

Trade in education services takes place via the other three modes of supply as well. Education services imported cross-border (mode 1) occur when a student in one country accesses education services in another country remotely, usually via the internet,⁴¹¹ a phenomenon that has grown dramatically since the onset of the COVID-19 pandemic. Trade via commercial presence (mode 3) occurs when universities establish branch campuses in foreign countries. Mode 4 trade (temporary presence of natural persons) occurs when an educator travels from one country to another to deliver education services, most commonly visiting professors and lecturers.

The primary service offered by universities is educational instruction, involving a specified course of study that is certified by an academic degree upon completion.⁴¹² The longevity of many universities,⁴¹³ as well as high start-up costs and market saturation in many countries, positions education services as a mature industry. In recent years, the enrollment of traditional-age students (18–24 years old) has declined precipitously, a trend that is expected to continue over the near term.⁴¹⁴ As a result, colleges and universities have sought to boost demand through ongoing efforts to recruit international students as well as tap underserved market segments, including mature students returning to complete a college degree, working students pursuing a degree part-time, and working professionals seeking to advance in—or change—their careers by obtaining a certification or degree.

U.S. Exports of Education Services Driven by Strong Enrollments of International Students

Foreign students are an important market segment for U.S. colleges and universities at both the undergraduate and graduate levels. Over the past 25 years, exports of education services have surged as many U.S. colleges and universities sought to attract international students.⁴¹⁵ Historically, universities sought to attract such students to increase campus diversity, thereby increasing the educational experience for both foreign and domestic students.⁴¹⁶ During the course of the 20th century, U.S. universities also sought to attract international students as a tool of informal cultural diplomacy, especially through the Fulbright Program.⁴¹⁷

⁴¹¹ Education services delivered online are included in the BEA category "Other personal, cultural, and recreational services."

⁴¹² Common academic degrees worldwide include bachelor's, master's, and doctoral degrees as well as specialized postgraduate degrees in the legal and health-care fields.

⁴¹³ Most modern universities were founded in the 1800s or the early to mid-1900s. Although the characteristics that define a university are subject to debate, the University of Bologna, which was founded in Bologna, Italy, in 1088, is generally considered to be the oldest university in the Western World. The oldest university in the United States is Harvard University, which was founded in 1636.

⁴¹⁴ Burke, "School's Out," July 29, 2024.

⁴¹⁵ Fischer, "The Shrinking of Higher Ed," August 12, 2022.

⁴¹⁶ Fischer and Aslanian, "Fading Beacon," August 2, 2021.

⁴¹⁷ USDOS, "What Is the Fulbright U.S. Student Program," accessed January 27, 2025; Fischer and Aslanian, "Fading Beacon," August 2, 2021.

More recently, U.S. universities, particularly—but not exclusively—public universities, sought to attract international students as a way to both boost declining domestic enrollments and offset funding shortfalls stemming from budget cuts by many state governments.⁴¹⁸ Indeed, research conducted by the National Bureau for Economic Research found that a 10 percent decrease in state-level educational appropriations at research-intensive institutions, including public flagship universities, was accompanied by a 17 percent increase in the enrollment of international students.⁴¹⁹ Overall, foreign students are often courted by universities because they typically pay not only full tuition rates—often from personal resources—but also for a suite of additional services, including accommodation and dining in university-run facilities as well as extra tuition for remedial courses, mainly English classes.⁴²⁰

During the 2023/24 academic year, the number of post-secondary international students studying in the United States increased by 6.6 percent to a record high of 1.1 million students, or 5.9 percent of total students in the United States. The top sources of such students were India (29.4 percent), China (24.6 percent), South Korea (3.8 percent), Canada (2.6 percent), and Taiwan (2.1 percent).⁴²¹ Updated student enrollment data published in April 2025 indicate that there were 130,600 fewer foreign students studying in the United States in March 2025, compared to March 2024. This decline in enrollments was driven by India, which fell by 28 percent, and West Africa, down 21 percent. In contrast, Chinese enrollments actually increased by 3 percent over the period, making it the largest source of foreign students in the United States in March 2025. Several factors are likely driving the decline in U.S. foreign student enrollments, including the high cost of studying in the United States, competition from colleges and universities in other countries, and geopolitical tensions. Student visa denials, which reached record highs in the United States in the fall of 2024, are also likely a factor, with countries in South Asia and West Africa having higher-than-average rejection rates.⁴²²

In the United States, 44.6 percent of international students were enrolled in graduate-level studies during the 2023/24 academic year, followed by those in undergraduate programs (30.4 percent), optional practical training⁴²³ (21.5 percent), and non-degree programs or classes (3.4 percent).⁴²⁴ The bulk of international students studied in California (140,858) and New York (135,813), followed by Texas (89,546), Massachusetts (82,306), and Illinois (62,299).

During the 2023/24 academic year, the U.S. institutions hosting the largest number of international students were New York University (New York; 27,247 students), Northeastern University–Boston (Massachusetts; 21,023 students), Columbia University (New York; 20,321), Arizona State University

⁴¹⁸ Fischer and Aslanian, "Fading Beacon," August 2, 2021.

⁴¹⁹ Fischer, "The Shrinking of Higher Ed," August 12, 2022.

⁴²⁰ Fischer and Aslanian, "Fading Beacon," August 2, 2021; Fischer, "How International Education's Golden Age Lost Its Sheen," March 28, 2019.

⁴²¹ Open Doors, *2024 Fast Facts*, November 25, 2024.

⁴²² Fischer, "Latitudes," April 9, 2025.

⁴²³ Open Doors, *2024 Fast Facts*, November 25, 2024. Optional practical training is a program that authorizes international students holding an F-1 visa to work in the United States (for a period of up to one year) in employment related to their degree program. The number of optional practical training students in the United States decreased from 223,539 in the 2019/20 academic year to 184,759 during the 2021/22 academic year but rebounded to 242,782 during the 2023/24 academic year.

⁴²⁴ Open Doors, 2024 Fast Facts, November 25, 2024.

Chapter 4: Demand for Professional Services: Changing Demographics, Income Growth, and Continued Digitalization of Businesses

(Arizona; 18,430 students), and the University of Southern California (California; 17,469 students).⁴²⁵ Overall, international students contributed an estimated \$43.8 billion to the U.S. economy for tuition, fees, housing, transportation, insurance, telecommunications services, and living expenses during the 2023/24 academic year and created or supported an estimated 378,175 jobs in the United States.⁴²⁶

In 2022, the United States was the top destination for international students worldwide, attracting 16 percent of the 6.9 million students who studied outside their home countries. Other top destinations were Canada (12 percent), the United Kingdom (UK) (11 percent), Australia (6 percent), and France (6 percent).⁴²⁷ Many international students seek to study in the United States because of the large number of highly regarded universities and academic programs as well as the opportunity to engage in world-leading research.⁴²⁸

During the 2022/23 academic year, the most current period for which data are available, 280,716 U.S. students studied for academic credit outside the United States, a figure that grew by 48.7 percent compared to the previous academic year but still down from the pre-COVID-19 pandemic peak. Of the 2022/23 total, 97.6 percent of U.S. students were enrolled in short-term programs or classes of less than one academic or calendar year.⁴²⁹ Overall, U.S. students preferred destinations in Europe; Italy (14.9 percent), the UK (12.5 percent), Spain (11.6 percent), France (6.1 percent), and Ireland (3.5 percent) hosted the largest number of U.S. students. Additionally, 25,692 U.S. students participated in noncredit learning activities abroad, including academic research, internships, and volunteering, and 14,890 students engaged in online activities like online internships and project-based study as well as academic conferences.⁴³⁰

Increases in Online Education Help Offset Declining Enrollments

Many U.S. colleges and universities have developed online programs to offset declining domestic inperson enrollments. This trend has accelerated since the COVID-19 pandemic, as nearly all educational institutions now have experience with online delivery of courses.

Despite isolated examples of universities attempting to offer classes or academic programs over the internet in the 1990s and early 2000s, the relatively underdeveloped technology of the era—including lower levels of computer chip processing power, immature long-haul telecommunications networks, low-bandwidth (dial-up modem) internet access technologies, and lower levels of personal computer penetration—limited widespread adoption. The broad-based rollout of online education classes and programs, facilitated by broadband telecommunications technologies capable of supporting streaming video, started in earnest roughly 15 years ago with the emergence of massive open online courses.

⁴²⁵ Open Doors, 2024 Fast Facts, November 25, 2024.

⁴²⁶ NAFSA, *The United States of America Benefits from International Students*, November 18, 2024.

⁴²⁷ Project Atlas, *Global Mobility Trends*, 2024.

⁴²⁸ Shorelight, "U.S. Loses International Student Market Share," May 22, 2023.

⁴²⁹ Open Doors, *2024 Fast Facts*, November 25, 2024.

⁴³⁰ Open Doors, 2024 Fast Facts, November 25, 2024.

During this period, websites like Academic Earth, Open Culture, and the Khan Academy offered videos of professors teaching classes ranging from corporate finance to the history of the Roman Empire.⁴³¹ Although viewers were unable to earn credit—and video quality and class selection were often limited—these early sites attracted attention because they featured top-level academic instruction from some of the world's leading universities, including Harvard University, the Massachusetts Institute of Technology, and Stanford University. During 2010–12, a new breed of massive open online courses emerged with the launch of several high-profile websites, namely Coursera, edX, and Udacity.⁴³²

Although a growing number of universities developed online classes and even degree programs over the next five or six years, much of academia remained cautious, even skeptical, of online programs, particularly degree-granting programs, although some introduced limited online course offerings that were eligible for academic credit. However, the stance of colleges and universities regarding online education changed dramatically with the arrival of the COVID-19 pandemic in March 2020, which led to the complete cessation of in-person learning in most colleges and universities in the United States (and around the world).⁴³³ Almost overnight, the delivery of educational content shifted almost entirely to a format in which professors lectured homebound students over web-based conferencing platforms, notably Zoom.⁴³⁴ So-called "Zoom classes" were standard at most colleges and universities through the 2020/21 and 2021/22 academic years, with many universities resuming in-person teaching in fall 2022.⁴³⁵

The experience of the pandemic not only normalized the online delivery of university-level classes at most colleges and universities but also led to the development of proficiencies—at both the instructor and institutional levels—related to creating and delivering online classes.⁴³⁶ As part of this process, most colleges and universities were also required to acquire and learn how to operate a set of technologies necessary to deliver online classes, including cameras, microphones, lights, and related studio equipment.⁴³⁷

Following the resumption of in-person instruction at most universities, the question became what role online education would occupy going forward. Many residential colleges and universities sought to reestablish in-person, classroom-based learning, with early adopters resuming in-person classes in 2021 and a majority resuming in-person status for the 2022/23 academic year.⁴³⁸ During fall 2022, for example, Princeton University informed its students that they should report to campus for in-person

⁴³¹ USITC, *Recent Trends 2013*, July 2013, 3–4.

⁴³² USITC, *Recent Trends 2013*, July 2013, 3–4.

⁴³³ National Center for Education Statistics, "84% of All Undergraduates Experienced Some or All Their Classes Moved to Online-Only Instruction Due to the Pandemic," June 16, 2021; Costa et al., "Moving Education Online During the COVID-19 Pandemic," October 25, 2021.

⁴³⁴ Gillis and Krull, "COVID-19 Remote Learning Transition in Spring 2020," October 2020.

⁴³⁵ Spar, "Today's Awkward Zoom Classes Could Bring a New Era of Higher Education," September 10, 2020.

⁴³⁶ McMurtrie, "Classroom Walls Are Shifting," October 18, 2023.

⁴³⁷ McMurtrie, "Classroom Walls Are Shifting," October 18, 2023.

⁴³⁸ BestColleges, "Will College Campuses Return to Normal in Fall?," September 22, 2022; Lu, "Some Colleges Are Ending Hybrid Learning," April 11, 2022; Roberts-Grmela, "More Students Want Virtual-Learning Options," June 5, 2023.

Chapter 4: Demand for Professional Services: Changing Demographics, Income Growth, and Continued Digitalization of Businesses

learning and that professors were no longer required to accommodate remote students.⁴³⁹ Increasingly, however, a growing number of students are demanding options for online classes.⁴⁴⁰

Working professionals have long preferred online classes for their convenience. A growing number of traditional-age students (18–22), who developed a taste for online classes during the pandemic, have stated preferences for at least some wholly online classes as well as so-called "hybrid" courses, which allow students to attend classes via a set schedule of in-person attendance and online sessions.⁴⁴¹ A survey of more than 800 students by Educause in spring 2022, for example, found that 41 percent of students preferred in-person classes (or mostly in-person classes), compared to 65 percent of students before the pandemic.⁴⁴² Similarly, 59 percent of chief online officers surveyed by Quality Matters and Eduventures Research noticed increased interest in online classes among the 18–24 year-old age cohort.⁴⁴³

Some universities have decided that offering an increased array of online classes—including both synchronous and asynchronous classes⁴⁴⁴—and programs may help to increase or maintain enrollments. Online classes and degrees also have the potential to increase or maintain student enrollment numbers by allowing students with unique circumstances to take classes and earn degrees, including students with disabilities, working students and commuters, and students caring for family members.⁴⁴⁵ Some small and regional colleges are also attempting to boost enrollments by updating academic programs and developing (or expanding) online learning programs, often for the purpose of attracting nontraditional students. Indeed, many colleges now see nontraditional candidates—i.e., working adults over the age of 24 who do not have at least an undergraduate degree—as an important target demographic in their efforts to boost student enrollments.⁴⁴⁶ As a group, nontraditional students offer significant potential. For example, a recent survey by Gallup revealed that 56 percent of college dropouts would be interested in reenrolling, a group that numbered more than 39 million people in the United States. In addition, this same poll revealed that 40 percent of adults who had never attended college would consider enrolling in college, mainly for an associate's degree or certificate program.⁴⁴⁷

International students also represent an important potential market for online education. According to the Institute for International Education, during the 2023/24 academic year—the first year for which

⁴³⁹ Princeton University, "Incoming Students for Fall 2022," accessed January 23, 2025.

⁴⁴⁰ Roberts-Grmela, "More Students Want Virtual-Learning Options," June 5, 2023

⁴⁴¹ Roberts-Grmela, "More Students Want Virtual-Learning Options," June 5, 2023.

⁴⁴² McMurtrie, "Classroom Walls Are Shifting," October 18, 2023.

⁴⁴³ Legon and Garrett, *Student Demand Moves Higher Ed*, 2023, 17, 49. The term Chief Online Officer was created by the Quality Matters and Eduventures researchers to describe the growing number of senior leadership positions responsible for online activities at institutions of higher education. Such leaders have a variety of job titles, and their roles may also include offline responsibilities.

⁴⁴⁴ Stanford, "What Is Synchronous and Asynchronous Learning?," accessed January 22, 2025. Synchronous learning is defined as online instruction in which instructors and students meet at the same time, usually for live-streaming lectures and video-conference discussions. In contrast, in asynchronous classes, students view prerecorded lectures and interact via discussion boards.

⁴⁴⁵ Roberts-Grmela, "More Students Want Virtual-Learning Options," June 5, 2023.

⁴⁴⁶ Gardner and June, "The Perilous Predicament of the Very Small College," August 17, 2022; Fischer, "The Shrinking of Higher Ed," August 12, 2022.

⁴⁴⁷ Fischer, "The Shrinking of Higher Ed," August 12, 2022.

such data are available—18,129 international students residing in their home countries were enrolled in online programs offered by U.S. colleges and universities.⁴⁴⁸ Although the number of international students living abroad and enrolled in an online program is currently a very small share of foreign students consuming U.S. education services, some industry observers believe that online programs will grow in popularity over the next few years. In addition, the statistic reported by the Institute of International Education does not include students enrolled in programs that only offer online programs, with such programs ranging from Coursera to Western Governors University.⁴⁴⁹

International Branch Campuses

An international branch campus (IBC) is defined as "an entity that is owned, at least in part, by a foreign higher education provider, operated in the name of the foreign education provider, and provides an entire academic program, substantially on site, leading to a degree awarded by the foreign education provider."⁴⁵⁰ Such campuses usually take the name of the founding institution, often to connect the new college or university with the established reputation of the founding institution, ⁴⁵¹ including Georgetown University Qatar (Qatar), Duke Kunshan University (China), and New York University's branch campuses in China and the United Arab Emirates (UAE). IBCs, which proliferated in the 1990s, were almost exclusively a phenomenon of large U.S. universities with global name recognition and large endowments. In more recent years, following the precedent set by U.S. universities, non-U.S. universities also started to set up IBCs.⁴⁵² As of March 2023, the latest year for which data are available, IBCs numbered 333 worldwide, characterized by institutions in 39 home countries with established branches in 83 host countries.⁴⁵³

Historically, the earliest U.S. IBCs were one-off endeavors based on foreign relationships as well as the specific mission of the founding institution. After World War II, the United States used branch campuses as a form of informal diplomacy around the world, helping the United States establish postwar international relations norms.⁴⁵⁴ In the 1980s, many IBCs were established in Japan as a means to build relations with a rising economic power.⁴⁵⁵ More recently, founding institutions are motivated to establish IBCs to both promote their image abroad and develop new sources of revenue. For host countries, the motivation is usually to build capacity in higher education.⁴⁵⁶

Although brand-name institutions tend to garner the most attention, fewer than 10 IBCs are affiliated with the top 50 U.S. home-country institutions, as ranked by *U.S. News and World Report*.⁴⁵⁷ Far more

⁴⁴⁸ Open Doors, 2024 Fast Facts, November 25, 2024.

⁴⁴⁹ Interview by USITC staff, February 13, 2025.

⁴⁵⁰ Kinser and Lane, *C-BERT International Campus Listing*, March 23, 2023.

⁴⁵¹ Chada, "What Are International Branch Campuses?," June 11, 2024.

⁴⁵² Chada, "What Are International Branch Campuses?," June 11, 2024.

⁴⁵³ Kinser and Lane, *C-BERT International Campus Listing*, March 23, 2023. In 2023, founding institutions in the United States had established the largest number of IBCs (84), followed by the United Kingdom (46), France (38), Russia (39), and Australia (20).

⁴⁵⁴ Toner, "Branching Out," September 1, 2019.

⁴⁵⁵ Toner, "Branching Out," September 1, 2019.

⁴⁵⁶ Toner, "Branching Out," September 1, 2019.

⁴⁵⁷ Long and Panangipalli, "Under the Radar," Summer 2023.

Chapter 4: Demand for Professional Services: Changing Demographics, Income Growth, and Continued Digitalization of Businesses

commonly, IBCs are established by regional or specialty institutions. Webster University, a private, regional university based in St. Louis, Missouri, for example, operates five IBCS in Europe, two IBCs in China, and 10 IBCs in Uzbekistan. Similarly, Empire State University offers several dozen bachelor's degree programs in Albania, Czechia, the Dominican Republic, Greece, Lebanon, and Türkiye. Small, specialized degree programs are also common, with examples including the Berklee College of Music Valencia Campus (Spain), Embry-Riddle Aeronautical University Europe (Germany), and Parsons Paris (France). IBCs that specialize in a classic liberal arts curriculum are far less common, although notable examples include Bard College Berlin (Germany), McDaniel College Budapest (Hungary), and Saint Louis University–Madrid (Spain).⁴⁵⁸

Despite the proliferation of branch campuses in recent years, many IBCs face serious challenges. First, many universities underestimated the sheer amount of money, time, effort, and infrastructure required to establish an IBC.⁴⁵⁹ Second, some universities have struggled to obtain accreditation, with the process often differing from established U.S. accreditation standards.⁴⁶⁰ In some cases, host-country accreditation is not recognized in the United States, or some third countries.⁴⁶¹ Third, the finances of many IBCs are largely dependent on host-country financing and student enrollment, with even some mature institutions struggling to attract enrollments of more than 500 students.⁴⁶² Relatedly, many IBCs have not been able to attract sufficient financial resources through fundraising avenues, such as alumni donations or research grants.⁴⁶³

IBCs that fail to adapt to local conditions or establish sufficient enrollments face closure. Indeed, of the roughly 100 IBCs established by U.S. institutions, at least 25 have permanently closed between 2000 and 2015.⁴⁶⁴ Overall, a cumulative total of 58 IBCs had closed permanently by the end of March 2023.⁴⁶⁵ One example of an IBC that closed was the branch campus operated by George Mason University (GMU) in Ras Al Kaimah, UAE. During its time in the UAE, GMU was not able to attract a sufficiently qualified pool of students for their programs, with a maximum achieved enrollment of only 120 students. GMU's campus in the UAE closed in 2009, with no students having graduated from its programs.⁴⁶⁶ Other IBCs that have closed over the past 15 years include Yale-NUS⁴⁶⁷ (Singapore), New York University Tisch School of the Arts Asia (Singapore), New York Institute of Technology (Bahrain), and Suffolk University Dakar Campus (Senegal).⁴⁶⁸

Of the 333 IBCs worldwide, a large proportion are currently located in China (47), the UAE (30), Singapore (16), Malaysia (15), and Qatar (11).⁴⁶⁹ IBCs operating in some of these countries report attempts by local and national governments to interfere with the content of classes and curriculums and

⁴⁵⁸ Bard College also offers academic programs in Kyrgyzstan and Israel.

⁴⁵⁹ Kent, "Challenges in a Disrupted World," 2020.

⁴⁶⁰ Kent, "Challenges in a Disrupted World," 2020.

⁴⁶¹ Ross, "Is the Offshore-Campus Boom Over?," April 28, 2022; Toner, "Branching Out," September 1, 2019.

⁴⁶² Toner, "Branching Out," September 1, 2019.

⁴⁶³ Kent, "Challenges in a Disrupted World," 2020.

⁴⁶⁴ Kent, "Challenges in a Disrupted World," 2020.

⁴⁶⁵ Kinser and Lane, *C-BERT International Campus Listing*, March 23, 2023.

⁴⁶⁶ Kent, "Challenges in a Disrupted World," 2020.

⁴⁶⁷ Yale-NUS was a collaboration between Yale University and the National University of Singapore.

⁴⁶⁸ Kent, "The Implications of Campus Closures," 2022.

⁴⁶⁹ Kinser and Lane, *C-BERT International Campus Listing*, March 23, 2023.

otherwise create a climate hostile to the academic freedoms that are common in many countries in Europe and North America.⁴⁷⁰ The experience of New York University (NYU) at its branch campuses in China and the UAE are illustrative.

According to Freedom House's 2021 report, at least 10 NYU faculty members have been denied entry to the UAE for the purpose of teaching or conducting research at the NYU branch campus in Abu Dhabi. Some NYU students and administrative and support staff have also been reportedly denied entry to the UAE.⁴⁷¹ UAE officials have also reportedly added scholars and students who have criticized government policy to a Gulf Cooperation Council security list, an action that effectively denies access to the entire region.⁴⁷² In China, representatives from NYU Shanghai have reported a pattern of rejecting teaching applications for faculty who are likely to criticize Chinese government policies. For example, in 2015, NYU faculty member Kwame Anthony was denied a visa to speak on the NYU Shanghai campus.⁴⁷³

Outlook

Going forward, the outlook for international student enrollments is expected to level off or even decline. U.S. universities face growing competition for international students, with several factors threatening to erode the U.S. share of such students. First, growth has been rapid in the number of options for university-level study in both traditional destinations like Canada and the UK and less traditional locations like Japan, Qatar, or Czechia.⁴⁷⁴ Many such programs are taught in English, with the number of English-language degree programs available in non-English-speaking countries growing over the past 20 years from 500 programs to more than 17,000 programs by 2021.⁴⁷⁵ Like U.S. students, some international students are also questioning the high costs associated with obtaining a degree from a U.S. university as well as the return on investment, particularly if U.S. university credentials are of limited use in obtaining a job in their home country.⁴⁷⁶ Regarding IBCs, campuses that enroll sufficient numbers of students and attract external funding are likely to thrive, whereas those with low enrollments and mediocre academic reputations will struggle.⁴⁷⁷

Management Consulting Services: Rising Demand, Shifting Priorities

As described in chapter 2, providers of management consulting services advise companies, governments, and nonprofits on a range of business, technology, and organizational issues. In recent years, demand for management consulting services has been strong amid technological advancements and shifts in workplace habits in the wake of the COVID-19 pandemic. Despite that growth, certain disruptive trends have been underway in the sector for at least a decade, signaling that competitive conditions in the

⁴⁷⁰ Ross, "Is the Offshore Campus Boom Over?," April 28, 2022.

⁴⁷¹ Freedom House, *United Arab Emirates*, 2021; Ross, "Is the Offshore Campus Boom Over?," April 28, 2022.

⁴⁷² Freedom House, *United Arab Emirates*, 2021.

⁴⁷³ Ross, "Is the Offshore Campus Boom Over?," April 28, 2022.

⁴⁷⁴ Fischer and Aslanian, "Fading Beacon," August 2, 2021.

⁴⁷⁵ Fischer and Aslanian, "Fading Beacon," August 2, 2021.

⁴⁷⁶ Fischer and Aslanian, "Fading Beacon," August 2, 2021.

⁴⁷⁷ Wilkins, "International Branch Campuses," 2023.

Chapter 4: Demand for Professional Services: Changing Demographics, Income Growth, and Continued Digitalization of Businesses

global management consulting industry are experiencing profound change. Several sources indicate that firms have moved away from the traditional focus on strategy consulting and now offer expertise in a broader range of areas where demand is rising, including digital transformation, cybersecurity, supply chain optimization, and sustainability. ⁴⁷⁸ Clients are increasingly seeking consultants who can effectively execute plans and produce measurable outcomes.

As demand rises for different kinds of expertise, competition in the consulting sector is increasing. Accounting firms currently dominate the management consulting sector, and information services companies play a prominent role as well. The importance of smaller firms is also growing, offering clients competitively priced, personalized support on a more localized scale.⁴⁷⁹ New consulting firms have emerged with specialized expertise, particularly in the information technology (IT) field, often producing sophisticated results at lower cost than traditional consulting firms. Competition has been further heightened by the emergence of freelance consulting platforms. Moreover, many traditional clients are expanding internal consulting capabilities rather than hiring outside consultants. This "insourcing" is another trend impacting demand for consulting services.⁴⁸⁰

Strong Demand for Management Consulting Services Has Been Accompanied by Heightened Competition

A *Harvard Business Review* study from 2013 indicated that after maintaining the same fundamental business model for several decades, the management consulting sector was on the verge of disruption, as companies faced the emergence of new competitors.⁴⁸¹ In line with growing demand, competition in the management consulting sector has heightened since that study was published. Accounting firms such as Deloitte and PricewaterhouseCoopers (see accounting and auditing services in chapter 3), have joined more traditional management consulting companies such as McKinsey & Company as dominant players.⁴⁸² Moreover, IT firms are increasingly entering the consulting space and can provide technology solutions directly to clients, benefiting from staff that have deep subject matter expertise.⁴⁸³ Small, niche companies are also becoming more active in the management consulting sector.

As companies seek assistance with addressing a range of complex challenges, many have shifted away from reliance on a single provider. Instead, clients have begun to move toward a multisourcing approach, hiring multiple partners with a range of capabilities to complete different aspects of a project. As a

⁴⁷⁸ Christensen et al., "Consulting on the Cusp of Disruption," October 1, 2013; MarketLine, US Management Consultancy Report, March 2024, 10.

⁴⁷⁹ Partis, "Consulting Contracts on the Rise," January 30, 2025.

⁴⁸⁰ Oarue-Itseuwa, "Al's Impact on the Consultancy Sector," February 2, 2024, 51; Rieppi, "Consulting Industry Trends in 2023," May 25, 2023.

⁴⁸¹ Christensen et al., "Consulting on the Cusp of Disruption," October 1, 2013.

⁴⁸² MarketLine, US Management Consultancy Report, March 2024, 2.

⁴⁸³ Oarue-Itseuwa, "Al's Impact on the Consultancy Sector," February 2, 2024, 53.

result, large consulting firms may work on a particular project with various partners, including technology companies and freelancers.⁴⁸⁴

Freelance management consulting marketplaces such as Catalant, Upwork, and Clarity have sprouted up in recent years, easing the shift toward multisourcing.⁴⁸⁵ According to one source, the number of individual service professionals providing full-time or part-time services to businesses increased 50 percent from 2020 to 2024.⁴⁸⁶ Online consulting marketplaces allow clients to hire specialized consultants using a piecemeal approach, often with cost savings and faster turnaround time compared with large consulting firms.⁴⁸⁷

New technologies such as virtual communications platforms and GenAI have helped to level the playing field for smaller firms and independent consultants. Individual consultants are able to use GenAI tools to accomplish complex tasks, such as large dataset analysis, which previously was prohibitively resource intensive.⁴⁸⁸ In the wake of the COVID-19 pandemic, the decrease in demand for in-person solutions has made consultancy more accessible for small firms and for staff across geographic boundaries.⁴⁸⁹ Despite the benefits of meeting clients in person, location is now considered less important than getting "the right talent at the right value."⁴⁹⁰

In addition to rising competition within the sector, demand for management consulting services has also been affected by clients' decisions to shift some traditional consultant roles inside their companies. Many large multinational enterprises (including MasterCard and Pfizer) are "insourcing" consulting practices, staffed by former consultants who may be more intimately familiar with the challenges faced by a specific industry than their counterparts at large consulting firms.⁴⁹¹ Another trend that is contributing to decreased reliance on external consultants for certain tasks is the decision by many large companies to hire internal data scientists in an effort to make strategic decisions more data driven.⁴⁹²

These developments have had implications for employment in the management consulting industry. Robust growth in management consulting services contributed to significant increases in the number of consulting staff during the late 2010s, a trend that continued into the early pandemic years. However,

⁴⁸⁸ Younger, "CEOs Explain How AI Will Super Charge Independent Management Consulting," September 5, 2023; Kaplan, "The Business Consulting Industry Is Being Disrupted," August 12, 2024.

⁴⁸⁴ Nagarajan, "The Future of Consulting," November 23, 2022; Rieppi, "Consulting Industry Trends in 2023," May 25, 2023.

⁴⁸⁵ Vitaud, "Consulting Will Never Be the Same Again," June 15, 2016; Fuller et al., "Building the On-Demand Workforce," November 2020.

⁴⁸⁶ MBO Partners, "The Independent by Choice Movement" accessed January 16, 2025.

⁴⁸⁷ Younger, "Freelance Management Consulting," May 20, 2023; Kaplan, "The Business Consulting Industry Is Being Disrupted," August 12, 2024.

⁴⁸⁹ Laffitte, "2024 In Review," November 25, 2024.

⁴⁹⁰ Nagarajan, "The Future of Consulting," November 23, 2022.

⁴⁹¹ Oarue-Itseuwa, "Al's Impact on the Consultancy Sector," February 2, 2024, 51; Rieppi, "Consulting Industry Trends in 2023," May 25, 2023.

⁴⁹² Vitaud, "Consulting Will Never Be the Same Again," June 15, 2016.

Chapter 4: Demand for Professional Services: Changing Demographics, Income Growth, and Continued Digitalization of Businesses

amid rising competition and falling profit margins, high-profile consulting firms such as Accenture and McKinsey & Company have experienced significant layoffs since 2023.⁴⁹³

Shifting Demand Has Brought Changes in Business Models

To stay relevant and more effectively meet client demand, management consulting companies are being forced to change their business models and offer a broader range of services. Traditional features of the management consulting industry have been shifting, highlighted by movement away from onsite work and strategy consulting and toward customized solutions and results-oriented consulting. Some of the changes facing the management consulting sector were highlighted in the Commission's 2021 report, which focused on shifts in the modes of supply of consulting services resulting from COVID-19 pandemic-related travel restrictions and the rise of technology consulting services.⁴⁹⁴

Strategy consulting, which involves identifying and analyzing an enterprise's long-term goals and making recommendations for future action, has been a prominent feature of the management consulting industry since the founding of Boston Consulting Group (BCG) in 1963.⁴⁹⁵ Strategy remains significant, but its share of total market value has been decreasing as demand for other types of consulting grows. The 2013 *Harvard Business Review* study indicated that the share of work devoted to classic strategy consulting was down to about 20 percent of the total, from 60–70 percent 30 years earlier.⁴⁹⁶ According to MarketLine, strategy consulting in the United States accounted for 10.8 percent of the sector's total revenues in 2023, from 15.1 percent in 2018.⁴⁹⁷

In response to increased client demand for tailored and customized solutions, top companies have been building non-strategy practices, shifting away from tools that are replicable and transferable across companies and industries.⁴⁹⁸ This trend has provided opportunities for the development of industry-specific consulting services.⁴⁹⁹ Large management consulting companies now tend to organize themselves across both industries (such as financial services, retail trade, health care, transport and logistics) and practice areas (strategy, people, digital technology, environmental sustainability).

Another change for the consulting sector relates to the shift toward results-oriented consulting. In the past, strategies recommended by management consultants were sometimes difficult to implement and did not always bring the expected outcomes.⁵⁰⁰ Clients reportedly are placing more emphasis on

⁴⁹³ Fubini, "Are Management Consulting Firms Failing to Manage Themselves?," July 9, 2024; Richter, "The Consulting Industry at a Crossroads," January 31, 2025.

⁴⁹⁴ USITC, Recent Trends in U.S. Services Trade: 2021 Annual Report, April 2021, 69–71.

⁴⁹⁵ Vitaud, "Consulting Will Never Be the Same Again," June 15, 2016.

⁴⁹⁶ Christensen et al., "Consulting on the Cusp of Disruption," October 1, 2013.

⁴⁹⁷ MarketLine, *US Management Consultancy Report*, March 2024, 10. Other sources show different estimates, depending on how breakdowns are measured. See, for example, IBISWorld, *Management Consulting in the US*, August 2024, 12.

⁴⁹⁸ Rieppi, "Consulting Industry Trends in 2023," May 25, 2023.

⁴⁹⁹ Vitaud, "Consulting Will Never Be the Same Again," June 15, 2016; Oarue-Itseuwa, "Al's Impact on the Consultancy Sector," February 2, 2024, 51.

⁵⁰⁰ Rieppi, "Consulting Industry Trends in 2023," May 25, 2023.

execution and achieving measurable results.⁵⁰¹ Cocreation is becoming more prevalent, where consultants work with company staff to ensure that solutions meet a client's unique needs and can be altered as circumstances change.⁵⁰²

In line with the movement toward outcome-oriented consulting, pricing models are beginning to evolve. Previously, most consulting companies billed clients according to the number of hours devoted to a particular project, with high margins between billable rates and actual salaries.⁵⁰³ However, some companies are now linking part of their fees to the achievement of outcomes, especially as lower-priced competition emerges. "Value-based pricing," where clients share risks with their consultants, is expected to become a standard payment model in the future, impacted partly by the advent of GenAI, which reduces the amount of time required to perform certain tasks.⁵⁰⁴ Subscription-based services may also become more common.⁵⁰⁵

Digitalization and AI Drive Consulting Demand

Although management consulting firms have been advising on technology issues since at least the mid-2010s, the COVID-19 pandemic accelerated the digital transformation and the movement toward remote and hybrid work, triggering increased demand for consultants in certain practice areas. Clients have needed consultants to assist them with workforce transformation and navigating adoption of new technologies, such as video conferencing, project management software, and virtual collaboration platforms.⁵⁰⁶

In the past, businesses typically delegated technology and IT enablement to specialists. However, with the aim of optimizing outcomes and improving user experience, technology and innovation have increasingly become a core part of the consulting strategy, through the integration of automation, AI, data analytics, digital, and cloud computing services.⁵⁰⁷ According to a 2024 survey of global corporate executives commissioned by IBM, 86 percent of consulting buyers are seeking services that incorporate AI and technology.⁵⁰⁸ Management consulting firms traditionally emphasized human expertise, but this has started to change. In an effort to meet demand by increasing their technological capabilities, top

⁵⁰¹ Oarue-Itseuwa, "Al's Impact on the Consultancy Sector," February 2, 2024, 51; Fubini, "Are Management Consulting Firms Failing to Manage Themselves?," July 9, 2024.

⁵⁰² Kaplan, "The Business Consulting Industry Is Being Disrupted," August 12, 2024.

⁵⁰³ Kaplan, "The Business Consulting Industry Is Being Disrupted," August 12, 2024.

⁵⁰⁴ Consulting.us, "Generative AI's Opportunities and Challenges for Consulting Firms and Consultants," October 3, 2024; Oarue-Itseuwa, "AI's Impact on the Consultancy Sector," February 2, 2024, 54.

⁵⁰⁵ Oarue-Itseuwa, "Al's Impact on the Consultancy Sector," February 2, 2024, 54; Rieppi, "Consulting Industry Trends in 2023," May 25, 2023.

⁵⁰⁶ Laffitte, "2024 In Review," November 25, 2024.

⁵⁰⁷ Nagarajan, "The Future of Consulting," November 23, 2022; Rieppi, "Consulting Industry Trends in 2023," May 25, 2023.

⁵⁰⁸ Candy et al., "Consulting Reimagined, Powered by AI," October 2024, 6.

consulting companies are making significant investments in new technologies such as AI and acquiring or partnering with specialty firms, a trend that has been ongoing since 2015.⁵⁰⁹

Since the launch of ChatGPT in late 2022, consulting companies have also been helping clients deploy GenAI solutions, a challenge that includes navigating concerns about data security and ethics.⁵¹⁰ Rising digitalization (and the prevalence of data breaches and cyberattacks) makes cybersecurity another high priority in which consultants can assist their clients.⁵¹¹

Supply Chain Optimization and Sustainability Initiatives Further Support Growth

Another practice area that gained prominence during the pandemic is supply chain resilience. Lockdowns in China and other countries led to shortages of key materials needed for the manufacturing sector, highlighting vulnerabilities in global supply chains and driving demand for consulting on supply chain optimization.⁵¹² Given the increased emphasis placed on global sourcing in connection with environmental sustainability, consulting firms are also advising on sustainable supply chain practices, carbon footprint reduction, and circular economy initiatives.⁵¹³

Demand for consulting in the environmental sphere has also been impacted by the green energy transition, as both governments and companies seek assistance in implementing sustainability projects in areas such as electric mobility and renewable energy generation, storage, and distribution.⁵¹⁴ Helping clients work toward more environmentally sustainable practices, through mechanisms such as carbon accounting⁵¹⁵ and assistance with applications for environmental programs under the U.S. Inflation Reduction Act (IRA) and the European Union's Recovery and Resilience Facility (RRF), for example, has become more prominent in the consulting sphere.⁵¹⁶ Consulting firms have also been active in the Chinese market, assisting government agencies and companies with sustainability initiatives.⁵¹⁷ One source pointed out that sustainability consulting is no longer seen primarily through the lens of

⁵⁰⁹ Oarue-Itseuwa, "Al's Impact on the Consultancy Sector," February 2, 2024, 52; Samokhvalov, "The Transformative Impact of Artificial Intelligence on the Management Consultancy Sector," February 2, 2024, 64; Gitelman et al., "Energy Engineering and Consulting," 2020, 273.

⁵¹⁰ Bannister, "BCG Delivers 20th Consecutive Year of Growth," April 15, 2024; *Harvard Business Review* and Salesforce, "Transforming Consulting Through Generative AI," May 13, 2024, 6.

⁵¹¹ Kennedy Consulting, "Insights and Resources," accessed December 11, 2024.

⁵¹² USITC, COVID-19 Related Goods, December 2020; Sutter, "Rethinking Supply Chains after China's Lockdown," July 7, 2022; IBISWorld, *Global Management Consultants*, December 2024, 7.

⁵¹³ Bain & Company, "Climate Change & Decarbonization," accessed December 12, 2024.

⁵¹⁴ Jakaityte, "The Sustainability Surge," February 20, 2025; IBISWorld, *Global Management Consultants*, December 2024, 12.

⁵¹⁵ Carbon accounting, or greenhouse gas (GHG) accounting, is the process of measuring an entity's direct and indirect GHG emissions, helping to quantify its environmental impact.

⁵¹⁶ Barth et al., "US States and the Successful Clean Energy Transition," January 29, 2024; Deloitte, "Next Generation EU: Pandemic Recovery Plan," accessed February 10, 2025; Inflation Reduction Act of 2022, H.R.5376 (2021–2022); European Parliament and the Council, Regulation 2021/241 establishing the Recovery and Resilience Facility (March 1, 2023).

⁵¹⁷ Ensun, "Top 60 Sustainability Consulting Companies in China (2025)," accessed April 9, 2025.

compliance; rather, it has become "a strategic advantage," providing companies with avenues for innovation and growth.⁵¹⁸

Outlook

Despite expectations of increased client demand for management consulting services, several of the challenges facing the sector in recent years are likely to persist, including rising competition and integration of new technologies. Given the low barriers to entry, the number of small players will likely continue to increase, with the entry of niche firms (particularly in the digital technologies sector) as well as independent consultants endangering the leading position of key players.⁵¹⁹ Consulting firms' costs will likely shift further toward investments in technology (particularly AI) and away from human capital.⁵²⁰ Consultants will need to have strong skills in technology, alongside knowledge of how to incorporate it into business processes.⁵²¹ As competition rises and the integration of AI into consulting services increases, pricing models are expected to continue to move away from traditional practices (hourly billing) and toward solutions-oriented or "success-based" fees.⁵²²

Advertising Services: A Shift from Traditional to Digital Media

Advertising services publicize and promote products and services to persuade consumers to purchase them.⁵²³ Historically, advertisements have been consumed through print and broadcast media such as newspaper ads, television commercials, cinema ads, billboards, and radio, collectively referred to as "traditional advertising."⁵²⁴ Today, most advertising revenue comes from "digital advertising," which refers to the practice of promoting products or services through online ad placements with the aim of driving traffic, increasing brand awareness, and generating leads or sales using data-driven strategies.⁵²⁵ Types of digital advertising channels include digital video, search engines, social media platforms, digital banner, digital audio, digital classifieds, and online influencer advertisements.⁵²⁶

⁵¹⁸ Laffitte, "2024 In Review," November 25, 2024.

⁵¹⁹ Kaplan, "The Business Consulting Industry Is Being Disrupted," August 12, 2024; Nagarajan, "The Future of Consulting," November 23, 2022; IBISWorld, *Management Consulting in the US*, August 2024, 27.

⁵²⁰ Oarue-Itseuwa, "AI's Impact on the Consultancy Sector," February 2, 2024, 53–54.

⁵²¹ IBISWorld, *Management Consulting in the US*, August 2024, 71; Kaplan, "The Business Consulting Industry Is Being Disrupted," August 12, 2024.

⁵²² Nagarajan, "The Future of Consulting," November 23, 2022; Oarue-Itseuwa, "AI's Impact on the Consultancy Sector," February 2, 2024, 54.

⁵²³ American Marketing Association, "Marketing vs. Advertising," accessed January 15, 2025; Statista, Advertising: Market Data & Analysis, November 2024, 7.

⁵²⁴ American Marketing Association, "Marketing vs. Advertising," accessed January 15, 2025; Statista, *Advertising: Market Data & Analysis*, November 2024, 7.

⁵²⁵ American Marketing Association, "Marketing vs. Advertising," accessed January 15, 2025; Statista, *Advertising: Market Data & Analysis*, November 2024, 7.

⁵²⁶ PwC, "Perspectives from the Global Entertainment & Media Outlook," July 16, 2024; American Marketing Association, "Marketing vs. Advertising," accessed January 15, 2025; Statista, *Advertising: Market Data & Analysis*, November 2024, 7.
Demand for advertising services as a whole has increased considerably during the past decade, both domestically and globally.⁵²⁷ Industry analysts have linked much of the growth to an increasingly digitalized economy, allowing firms to reach consumers through streaming services, social media platforms, and online search engines.⁵²⁸ With the shift away from traditional channels to digital advertising, firms that specialize in advertising services have increasingly used artificial intelligence (AI) tools to target audiences with tailored content, an emerging trend that has also spurred revenue growth in advertising services.⁵²⁹

The Decline of Linear Television and Rise of Digital Television and Video Streaming Services

Linear television, sometimes called traditional or analog television, refers to a service where scheduled television content is broadcast on cable or satellite networks and has been a source of traditional advertising. Linear television has experienced a decline in viewership and subscribers, while digital television and video streaming services have increased. This shift has been a key driver of the decline of traditional media advertising spending.⁵³⁰ In 2024, about 78 percent of total U.S. advertising expenditures and 73 percent of total global advertising expenditures were for digital advertising, with traditional advertising accounting for the remainder.⁵³¹ Traditional advertising represented a larger share of total advertising expenditures both in the United States and globally until 2019.⁵³²

The shift from traditional to digital advertising reflects the increase in the average amount of time U.S. adults spend per day engaging with digital media instead of traditional media. Digital ads are shown in most types of internet activities, including searching, mobile gaming, streaming, online shopping, and reading digital publications.⁵³³ One notable reason for the decline in time spent with traditional media is that internet access has increased significantly over the past 15 years. According to one study, 74 percent

⁵²⁷ For instance, U.S. exports of advertising services increased from \$8.6 billion in 2013 to \$23.8 billion in 2023, but U.S. imports increased from \$3.6 billion to \$5.7 billion over the same period. USDOC, BEA, table 2.1, "U.S. Trade in Services, by Type of Service," July 3, 2024; Statista, *Advertising: Market Data & Analysis*, November 2024; PwC, "Perspectives from the Global Entertainment & Media Outlook," July 16, 2024; Cramer-Flood, "US Time Spent With Media Forecast 2023," July 11, 2023.

⁵²⁸ PwC, "Perspectives from the Global Entertainment & Media Outlook," July 16, 2024; Harlow, "How Effective Are Ads on Social Media?," June 5, 2023; Aho Williamson, "Social Media Isn't Dying—It's Still Growing," July 11, 2023; Statista, *Advertising: Market Data & Analysis*, November 2024.

⁵²⁹ PwC, "Perspectives from the Global Entertainment & Media Outlook," July 16, 2024; Statista, *Advertising: Market Data & Analysis*, November 2024; Ezzat, "Enhance the Advertising Effectiveness by Using Artificial Intelligence (AI)," January 3, 2024.

⁵³⁰ Linear television is defined as scheduled television content that is broadcast on cable or satellite networks. Feger, "Linear TV: What It Is," September 6, 2024.

⁵³¹ Feger, "Digital Makes up over Three-Quarters of Total Ad Spend in the US," August 27, 2024; Statista, *Advertising: Market Data & Analysis*, November 2024, 17.

⁵³² Statista, *Advertising: Market Data & Analysis*, November 2024, 17; Shaban, "Digital Advertising to Surpass Print and TV," February 20, 2019; Statista Market Insights, "Digital Ad Spending," October 2024.

⁵³³ Cramer-Flood, "US Time Spent With Media Forecast 2023," July 11, 2023.

of U.S. households had access to broadband internet in their home in 2010, with such access rising to 82 percent in 2019 and 90 percent in 2023.⁵³⁴

In 2023, for the first time, linear television makes up less than half of total television usage.⁵³⁵ The top cable and satellite television providers in the United States, which represent most of the cable and satellite television market share, had a net loss of more than 21 million subscribers from 2018 to 2023, from 76 million in 2018 to 55 million in 2023.⁵³⁶ At the end of 2022, less than 50 percent of households had a cable or satellite television service.⁵³⁷

Because cable and satellite television's total subscriber base has diminished, linear television advertising has declined sharply as a share of total television advertising, which also includes digital video advertising.⁵³⁸ One industry study found that linear television advertising expenditures as a percentage of total video advertising fell from 71 percent in 2020 to only 52 percent in 2023 and is projected to decline to 48 percent in 2024, dropping below digital video advertising for the first time. The same study projected that digital video advertising would increase 80 percent faster than total media overall in 2024 and forecasted that this trend will continue.⁵³⁹

Some of the largest television networks, such as Paramount Global, Warner Bros. Discovery, ABC, and NBC Universal, all experienced significant declines in advertising revenue from their cable channels in 2023.⁵⁴⁰ Of this group, Paramount experienced the largest drop in revenues from Q3 2022 to Q3 2023, with a 14 percent decline in linear television advertising revenues, followed by Warner Bros. Discovery at a decline of about 13 percent, ABC at a roughly 9 percent decline, and NBC Universal at a roughly 8 percent decline.⁵⁴¹

Television networks are shifting toward delivering their content as over-the-top video streaming services to adjust to consumers' viewing preference leaning toward video streaming rather than traditional cable

⁵³⁴ LRG, "Actionable Research on the Broadband, Media & Entertainment Industries," 2023.

⁵³⁵ Nielsen Company, "Streaming Grabs a Record 38.7% of Total TV Usage," August 2023; Davis and Cranz, "It's Official, People Aren't Watching TV," August 15, 2023; Lebow, "New Data," August 16, 2023.

⁵³⁶ For example, the top cable and satellite TV providers include Charter, Comcast, DIRECTV, DISH TV, and others. LRG, "Major Pay-TV Providers Lost About 5,000,000 Subscribers In 2023," March 8, 2024; LRG, "Major Pay-TV Providers Lost About 2,875,000 Subscribers In 2018," March 6, 2019.

⁵³⁷ Gruenwedel, "Leichtman: 2023 U.S. Household SVOD Subscription Growth Flat," August 14, 2023; Winslow, "Major Pay TV Providers Lost Record 5M Subs in 2023," March 8, 2024; LRG, "Major Pay-TV Providers Lost About 5,000,000 Subscribers In 2023," March 8, 2024.

⁵³⁸ Digital video advertising includes advertisements placed in over-the-top video streaming services (e.g., Paramount Plus, Disney Plus, Hulu, Hulu Live TV, YouTube TV, and others), social media video (e.g., TikTok, Instagram, Facebook, Snap, Twitter, YouTube, and others), and online video (i.e., video from web-/app-based publishers). IAB, *2024 Digital Video Ad Spend & Strategy Report*, April 2024.

⁵³⁹ IAB, 2024 Digital Video Ad Spend & Strategy Report, April 2024, 7.

⁵⁴⁰ Steinberg and Earl, "Advertisers Ride the Brakes," January 18, 2024.

⁵⁴¹ Jacobson, "Ad Revenue Decrease Impacts Disney's Linear Network," November 9, 2023; Walt Disney Company, "The Walt Disney Company Reports Fourth Quarter and Full Year Earnings for Fiscal 2023," November 8, 2023; Steinberg and Earl, "Advertisers Ride the Brakes," January 18, 2024.

and satellite television.⁵⁴² One study estimates that in 2015, 52 percent of households each had a subscription to at least one video streaming service, rising to 83 percent of all households in 2022.⁵⁴³ Advertising via free advertising-supported television (FAST) streaming has also grown significantly, with FAST advertising expenditures nearly quadrupling from 2020 to 2023 and expected to double from 2023 to 2025.⁵⁴⁴ eMarketer projects that a third of the U.S. population will be FAST viewers by 2027.⁵⁴⁵

Social Media Platforms and the Emergence of Social Commerce

As people spend more of their time engaging with social media platforms, such as Facebook, Instagram, TikTok, and X (formerly Twitter), these platforms comprise part of social commerce, which has become important venues for online advertising.⁵⁴⁶ For instance, one 2023 study found that an estimated 28 percent of consumers discovered new brands and products through social media advertising.⁵⁴⁷ This report also found that more people learn about brands and new products through social media than through ads displayed on websites or before online videos.⁵⁴⁸ Industry surveys also suggest that consumers use social media almost as much as online search engines when they research a product.⁵⁴⁹

Social media's subscriber numbers continue to grow at a rapid pace and will soon outnumber linear television viewers in the United States.⁵⁵⁰ eMarketer estimated that in 2025, the United States would have 235 million social media users and 230 million television viewers, the first time that U.S. social media users would exceed television viewers.⁵⁵¹

⁵⁴² Over-the-top video streaming services includes subscription video-on-demand (SVOD) services, connected TV, virtual multichannel video programming distributors (MVPDs), and free-ad supported streaming television (FAST). SVOD is a subscription service where subscribers are charged a regular fee to access video content. SVOD includes services such as Netflix, Amazon Prime Video, Disney+, Max, and Hulu. Connected TV refers to the delivery of video content through smart TVs that have internal internet capabilities and external devices (e.g., Amazon Fire sticks, Google Chromecast streaming device, Roku streaming device, and gaming consoles). MVPDs are video services that deliver live and on-demand content over the internet through multiple channels (e.g., YouTube TV, Hulu+ Live TV, and Sling TV). Wohr, "OTT Advertising: What It Is and Why Streaming Matters to Marketers," July 18, 2024; Vimond, "What Are SVOD, AVOD, and TVOD?" accessed January 17, 2025; Lebow, "Guide to Connected TV," September 5, 2024.

⁵⁴³ LRG, "Actionable Research on the Broadband, Media & Entertainment Industries," 2023; Gruenwedel, "Leichtman: 2023 U.S. Household SVOD Subscription Growth Flat," August 14, 2023.

 ⁵⁴⁴ FAST services (also referred to as ad-supported video-on-demand) are video services that deliver free ad-supported streaming of video content (e.g., Pluto TV and Amazon Freevee). Wohr, "OTT Advertising: What It Is and Why Streaming Matters to Marketers," July 18, 2024; Fletcher, "FAST Ad Spend to Surpass That of Cable," January 12, 2023; Majidi, "Streaming Services Advertising in the United States—Statistics & Facts," September 11, 2024.
 ⁵⁴⁵ Yuen, "What Media Planners Need to Know," February 26, 2024.

⁵⁴⁶ Social media comprises applications and websites that create a digital network for people to interact with other users. McKinsey & Company, "What Is Social Media?," June 8, 2023.

⁵⁴⁷ Harlow, "How Effective Are Ads on Social Media?," June 5, 2023.

⁵⁴⁸ Harlow, "How Effective Are Ads on Social Media?," June 5, 2023.

⁵⁴⁹ Harlow, "How Effective Are Ads on Social Media?," June 5, 2023.

⁵⁵⁰ Aho Williamson, "Social Media Isn't Dying—It's Still Growing," July 11, 2023.

⁵⁵¹ Aho Williamson, "Social Media Isn't Dying—It's Still Growing," July 11, 2023.

As a result of this shift, social media platforms have captured a growing share of digital advertising spending in recent years.⁵⁵² According to one industry report, in 2024 social media was forecasted to become the second-highest spending media format with \$86.75 billion in expenditures, trailing only traditional search advertising by \$4 billion.⁵⁵³ The same report also projects that social media will have more advertising spending than internet search engines by 2026.⁵⁵⁴ In a report using data from most of 2024, Facebook held the largest share of total social media advertising spending with around 47.0 percent of the share, followed by Instagram (36.0 percent), TikTok (12.2 percent), Snapchat (4.0 percent), and X (1.0 percent) (figure 4.1).⁵⁵⁵



Source: eMarketer, "US Social Media Ad Spending Share, by Platform," August 22, 2024.

Social commerce has become a subcategory of e-commerce, and U.S. social media firms recognized that they had created a space that can gain the attention of users for the purpose of advertising products and services and have created various ways for consumers to shop and search for products directly in their applications (apps), which is also referred to as social commerce.⁵⁵⁶ When searching for products or

⁵⁵² Social media platforms include Facebook, X, Instagram, TikTok, Snapchat, and others. McKinsey & Company, "What Is Social Media?," June 8, 2023; Harlow, "How Effective Are Ads on Social Media?," June 5, 2023.

⁵⁵³ Traditional search includes advertising that appears on internet-connected devices and excludes retail media search (searching through a retail media website or app that primarily are engaged in retail ecommerce). Yuen, "Traditional Search . . . US Ad Dollars in 2024," June 14, 2024.

⁵⁵⁴ Yuen, "Traditional Search . . . US Ad Dollars in 2024," June 14, 2024.

⁵⁵⁵ EMarketer, "US Social Media Ad Spending Share, by Platform," August 22, 2024.

⁵⁵⁶ Social commerce is the use of social media platforms to promote and sell products or services. Stanley, "What Is Social Commerce?," April 29, 2024; Shah and Borkakoty, "Social Media Trends and Predictions for 2025," December 19, 2024.

services, about 73 percent of consumers favor short-form video content, such as that popularized by TikTok, Instagram Reels, and YouTube Shorts.⁵⁵⁷

Furthermore, social commerce platforms aim to keep consumers within their social media apps' ecosystems. By doing so, the apps gather data from the consumers—such as the brands that they follow and products that they have "liked" or purchased.⁵⁵⁸ These data enable social media platform algorithms to produce more targeted and personalized ads. The placement of such ads may vary: desktop feeds, mobile feeds, social media posts, video feeds, in profiles of the user or other users of the platform, in an email in-box, among search results, or other platform locations in consumers' media feeds. Such options provide multiple incentives for companies to advertise on these platforms.⁵⁵⁹

TikTok, for example, recently launched a new TikTok Shop feature that allows companies to advertise and sell their products directly on TikTok. Facebook Marketplace allows consumers and businesses to advertise and sell their products to local customers.⁵⁶⁰ A significant portion of social media users in the United States make purchases through social media platforms.⁵⁶¹ eMarketer estimated that in 2024, roughly 37 percent of U.S. users of Facebook and Instagram made at least one purchase through the social media platform.⁵⁶² TikTok led in this category with 44 percent of U.S. users making at least one purchase through the social media platform in 2024.⁵⁶³ In 2022, McKinsey & Company estimated that by 2025, U.S. social commerce retail earnings would reach \$80 billion, which would make up 5 percent of total U.S. e-commerce.⁵⁶⁴

Despite the size and growth of social commerce in the United States, sales revenue remains well below that of China. China's social commerce industry has been heavily developed through platforms such as WeChat and Xiaohongshu and is estimated to have reached nearly \$508 billion in revenue in 2024.⁵⁶⁵ The increase in the amount of time U.S. consumers spend on social media and the increase in making purchases online, however, point to growth in U.S. social commerce, boosted especially by younger consumers. For example, according to a 2021 Forrester survey, 61 percent of U.S. adults younger than 25 years old who were online had previously completed a purchase on social media apps.⁵⁶⁶

⁵⁵⁷ Shah and Borkakoty, "Social Media Trends and Predictions for 2025," December 19, 2024; Kumar, "29 Video Marketing Statistics 2025," March 6, 2025.

⁵⁵⁸ McKinsey & Company, "Social Commerce," October 19, 2022.

⁵⁵⁹ McKinsey & Company, "Social Commerce," October 19, 2022.

⁵⁶⁰ Shah and Borkakoty, "Social Media Trends and Predictions for 2025," December 19, 2024; Stanley, "What Is Social Commerce?," April 29, 2024; McKinsey & Company, "Social Commerce," October 19, 2022.

⁵⁶¹ Enberg, "US Social Commerce Forecast 2024," December 2, 2024.

⁵⁶² Enberg, "US Social Commerce Forecast 2024," December 2, 2024.

⁵⁶³ Enberg, "US Social Commerce Forecast 2024," December 2, 2024.

⁵⁶⁴ McKinsey & Company, "Social Commerce," October 19, 2022.

⁵⁶⁵ Verot, "Social Commerce in China: Sell on Chinese Social Media," December 11, 2024; McKinsey & Company, "Social Commerce," October 19, 2022.

⁵⁶⁶ McKinsey & Company, "Social Commerce," October 19, 2022.

Adoption of Artificial Intelligence (AI) in Advertising

Advertisers' demand for more cost-effective ad formats prompted digital advertising firms to increasingly use AI technologies to target advertising more effectively, create content more quickly, and manage and analyze performance levels of individual advertisements.⁵⁶⁷ AI programs allow firms to analyze consumer behavior data and identify patterns, using algorithms to more accurately forecast future consumer preferences.⁵⁶⁸ Targeted content via AI enables advertisers to boost engagement of their target audiences and drive up sales of their products and services, which in turn potentially increases the efficiency and return on investment in advertising.⁵⁶⁹ For example, companies use AI to enhance and optimize their email marketing tasks, such as writing and sending emails to customers with personalized and tailored content based on the customer-specific data that have been obtained from previous interactions.⁵⁷⁰ In addition, AI has become a tool for firms to use in programmatic advertising.⁵⁷¹ By analyzing consumers' data on the kind of online content they consume, firms use the data to make decisions or suggestions when buying or selling advertisement spaces.⁵⁷²

Through the use of AI, content creators can reduce the time it takes to produce social media posts and emails.⁵⁷³ AI programs such as Adobe Firefly, Shopify Magic, and Jasper AI allow content creators to quickly generate personalized content and pictures for advertisements to specific customers.⁵⁷⁴ In addition, AI-generated images can be a more cost-effective solution for social media creators compared to licensed photos.⁵⁷⁵ Content creators also may benefit from using AI to analyze keywords that can help achieve higher rankings on search engine results pages.⁵⁷⁶

⁵⁶⁷ Ezzat, "Enhance the Advertising Effectiveness by Using Artificial Intelligence (AI)," January 3, 2024; Marr, "Artificial Intelligence and the Future of Marketing," September 9, 2022; Flinders and Hayes, "A Guide to AI in Marketing," March 10, 2025; Baldassarre, "How AI Is Revolutionizing Digital Advertising In 2024," August 9, 2024; Shopify, "Top 5 AI Content Creation Tools," September 30, 2024.

⁵⁶⁸ Ezzat, "Enhance the Advertising Effectiveness by Using Artificial Intelligence (AI)," January 3, 2024; Baldassarre, "How AI Is Revolutionizing Digital Advertising In 2024," August 9, 2024.

⁵⁶⁹ Ezzat, "Enhance the Advertising Effectiveness by Using Artificial Intelligence (AI)," January 3, 2024; Flinders and Hayes, "A Guide to AI in Marketing," March 10, 2025.

⁵⁷⁰ Marr, "Artificial Intelligence and the Future of Marketing," September 9, 2022; Ezzat, "Enhance the Advertising Effectiveness by Using Artificial Intelligence (AI)," January 3, 2024.

⁵⁷¹ Flinders and Hayes define programmatic advertising as "the automation of the purchasing and placement of ads on websites and applications." Flinders and Hayes, "A Guide to AI in Marketing," March 10, 2025.

⁵⁷² Ezzat, "Enhance the Advertising Effectiveness by Using Artificial Intelligence (AI)," January 3, 2024; Baldassarre, "How AI Is Revolutionizing Digital Advertising In 2024," August 9, 2024.

⁵⁷³ Davenport and Mittal, "How Generative AI Is Changing Creative Work," November 14, 2022; Flinders and Hayes, "A Guide to AI in Marketing," March 10, 2025; Shopify, "Top 5 AI Content Creation Tools," September 30, 2024.

⁵⁷⁴ Flinders and Hayes, "A Guide to AI in Marketing," March 10, 2025; Shopify, "Top 5 AI Content Creation Tools," September 30, 2024; Davenport and Mittal, "How Generative AI Is Changing Creative Work," November 14, 2022; Nagl and King, "How AI Is Impacting Digital Marketing," August 27, 2024.

⁵⁷⁵ Flinders and Hayes, "A Guide to AI in Marketing," March 10, 2025; Shopify, "Top 5 AI Content Creation Tools," September 30, 2024.

⁵⁷⁶ Shopify, "Top 5 AI Content Creation Tools," September 30, 2024.

Data management is another way firms use AI to enhance their advertising capabilities. They use AI in managing multiple ad campaigns by streamlining the data review process. Managing multiple ad campaigns typically requires time and resources to review ad performance metrics to determine and identify underperforming ads. A 2024 report suggests that advertisers can potentially use AI to generate a reporting tool to identify when an ad has collected a sufficient amount of data, analyze it, and recommend whether the firm should continue or stop an ad campaign.⁵⁷⁷

Despite the potential benefits of AI in advertising services, concerns persist regarding the implementation of AI tools in the advertising industry. One notable concern is the protection of customer data and adherence to data protection regulations.⁵⁷⁸ Because AI tools used in advertising often incorporate personal data from customers, industry analysts have voiced concerns that customer data may be misused or compromised.⁵⁷⁹ An additional concern for advertising firms is whether their use of AI will increase the difficulty of demonstrating compliance with data governance regulations.⁵⁸⁰ As customer data are acquired and AI tools use these data in new ways over time, companies will need to ensure that they comply with existing AI-related regulations and that their terms and conditions statements and customers' data privacy rights are not violated.⁵⁸¹ These issues may prompt advertisers to audit their AI systems and develop data governance tools to ensure that customer data are not misused.⁵⁸²

Outlook

Advertising services are expected to grow further in the future through the effects of an increasingly digitalized economy that will allow firms to reach consumers through digital media such as streaming services, social media platforms, and online search engines.⁵⁸³ More people have gained access to the internet in their homes in the United States, providing more avenues for advertisers to reach consumers.

⁵⁷⁷ Baldassarre, "How AI Is Revolutionizing Digital Advertising In 2024," August 9, 2024.

⁵⁷⁸ Forbes, "11 Risks To Using AI In Marketing," April 25, 2024; ProcureCon Marketing, "The Legal and Privacy Implications of AI in Marketing," November 2025; Baldassarre, "How AI Is Revolutionizing Digital Advertising In 2024," August 9, 2024.

⁵⁷⁹ Clark, "The Ethical Dilemma of AI In Marketing," March 14, 2024; ProcureCon Marketing, "The Legal and Privacy Implications of AI in Marketing," November 2025.

⁵⁸⁰ Clark, "The Ethical Dilemma of AI In Marketing," March 14, 2024; ProcureCon Marketing, "The Legal and Privacy Implications of AI in Marketing," November 2025.

⁵⁸¹ Clark, "The Ethical Dilemma of AI In Marketing," March 14, 2024; *Forbes*, "11 Risks To Using AI In Marketing," April 25, 2024; ProcureCon Marketing, "The Legal and Privacy Implications of AI in Marketing," November 2025; Baldassarre, "How AI Is Revolutionizing Digital Advertising In 2024," August 9, 2024.

⁵⁸² Clark, "The Ethical Dilemma of AI In Marketing," March 14, 2024; ProcureCon Marketing, "The Legal and Privacy Implications of AI in Marketing," November 2025.

⁵⁸³ PwC, "Perspectives from the Global Entertainment & Media Outlook," July 16, 2024; Harlow, "How Effective Are Ads on Social Media?," June 5, 2023; Aho Williamson, "Social Media Isn't Dying—It's Still Growing," July 11, 2023; Statista, *Advertising: Market Data & Analysis*, November 2024.

Projections indicate that consumers in the United States will continue to spend more of their time consuming digital media, increasing their demand for advertising services.⁵⁸⁴

Advertisers have increasingly used social media to reach consumers.⁵⁸⁵ In the United States, the number of social media users is expected to continue to grow and spur further increases to advertisement spending, taking advantage of AI to advertise and target audiences.⁵⁸⁶ The adoption of AI is also expected to grow and become a greater tool in advertising services by analyzing customer behavior and more effectively targeting consumers.⁵⁸⁷

⁵⁸⁴ LRG, "Actionable Research on the Broadband, Media & Entertainment Industries," 2023; Cramer-Flood, "US Time Spent with Media Forecast 2023," July 11, 2023; Statista, *Advertising: Market Data & Analysis*, November 2024; PwC, "Perspectives from the Global Entertainment & Media Outlook," July 16, 2024; Cramer-Flood, "US Time Spent With Media Forecast 2023," July 11, 2023.

⁵⁸⁵ Yuen, "Traditional Search . . . US Ad Dollars in 2024," June 14, 2024; Harlow, "How Effective Are Ads on Social Media?," June 5, 2023.

⁵⁸⁶ Davenport and Mittal, "How Generative AI Is Changing Creative Work," November 14, 2022; Flinders and Hayes, "A Guide to AI in Marketing," March 10, 2025; Shopify, "Top 5 AI Content Creation Tools," September 30, 2024; Aho Williamson, "Social Media Isn't Dying—It's Still Growing," July 11, 2023; Yuen, "Traditional Search . . . US Ad Dollars in 2024," June 14, 2024; Harlow, "How Effective Are Ads on Social Media?," June 5, 2023.

⁵⁸⁷ PwC, "Perspectives from the Global Entertainment & Media Outlook," July 16, 2024; Ezzat, "Enhance the Advertising Effectiveness by Using Artificial Intelligence (AI)," January 3, 2024; Marr, "Artificial Intelligence and the Future of Marketing," September 9, 2022; Flinders and Hayes, "A Guide to AI in Marketing," March 10, 2025; Baldassarre, "How AI Is Revolutionizing Digital Advertising In 2024," August 9, 2024; Shopify, "Top 5 AI Content Creation Tools," September 30, 2024.

Bibliography

- Aho Williamson, Debra. "Social Media Isn't Dying—It's Still Growing." EMarketer, July 11, 2023. https://content-naf.emarketer.com/social-media-isn-t-dying-it-s-still-growing.
- American Marketing Association. "Marketing vs. Advertising." Accessed January 15, 2025. https://www.ama.org/marketing-vs-advertising/.
- Bain & Company. "Climate Change & Decarbonization." Accessed December 12, 2024. <u>https://www.bain.com/consulting-services/sustainability-corporate-responsibility/climate-change-decarbonization/.</u>
- Baldassarre, Rocco. "How AI Is Revolutionizing Digital Advertising In 2024." *Forbes*, August 9, 2024. <u>https://www.forbes.com/councils/forbesagencycouncil/2024/04/09/how-ai-is-revolutionizing-digital-advertising-in-2024/</u>.
- Bannister, Brian. "BCG Delivers 20th Consecutive Year of Growth." Boston Consulting Group, April 15, 2024. <u>https://www.bcg.com/press/15april2024-bcg-delivers-20th-consecutive-year-growth</u>.
- Barth, Adam, Karina Gerstenschlager, Ksenia Kaladiouk, and Adi Kumar. "How US States Can Advance a Successful Clean-Energy Transition." McKinsey & Company, January 29, 2024. <u>https://www.mckinsey.com/industries/public-sector/our-insights/how-us-states-can-advance-a-successful-clean-energy-transition</u>.
- BestColleges. "Will College Campuses Return to Normal in Fall?" September 22, 2022. https://www.bestcolleges.com/blog/college-campuses-covid-19-guidelines-fall/.
- Bergson-Shilcock, Amanda, and Roderick Taylor. *Closing the Digital Skill Divide: The Payoff for Workers, Business, and the Economy*. National Skills Coalition and the Federal Reserve Bank of Atlanta, February 2023. <u>https://nationalskillscoalition.org/wp-</u> <u>content/uploads/2023/02/NSC DigitalDivide execsummary Feb2023.pdf</u>.
- Bohne, Raphael. "Revenue Growth of Professional Services Organizations Worldwide 2013–2022." Statista, July 12, 2023. <u>https://www.statista.com/statistics/933674/professional-services-worldwide-annual-revenue-growth/</u>.
- Burke, Lilah. "School's Out." American Prospect, July 29, 2024. <u>https://prospect.org/education/2024-07-</u>29-schools-out-demographics-costs-higher-ed/.
- Candy, Matthew, Blaine Dolph, Salima Lin, and Jacob Dencik. "Consulting Reimagined, Powered by AI." IBM, October 2024. <u>https://www.ibm.com/thought-leadership/institute-business-value/en-us/report/consulting-ai</u>.
- Caplan, Zoe. "U.S. Older Population Grew From 2010 to 2020 at Fastest Rate Since 1880 to 1890." U.S. Census Bureau. America Counts: Stories Behind the Numbers. May 25, 2023. <u>https://www.census.gov/library/stories/2023/05/2020-census-united-states-older-population-grew.html</u>.

- Chada, Sasha. "What Are International Campuses?" *Ivy Scholars*, June 11, 2024. https://www.ivyscholars.com/what-are-international-campuses/.
- Christensen, Clayton M., Dina Wang, and Derek van Bever. "Consulting on the Cusp of Disruption." Harvard Business Review, October 1, 2013. <u>https://hbr.org/2013/10/consulting-on-the-cusp-of-disruption</u>.
- Clark, Elijah. "The Ethical Dilemma of Al In Marketing: A Slippery Slope." *Forbes*, March 14, 2024. <u>https://www.forbes.com/sites/elijahclark/2024/03/14/the-ethical-dilemma-of-ai-in-marketing-a-slippery-slope/</u>.
- Consulting.us. "Generative AI's Opportunities and Challenges for Consulting Firms and Consultants." October 3, 2024. <u>https://www.consulting.us/news/11243/generative-ais-opportunities-and-challenges-for-consulting-firms-and-consultants</u>.
- Costa, Sergio A., Kavouras, Ilias, Cohen, Nevin, and Huang, Terry. "Moving Education Online During the COVID-19 Pandemic: Thinking Back and Looking Ahead." *Frontiers in Public Health*, vol. 9 (October 25, 2021): 1–3. <u>https://pmc.ncbi.nlm.nih.gov/articles/PMC8572807/</u>.
- Coulson, Justine. "Ensuring All Older People Can Benefit from the Development of the Silver Economy." United Nations Populations Fund China, February 2, 2024. <u>https://china.unfpa.org/en/news/opinion-editorial-ensuring-all-older-people-can-benefit-development-silver-economy</u>.
- Cramer-Flood, Ethan. "US Time Spent with Media Forecast 2023." EMarketer, July 11, 2023. https://content-naf.emarketer.com/us-time-spent-with-media-forecast-2023.
- Davenport, Thomas, and Nitin Mittal. "How Generative AI Is Changing Creative Work." *Harvard Business Review*, November 14, 2022. <u>https://hbr.org/2022/11/how-generative-ai-is-changing-creative-work</u>.
- Davis, Wes, and Alex Cranz. "It's Official, People Aren't Watching TV as Much as They Used to." *The Verge*, August 15, 2023. <u>https://www.theverge.com/2023/8/15/23833516/nielsen-tv-cable-50-percent-decline-viewership-bum-bums</u>.
- Deloitte. "Next Generation EU: Pandemic Recovery Plan to Build a Greener, More Innovative, Stronger Europe." Accessed February 10, 2025. <u>https://www.deloitte.com/lk/en/issues/work/next-generation-eu.html</u>.
- Eisend, Martin. "Older People in Advertising." *Journal of Advertising* 51, no. 3 (July 15, 2022): 308–22. https://doi.org/10.1080/00913367.2022.2027300.
- EMarketer. "US Social Media Ad Spending Share, by Platform, 2022-2024." August 22, 2024. <u>https://chart-na1.emarketer.com/268475/us-social-media-ad-spending-share-by-platform-2022-2024-of-total</u>.
- Enberg, Jasmine. "US Social Commerce Forecast 2024." EMarketer, December 2, 2024. <u>https://content-naf.emarketer.com/us-social-commerce-forecast-2024</u>.

- Ensun. "Top 60 Sustainability Consulting Companies in China (2025)." Accessed April 9, 2025. https://ensun.io/search/sustainability-consulting/china.
- European Parliament and the Council, Regulation 2021/241 Establishing the Recovery and Resilience Facility (March 1, 2023). <u>http://data.europa.eu/eli/reg/2021/241/2023-03-01/eng</u>.
- Ezzat, Rania. "Enhance the Advertising Effectiveness by Using Artificial Intelligence (AI)." *Journal of Art, Design & Music* 3, no. 1 (January 3, 2024): 1–13. <u>https://www.jadm.eg.net/journal/vol3/iss1/1/</u>.
- Feger, Arielle. "Digital Makes up over Three-Quarters of Total Ad Spend in the US." EMarketer, August 27, 2024. <u>https://content-naf.emarketer.com/digital-makes-up-over-three-quarters-total-ad-spend-us</u>.
- Feger, Arielle. "Linear TV: What It Is and What Marketers Need to Know about It." EMarketer, September 6, 2024. <u>https://www.emarketer.com/insights/definition-linear-tv-advertising/</u>.
- Fischer, Karin. "How International Education's Golden Age Lost Its Sheen." *The Chronicle of Higher Education*, March 28, 2019. <u>https://www.chronicle.com/article/how-international-educations-golden-age-lost-its-sheen/</u>.
- Fischer, Karin. "Latitudes: International Enrollments Are Down. Trump's Not the Reason." The Chronicle of Higher Education, April 9, 2025. <u>https://www.chronicle.com/newsletter/latitudes/2025-04-09</u>.
- Fischer, Karin. "The Shrinking of Higher Ed." *The Chronicle of Higher Education*, August 12, 2022. <u>https://www.chronicle.com/package/the-shrinking-of-higher-ed</u>.
- Fischer, Karin, and Sasha Aslanian. "Fading Beacon." *The Chronicle of Higher Education*, August 2, 2021. https://www.chronicle.com/article/fading-beacon.
- Fletcher, Bevin. "FAST Ad Spend to Surpass That of Cable, Broadcast, SVOD by 2025: TVREV." *StreamTV Insider*, January 12, 2023. <u>https://www.streamtvinsider.com/advertising/fast-ad-spend-surpass-</u> <u>cable-broadcast-or-svod-2025-tvrev</u>.
- Flinders, Mesh, and Molly Hayes. "A Guide to AI in Marketing." IBM, March 10, 2025. https://www.ibm.com/think/topics/ai-in-marketing.
- *Forbes.* "11 Risks to Using AI In Marketing (And How To Mitigate Them)." April 25, 2024. <u>https://www.forbes.com/councils/forbescommunicationscouncil/2024/04/25/11-risks-to-using-ai-in-marketing-and-how-to-mitigate-them/</u>.
- Freedom House. United Arab Emirates. 2021. <u>https://freedomhouse.org/country/united-arab-emirates/freedom-world/2021</u>.
- Fubini, David. "Are Management Consulting Firms Failing to Manage Themselves?" Working Knowledge. Harvard Business School, July 9, 2024. <u>https://www.library.hbs.edu/working-knowledge/management-consulting-firms-are-failing-to-manage-themselves</u>.

- Fuller, Joseph B., Manjari Raman, Allison Bailey, Nithya Vaduganathan, et al, "Building the On-Demand Workforce." Harvard Business School and Boston Consulting Group, November 2020. <u>https://www.hbs.edu/managing-the-future-of-work/research/Pages/building-the-on-demand-workforce.aspx</u>.
- Gandhi, Prashant, Somesh Khanna, and Sree Ramaswamy. "Which Industries Are the Most Digital (and Why)?" *Harvard Business Review*, April 1, 2016. <u>https://hbr.org/2016/04/a-chart-that-shows-which-industries-are-the-most-digital-and-why</u>.
- Gardner, Lee, and Audrey Williams June. "The Perilous Predicament of the Very Small College." *The Chronicle of Higher Education*, August 17, 2022. <u>https://www.chronicle.com/article/the-perilous-predicament-of-the-very-small-college</u>.
- Gillis, Alanna and Laura Krull. "COVID-19 Remote Learning Transition in Spring 2020: Class Structures, Student Perceptions, and Inequality in College Courses." *Teaching Sociology, vol. 48(4) (October 2020): 283–299.* https://pmc.ncbi.nlm.nih.gov/articles/PMC7509240/.
- Gitelman, L. D., V. Silbermann, A. Y. Makarov, and D. G. Sandler. "Energy Engineering and Consulting: New Challenges and Reality." *International Journal of Energy Production and Management* 5, no. 3 (2020): 272–84. <u>https://doi.org/10.2495/EQ-V5-N3-272-284</u>.
- Gradstein, Mark, and Michael Kaganovich. "Aging Population and Education Finance." Journal of Public Economics 88, no. 12 (December 2004): 2469–85. <u>https://doi.org/10.1016/S0047-</u> 2727(03)00065-3.
- Gruenwedel, Erik. "Leichtman: 2023 U.S. Household SVOD Subscription Growth Flat." *Media Play News*, August 14, 2023. <u>https://www.mediaplaynews.com/leichtman-2023-us-household-svod-</u> <u>subscription-growth-flat/</u>.
- Harlow, Stephanie. "How Effective Are Ads on Social Media?" *GWI*, June 5, 2023. <u>https://blog.gwi.com/trends/ads-on-social-media/</u>.
- Hawkins, Amy. "China's Kindergarten Numbers Shrink as Policymakers Struggle to Arrest Falling Birthrate." *The Guardian*, October 28, 2024. <u>https://www.theguardian.com/world/2024/oct/28/chinas-kindergarten-numbers-shrink-as-policymakers-struggle-to-arrest-falling-birthrate</u>.
- IBISWorld. *Global Management Consultants*. December 2024. <u>https://www.ibisworld.com/default.aspx</u>.
- IBISWorld. *Management Consulting in the US*. August 2024. <u>https://www.ibisworld.com/united-states/market-research-reports/management-consulting-industry/</u>.
- Inflation Reduction Act of 2022, H.R.5376 (2021-2022). <u>https://www.congress.gov/bill/117th-congress/house-bill/5376/text</u>.
- Interactive Advertising Bureau (IAB). 2024 Digital Video Ad Spend & Strategy Report; Part One: Market Size and Growth Projections. April 2024. <u>https://www.iab.com/wp-</u> <u>content/uploads/2024/04/IAB_Video_Ad_Spend_Study_Part_1_April_2024.pdf</u>.

- Jacobson, Adam. "Ad Revenue Decrease Impacts Disney's Linear Network." *Radio and Television Business Report*, November 9, 2023. <u>https://rbr.com/ad-revenue-decrease-impacts-disneys-linear-networks-in-fiscal-q4/</u>.
- Jakaityte, Indre. "The Sustainability Surge: Key Players and Market Dynamics in E&S Consulting." Environment Analyst Global, February 20, 2025. <u>https://environment-analyst.com/110810/the-sustainability-surge-key-players-and-market-dynamics-in-es-consulting</u>.
- Jaumotte, Florence, Myrto Oikonomou, Carlo Pizzinelli, and Marina Tavares. "How Pandemic Accelerated Digital Transformation in Advanced Economies." International Monetary Fund. *IMF Blog*, March 21, 2023. <u>https://www.imf.org/en/Blogs/Articles/2023/03/21/how-pandemic-accelerateddigital-transformation-in-advanced-economies.</u>
- Kaplan, Soren. "The Business Consulting Industry Is Being Disrupted, and Nothing Can Stop It." *Inc.*, August 12, 2024. <u>https://www.inc.com/soren-kaplan/the-business-consulting-industry-is-being-disrupted-nothing-can-stop-it.html</u>.
- Kennedy Consulting. "Insights and Resources." Accessed December 11, 2024. https://www.kennedyconsulting.com/insights-and-resources.
- Kent, Daniel C. "Challenges in a Disrupted World: Branch Campuses from the United States." International Higher Education 104 (October 15, 2020): 14–15. <u>https://ejournals.bc.edu/index.php/ihe/article/view/14343</u>.
- Kent, Daniel C. "The Implications of Campus Closures." *International Higher Education* 110 (2022): 23–24. <u>https://ihe.bc.edu/pub/cjwwwuby/release/1</u>.
- Kinser, Kevin, and Joseph E. Lane. *C-BERT International Campus Listing*. Oxford, OH: Cross-Border Education Research Team, March 23, 2023. <u>https://www.cbert.org/intl-campus</u>.
- Kotschy, Rainer, David E Bloom, and Andrew J Scott. "On the Limits of Chronological Age." National Bureau of Economic Research. NBER Working Paper no. 33124, November 2024. <u>https://www.nber.org/system/files/working_papers/w33124/w33124.pdf</u>.
- Kumar, Naveen. "29 Video Marketing Statistics 2025 (Latest Data & Trends)." DemandSage, March 6, 2025. <u>https://www.demandsage.com/video-marketing-statistics/</u>.
- Laffitte, Helene. "2024 In Review: Transformations Shaping Global Consulting." Consulting Quest, November 25, 2024. <u>https://consultingquest.com/insights/2024-global-consulting-trends/</u>.
- Lebow, Sara. "New Data: Linear TV Fell below 50% Viewing Share in July for the First Time." EMarketer, August 16, 2023. <u>https://content-naf.emarketer.com/new-data-linear-tv-fell-below-50-viewing-share-july-first-time</u>.
- Lebow, Sara. "Guide to Connected TV: Content, Measurement, and Advertising." EMarketer, September 5, 2024. <u>https://www.emarketer.com/insights/guide-ctv/</u>.

- Legon, Ron, and Richard Garrett. *CHLOE 8: Student Demand Moves Higher Ed Toward a Multi-Modal Future*. Quality Matters, Eduventures, 2023. <u>https://www.qualitymatters.org/qa-resources/resource-center/articles-resources/CHLOE-project</u>.
- Leichtman Research Group (LRG). "Actionable Research on the Broadband, Media & Entertainment Industries." 2023. <u>https://leichtmanresearch.com/wp-content/uploads/2023/12/LRG-Research-Notes-4Q-2023.pdf</u>.
- Leichtman Research Group (LRG). "Major Pay-TV Providers Lost About 2,875,000 Subscribers In 2018." Press release, March 6, 2019. <u>https://www.leichtmanresearch.com/wp-content/uploads/2019/03/LRG-Press-Release-03-06-2019.pdf</u>.
- Leichtman Research Group (LRG). "Major Pay-TV Providers Lost About 5,000,000 Subscribers In 2023." Press release, March 8, 2024. <u>https://leichtmanresearch.com/wp-</u> <u>content/uploads/2024/03/LRG-Press-Release-3-08-2024.pdf</u>.
- Long, Kyle, and Saiansha Panangipalli. "Under the Radar: American Universities Abroad." *International Higher Education* No. 15 (Summer 2023). <u>https://ihe.bc.edu/pub/jicop357/release/1</u>.
- Lu, Adrienne. "Some Colleges Are Ending Hybrid Learning. Students Are Pushing Back." *The Chronicle of Higher Education*, April 11, 2022. <u>https://www.chronicle.com/article/some-colleges-are-ending-hybrid-learning-students-are-pushing-back</u>.
- Majidi, Michele. "Streaming Services Advertising in the United States—Statistics & Facts." Statista, September 11, 2024. <u>https://www.statista.com/topics/11807/streaming-services-advertising-in-the-united-states/#topicOverview</u>.
- Marcus, Jon. "A Looming 'Demographic Cliff': Fewer College Students and Ultimately Fewer Graduates." NPR, January 8, 2025. <u>https://www.npr.org/2025/01/08/nx-s1-5246200/demographic-cliff-fewer-college-students-mean-fewer-graduates</u>.
- MarketLine. United States (US) Management and Marketing Consultancy Market Summary, Competitive Analysis, and Forecast to 2028. March 2024. <u>https://store.marketline.com/report/usa-management-and-marketing-consultancy-market-analysis/</u>.
- Marr, Bernard. "Artificial Intelligence and The Future Of Marketing." *Forbes*, September 9, 2022. <u>https://www.forbes.com/sites/bernardmarr/2022/09/09/artificial-intelligence-and-the-future-of-marketing/</u>.
- MBO Partners. *The Independent by Choice Movement: Authentic and Intentional.* Accessed January 16, 2025. <u>https://www.mbopartners.com/state-of-independence/</u>.
- McKinsey & Company. "Social Commerce: The Future of How Consumers Interact with Brands." McKinsey & Company, October 19, 2022. <u>https://www.mckinsey.com/capabilities/growth-</u> <u>marketing-and-sales/our-insights/social-commerce-the-future-of-how-consumers-interact-with-</u> <u>brands</u>.
- McKinsey & Company. "What Is Social Media?" June 8, 2023. <u>https://www.mckinsey.com/featured-insights/mckinsey-explainers/what-is-social-media</u>.

- McMurtrie, Beth. "Classroom Walls Are Shifting." *The Chronicle of Higher Education*, October 18, 2023. <u>https://www.chronicle.com/article/classroom-walls-are-shifting?cid=gen_sign_in</u>.
- NAFSA: Association of International Educators. *The United States of America Benefits from International Students*. November 18, 2024. https://www.nafsa.org/sites/default/files/media/document/EconValue2024.pdf.
- Nagarajan, Ravi. "The Future of Consulting." CIO, November 23, 2022. https://www.cio.com/article/413101/the-future-of-consulting.html.
- Nagl, Steph, and Ged King. "How AI Is Impacting Digital Marketing." Wake Forest University School of Professional Studies, August 27, 2024. <u>https://sps.wfu.edu/articles/how-ai-impacts-digital-marketing/</u>.
- National Center for Education Statistics. "84% of All Undergraduates Experienced Some or All Their Classes Moved to Online-Only Instruction Due to the Pandemic." Press Release, June 16, 2021. <u>https://nces.ed.gov/whatsnew/press_releases/06_16_2021.asp</u>.
- Nielsen Company. "Streaming Grabs a Record 38.7% of Total TV Usage in July, with Acquired Titles Outpacing New Originals." August 2023. <u>https://www.nielsen.com/insights/2023/streaming-grabs-a-record-38-7-of-total-tv-usage-in-july-with-acquired-titles-outpacing-new-originals/?utm_campaign=nielsen-audience-outcomes_nlsn_&utm_content=the-gauge_2023&utm_source=twitter&utm_medium=organic-social.</u>
- Oarue-Itseuwa, Esioze. "Artificial Intelligence's Impact of the Management Consultancy Sector over the Next Five Years." *Management Consulting Journal* 7, no. 1 (February 2, 2024). <u>https://doi.org/10.2478/mcj-2024-0005</u>.
- Open Doors. 2024 Fast Facts. Institute of International Education, November 25, 2024. https://opendoorsdata.org/fact_sheets/fast-facts/.
- Partis, Emma. "Consulting Contracts for Mid-Tier and Small Firms on the Rise, Says Anchoram." *Accounting Times*, January 30, 2025. <u>https://www.accountingtimes.com.au/profession/consulting-trends-in-the-post-pwc-scandal-world-harry-cheema-anchoram</u>.
- PricewaterhouseCoopers (PwC). "Perspectives from the Global Entertainment & Media Outlook 2024– 2028." July 16, 2024. <u>https://www.pwc.com/gx/en/issues/business-model-</u> <u>reinvention/outlook/insights-and-perspectives.html</u>.
- Princeton University. "Incoming Students for Fall 2022." Accessed January 23, 2025. https://gradschool.princeton.edu/incoming-students.
- ProcureCon Marketing. "The Legal and Privacy Implications of AI in Marketing." November 2025. https://procurecondm.wbresearch.com/blog/legal-privacy-implications-ai-marketing-strategy.
- Project Atlas. *Global Mobility Trends*. Institute of International Education, 2024. <u>https://www.iie.org/research-initiatives/project-atlas/</u>.

- Richter, Stephane. "The Consulting Industry at a Crossroads: How Layoffs Reflect Deeper Changes in Supply Chain Consulting." *SupplyChains Magazine*, January 31, 2025. <u>https://supplychains.com/the-consulting-industry-at-a-crossroads-how-layoffs-reflect-deeperchanges-in-supply-chain-consulting/.</u>
- Rieppi, Francesco. "Consulting Industry Trends in 2023." PrepLounge, May 25, 2023. https://www.preplounge.com/en/articles/consulting-industry-trends.
- Roberts-Grmela, Julian. "More Students Want Virtual-Learning Options. Here's Where the Debate Stands." *The Chronicle of Higher Education*, June 5, 2023. <u>https://www.chronicle.com/article/more-students-want-virtual-learning-options-heres-where-the-debate-stands</u>.
- Ross, Andrew. "Is the Offshore Campus Boom Over?" *The Chronicle of Higher Education*, April 28, 2022. <u>https://www.chronicle.com/article/is-the-offshore-campus-boom-over</u>.
- Samokhvalov, Konstantin. "The Transformative Impact of Artificial Intelligence on the Management Consultancy Sector." *Management Consulting Journal* 7, no. 1 (February 2, 2024): 60–68. <u>https://doi.org/10.2478/mcj-2024-0006</u>.
- Salesforce. "Transforming Consulting Through Generative AI." *Harvard Business Review* (sponsored content), May 13, 2024. <u>https://hbr.org/sponsored/2024/05/transforming-consulting-through-generative-ai</u>.
- Shaban, Hamza. "Digital Advertising to Surpass Print and TV for the First Time, Report Says." *The Washington Post*, February 20, 2019. <u>https://www.washingtonpost.com/technology/2019/02/20/digital-advertising-surpass-print-tv-first-time-report-says/</u>.
- Shah, Rachna, and Karishma Borkakoty. "Social Media Trends and Predictions for 2025." Sprinklr, December 19, 2024. <u>https://www.sprinklr.com/blog/social-media-trends/</u>.
- Shopify. "Top 5 AI Content Creation Tools." September 30, 2024. <u>https://www.shopify.com/blog/ai-content-creation</u>.
- Shorelight. "Global Student Demand Picks Up, U.S. Loses International Student Market Share to Competitors." May 22, 2023. <u>https://shorelight.com/news-and-insights/global-student-demand-picks-up-us-loses-international-student-market-share/</u>.
- Spar, Debora. "Today's Awkward Zoom Classes Could Bring a New Era of Higher Education." *EdSurge*, September 10, 2020. <u>https://www.edsurge.com/news/2020-09-10-today-s-awkward-zoom-classes-could-bring-a-new-era-of-higher-education</u>.
- Stanford Graduate School of Education. "What Is Synchronous and Asynchronous Learning?" Classroom Resources. Accessed January 22, 2025. <u>https://teachingresources.stanford.edu/resources/what-is-synchronous-and-asynchronous-learning/</u>.
- Stanley, Holly. "What Is Social Commerce? Trends and Key Insights for 2024." *Shopify* (blog), April 29, 2024. <u>https://www.shopify.com/enterprise/blog/social-commerce-trends</u>.

- Statista. Advertising: Market Data & Analysis. November 2024. https://www.statista.com/study/166789/advertising-market-data-and-analysis/.
- Statista. "Digital Ad Spending." October 2024. https://www.statista.com/outlook/amo/advertising/worldwide#digital-ad-spending.
- Statista. "Professional Services—China." June 2024. <u>https://www.statista.com/outlook/tmo/cybersecurity/security-services/professional-</u> <u>services/china</u>.
- Statista. "Professional Services—India." June 2024. <u>https://www.statista.com/outlook/tmo/cybersecurity/security-services/professional-services/india</u>.
- Steinberg, Brian, and William Earl. "Advertisers Ride the Brakes: Sharp Drops in TV Spending Make Media Companies Vulnerable." Variety, January 18, 2024. <u>https://variety.com/2024/tv/news/advertisers-drop-television-spending-hollywood-vulnerable-1235875749/</u>.
- Sutter, Felix. "Rethinking Supply Chains after China's Lockdown." PricewaterhouseCoopers, July 7, 2022. <u>https://www.pwc.de/en/international-markets/german-business-groups/china-business-group/rethinking-supply-chains-after-chinas-lockdown.html</u>.
- Toner, Mark. "Branching Out." NAFSA: Association of International Educators, September 1, 2019. https://www.nafsa.org/ie-magazine/2019/9/1/branching-out.
- U.S. Department of Commerce (USDOC). Bureau of Economic Analysis (BEA). Table 2.1. "U.S. Trade in Services, by Type of Service." July 3, 2024. https://apps.bea.gov/iTable/?reqid=62&step=9&isuri=1&product=4&_gl=1*ci062r*_ga*MTAxNj M4Mzg1OS4xNzM2OTE3NDE2*_ga_J4698JNNFT*MTczOTk1MTc0NS41LjEuMTczOTk1MTkyMS42 MC4wLjA.#eyJhcHBpZCI6NjlsInN0ZXBzIjpbMSw5LDZdLCJkYXRhIjpbWyJwcm9kdWN0liwiNCJdLFsi VGFibGVMaXN0liwiMjQ1II1dfQ==.
- U.S. Department of State (USDOS). Bureau of Education and Cultural Affairs (ECA). "What Is the Fulbright U.S. Student Program." U.S. Fulbright Program. Accessed January 27, 2025. https://us.fulbrightonline.org/about/fulbright-us-student-program.
- U.S. International Trade Commission (USITC). COVID-19 Related Goods: The U.S. Industry, Market, Trade, and Supply Chain Challenges. Publication no. 5145, December 2020. https://www.usitc.gov/publications/332/pub5145.pdf.
- U.S. International Trade Commission (USITC). *Recent Trends in U.S. Services Trade: 2013 Annual Report*. Publication no. 4412, July 2013. <u>https://www.usitc.gov/publications/332/pub4412.pdf</u>.
- U.S. International Trade Commission (USITC). *Recent Trends in U.S. Services Trade: 2021 Annual Report*. Publication no. 5192, April 2021. <u>https://www.usitc.gov/publications/332/pub5192.pdf</u>.
- Verot, Olivier. "Social Commerce in China: Sell on Chinese Social Media." Gentlemen Marketing Agency (GMA), December 11, 2024. <u>https://marketingtochina.com/social-commerce-in-china/</u>.

- Vimond. "What Are SVOD, AVOD, and TVOD?" Accessed January 17, 2025, https://www.vimond.com/post/what-are-svod-avod-and-tvod.
- Vitaud, Laetitia. "Consulting Will Never Be the Same Again." *Want More Work*. June 15, 2016. <u>https://medium.com/want-more-work/consulting-will-never-be-the-same-again-210bc3e37951</u>.
- Walt Disney Company. "The Walt Disney Company Reports Fourth Quarter and Full Year Earnings for Fiscal 2023." November 8, 2023. <u>https://thewaltdisneycompany.com/the-walt-disney-company-reports-fourth-quarter-and-full-year-earnings-for-fiscal-2023/</u>.
- Wilkins, Stephen. "International Branch Campuses, Current Trends and Future Possibilities." *International Higher Education* 110 (2023): 8–9. <u>https://ejournals.bc.edu/index.php/ihe/article/view/17353</u>.
- Winslow, George. "Major Pay TV Providers Lost Record 5M Subs in 2023." TV Tech, March 8, 2024. https://www.tvtechnology.com/news/major-pay-tv-providers-lost-record-5m-subs-in-2023.
- Wohr, James. "OTT Advertising: What It Is and Why Streaming Matters to Marketers." EMarketer, July 18, 2024. <u>https://www.emarketer.com/insights/ott-video-streaming-services/</u>.
- World Bank Group. "World Development Indicators." Accessed January 6, 2025. https://databank.worldbank.org/source/world-development-indicators#.
- Younger, Jon. "Freelance Management Consulting: Back to Growth." *Forbes*, May 20, 2023. <u>https://www.forbes.com/sites/jonyounger/2023/05/17/freelance-management-consulting-back-to-growth/</u>.
- Younger, Jon. "CEOs Explain How AI Will Super Charge Independent Management Consulting." Forbes, September 5, 2023. <u>https://www.forbes.com/sites/jonyounger/2023/09/05/ceos-explain-how-ai-will-super-charge-independent-management-consulting/</u>.
- Yuen, Meaghan. "Traditional Search Will Draw the Largest Amount of US Ad Dollars in 2024." EMarketer, June 14, 2024. <u>https://content-naf.emarketer.com/traditional-search-draws-largest-amount-of-us-ad-dollars</u>.
- Yuen, Meaghan. "What Media Planners Need to Know about Free Ad-Supported Streaming TV." EMarketer, February 26, 2024. <u>https://content-naf.emarketer.com/what-media-planners-need-know-about-free-ad-supported-streaming-tv</u>.

Chapter 5 Roundtable Summary

Each year, the U.S. International Trade Commission (Commission or USITC) hosts a roundtable to facilitate conversations among individuals from government, industry, and academia on issues affecting services trade.⁵⁸⁸ The 18th Annual Services Roundtable was held on Wednesday, October 30, 2024, at Commission headquarters. This year's roundtable focused on the discussion question: "How can we ensure that the United States has the workforce it needs to be competitive in global services markets?" Following introductory remarks by Commission Chair Amy Karpel, Commissioner David S. Johanson moderated the discussion.

Persistent Labor Shortages Reported in Services Industries

In response to the question, "Where do you see significant labor shortages and skills gaps," roundtable participants suggested that widespread workforce gaps affect U.S. services industries. A participant from a business advocacy group cited worker shortage index data indicating that only 85 workers are available for every 100 job openings nationwide, with even fewer available in specific industries such as professional services (50 workers per 100 jobs) and financial services (60 workers per 100 jobs).

A participant from a management consulting firm said that the labor shortage is a global challenge, particularly in developed economies and parts of emerging East Asia. Stable demand, aging demographics, and constrained labor supply are creating a structural shortage that is not limited to the United States or Japan but is instead a worldwide issue with regional differences. The participant also discussed the intersection of industrial policy and labor shortages, stating that sectors such as manufacturing, construction, and supply chains are facing both supply-side constraints (aging workforce, health-care issues, limited immigration) and demand-side pressures driven by large-scale investments in infrastructure and technology. Another participant from an industry-led body cited a UK survey indicating that 57 percent of participants in the professional and business services industries reported labor shortages and difficulty in recruiting, indicating that workforce shortages are an issue in other developed economies as well as the United States.

The engineering profession was cited as experiencing shortages, with a projected need for 23,000 civil engineers annually but only 12,000–13,000 graduates entering the workforce each year. A participant from a legal services firm noted the challenge of opening offices outside the United States due to restrictions on hiring locally licensed lawyers in foreign countries. Even though many foreign nationals attend U.S. law schools and take the bar exam—particularly in states like New York—navigating these restrictions remains a significant hurdle for legal firms looking to expand internationally.

⁵⁸⁸ The Services Roundtable is an off-the-record event; its participants are not named in this summary and no transcript is available to the public.

Another participant stated that the legal profession faces geographic labor imbalances, with one-quarter of the 1.3 million U.S. lawyers concentrated in New York and California; regions considered "legal deserts"—such as South Carolina, Arizona, and Idaho—had significantly fewer attorneys. The participant noted that the legal profession was facing challenges due to declining enrollment in law schools and the need for lawyers in underserved regions.

Education and Development Initiatives

The discussion focused on gaps in education and workforce preparedness. A participant from the Commission noted that U.S. schools and universities produce a significant number of graduates but many struggle to find employment within the first two years. The participant cited studies indicating that graduates who fail to secure a strong entry-level position in the first two years are more likely to remain underemployed 10 years later. The participant also noted that schools and universities are providing needed technical skills but that certain soft skills—such as professional communication skills—are missing from the emerging workforce. A participant from a business advocacy group highlighted the importance of digital skills in the services industry and noted the skills gap particularly in high-skilled services such as information technology (IT) research and design. The participant said that, because of this gap, a wide-ranging effort exists across the services sector to provide worker reskilling programs—primarily digitally focused—beginning at the high school level.

Several participants expressed the view that a greater investment in apprenticeships and on-the-job training programs may help to bridge the skills gap. Participants from industries such as digital services, insurance, and engineering suggested that structured apprenticeships and mentoring programs could help workers acquire both technical and soft skills. One participant discussed how the efforts to diversify the Las Vegas economy after the COVID-19 pandemic led to a downturn in the city's tourism industry. The participant noted that the lack of skilled labor in artificial intelligence (AI) and digital services made it hard to attract businesses to the region, despite advantages such as a low-income tax rate and an affordable cost of living. The participant also said that to address these challenges, a local university has launched a paid internship program that funds student placements at startups and small businesses.

Characteristics of the U.S. Services Workforce

As part of a discussion on workforce characteristics, a participant from academia noted that workers in tradable services industries—such as financial services, legal services, and consulting services—tend to have higher education levels than the average U.S. worker. The participant noted that these workers are largely concentrated in major metropolitan areas and earn relatively high wages.

A participant representing the insurance industry indicated that the majority of new workers in that industry are women. Another participant, from the film and television industry, reported achieving relatively balanced gender representation with a highly unionized workforce, and noted that union jobs include positions at all skill levels, many of which do not require a college degree. A participant from the legal services industry shared that only one in five lawyers is a person of color (a term used by the participant), and 39 percent of lawyers are women. A participant from the architectural services industry noted that women make up 26 percent of that workforce, which represents significant historical growth.

The participant said that the architecture industry suffers from a gap in mid-career professionals, largely due to the "Great Recession," which poses a challenge for succession planning and the future of firms.

Role of Foreign Nationals in the U.S. Services Workforce

In response to questions on the role of foreign workers in the U.S. services industry, an engineering participant discussed the impact of government programs such as the Bipartisan Infrastructure Law, the CHIPS and Science Act, and the Inflation Reduction Act (IRA) on labor demand. The participant noted that sustained demand and aging demographics create a trend of deepening workforce gaps in the United States and other developed economies. The participant reported that migrants help fill the workforce gaps created by aging demographics and declining birth rates experienced in developed economies. A participant from the finance industry discussed the importance of foreign expertise and foreign management. The participant noted that the ability to transfer foreign senior management between countries has been gradually constrained in the last five years and that this has restricted competitiveness for firms that are now not able to fluidly move the best personnel for senior positions. Another participant, speaking broadly about all industries, added that U.S. competitiveness is hampered by limitations on international staff mobility due to taxation issues, the ability to move pensions across borders, and recognition of the qualifications of foreign workers.

A participant representing Australia discussed Australia's skills shortage and its effort to increase skilled migration. The participant said that Australia uses a point system based on qualifications to fill workforce gaps in targeted industries. The participant also shared information about the U.S.-Australia skilled migration visa program—associated with the 2005 U.S.-Australia Free Trade Agreement—which allows qualified Australians to work in the United States. The participant noted that the program has never reached the maximum-agreed number of visas of 10,500, likely because efforts to address skills gaps are not centralized in the United States. Participants from the engineering and legal services industries discussed challenges of licensing and credential recognition between the United States and foreign countries. A participant noted that the United States has 55 engineering licensing jurisdictions and that can be a barrier to foreign engineers filling workforce gaps. Reciprocity agreements, which have been made by several states, may help to overcome the barriers, and more jurisdictions may adopt the agreements over time.

Several participants from academia agreed on the importance of foreign students as a source of revenue for public and private U.S. universities. Participants viewed the visa system as an impediment to maintaining universities as a healthy supply of skilled labor for the U.S. services workforce. A participant from the film industry commented that visa turnarounds from the U.S. State Department have been remarkably quick for talent and crew from overseas to come to the United States and that 2023 saw the highest number of visas processed in the industry since 2017.

The Role of AI in Reshaping Services

A common theme of the roundtable discussion was the impact of digital transformation, particularly by AI, on the services workforce. A participant from the digital services industry cited data indicating that AI

is expected to augment or replace a significant portion of tasks across the broad workforce. A participant from academia noted firm-level research showed that a higher level of AI exposure increases demand for high-skilled workers.

Several participants noted that AI adoption is not uniform across industries. Sectors such as legal services, financial services, and professional consulting are among those most exposed to AI-driven automation; in contrast, fields like health care, construction, and logistics remain relatively insulated due to licensing requirements and their reliance on human interaction and physical tasks. A participant from the technology services industry remarked that AI technology is a fast-changing space in the early days of use and adoption—that people are moving beyond chat bots toward broader adoption.

AI and Workforce Adaptation

Participants discussed how reskilling programs have helped workers adapt to changing job requirements. A participant from a professional services platform said that 80 percent of workers will have their jobs augmented in some way by generative AI (GenAI) and noted that effectively no one's job will look the same five years from now. A participant from the consulting services industry estimates only a 20 percent addition to employment disruption caused by GenAI compared to pre-GenAI disruption. Despite the uncertainty, the participant expected a lower level of disruption than previous industrial revolutions.

One participant from a policy think tank referenced a study that suggests AI has the greatest positive impact on productivity when it supports lower-skilled or novice workers, allowing them to perform tasks more efficiently. The participant noted that the concern of AI replacing novice or lower-skilled workers is well-placed—compared to a minimal impact for higher-skilled workers—but also suggested that AI is an opportunity to enhance productivity and learning. Another participant indicated that the engineering profession is exploring AI-driven design tools and automation in infrastructure planning, though human judgment remains crucial in decision-making.

Several participants mentioned the importance of developing a national AI and digital skills strategy. Additionally, several industry representatives expressed support for stronger collaboration among government, businesses, and educational institutions to ensure that the workforce is equipped with needed technical and problem-solving skills. One participant from a think tank noted that current educational systems tend to focus on technical skills but neglect critical thinking and workplace communication, which are becoming increasingly important in the AI-driven economy.

A participant from a policy think tank discussed the role of apprenticeships in workforce development, highlighting that registered apprenticeship programs provide structured, paid training opportunities with built-in career progression. The participant reported that these programs are particularly useful for reskilling workers impacted by automation and AI.

Challenges and Opportunities in AI Implementation

A participant from the legal services industry noted that AI is already transforming legal work—a highskilled service—particularly in transactional law, due diligence, and contract analysis. The participant indicated that the shift raises questions about ethics (including confidentiality issues), regulatory oversight, and the future role of entry-level lawyers. The participant also noted how rapidly GenAI is becoming a substitute for many of the traditional skillsets of lawyers. The participant cited a Thomson Reuters survey that found that GenAI is the number one issue that will impact law firms in the next five years. Another challenge discussed by this participant was pricing. The participant indicated that legal services have traditionally been priced using billable hours and GenAI will fundamentally alter that paradigm. The participant suggested that broader adoption will depend on factors such as regulatory clarity, data privacy protections, and investment in infrastructure.

Another participant from a trade association compared Al's impact to the advent of electricity, noting that AI has the potential to revolutionize business operations but disparities in access to digital infrastructure and training could widen economic inequality. Several participants raised concerns about the digital divide, particularly in rural areas and other communities in which access to AI-driven tools and training remains limited.

Recent Trends in U.S. Services Trade: 2025 Annual Report

Appendix A: Selected Services-Related Commission Publications and Staff Research

Appendix A Selected Services-Related Commission Publications and Staff Research

This appendix provides summaries of and links to recent U.S. International Trade Commission (USITC or Commission) publications—reports and shorter papers—that feature topics in services trade. Reports are prepared under section 332(g) of the Tariff Act of 1930 (19 U.S.C. 1332(g)) in response to a request from the U.S. Trade Representative, the U.S. House of Representatives Committee on Ways and Means, the U.S. Senate Committee on Finance, or either chamber in Congress. The working paper and Executive Briefing on Trade summarized below present research by the Commission's Services Division staff on services trade topics. These shorter papers solely represent the opinions and professional research of their authors. They are not meant to represent in any way the views of the Commission, any of its individual Commissioners, or the U.S. government.

332 Investigations

Recent Trends in U.S. Services Trade: 2024 Annual Report

Investigation Number 332-601, May 2024

https://www.usitc.gov/publications/industry_econ_analysis_332/2024/recent_trends_us_services_trade 2024_annual_report

Rudy Telles Jr. (Office of Industry and Competitiveness Analysis, Services Division) and Theron Gray (Office of Industry and Competitiveness Analysis, Services Division)

This report describes recent trends in U.S. services trade and developments in U.S. services industries' competitiveness. Specifically, it focuses on recent developments in U.S. trade in financial services, notably in the context of a pronounced increase in global inflation from 2021 to 2022 and subsequent volatility in global interest rates. The report also includes specific discussions of recent global developments in banking, insurance, and securities services, which are addressed in two chapters that focus on two overarching themes. One discusses how financial services firms are embracing digitalization and looking to incorporate new artificial intelligence technologies. The other focuses on the rising demand for new financial products and services that address concerns for sustainability and financial inclusion as well as support growth in emerging markets.

Working Papers

"Estimating Trade Costs of Non-Tariff Measures in Services: A Comparison of Methods and Measures"

https://www.usitc.gov/publications/332/working_papers/estimating_trade_costs_of_nontariff_mesaures_in_services.pdf

Sarah Oliver (Office of Industry and Competitiveness Analysis, Services Division), June 2024

This paper considers the effectiveness of recent methodologies proposed by Heid, Larch, and Yotov (2021) and Herman (2022) to estimate tariff equivalent trade costs for non-tariff measures (NTMs) in

cross-border services trade. Both estimation methods show that for most services trade categories across four indices measuring services NTMs, tariff equivalent trade costs are significantly higher for exporters to countries with stricter NTMs, suggesting that both methodologies are successful at capturing variation in trade costs across different markets. Additionally, for transportation, banking, and insurance services, the difference in estimated tariff equivalent trade costs between the Heid, Larch, and Yotov (2021) and Herman (2022) methods were smaller than 5 percentage points, suggesting that the Herman (2022) methodology could be useful in cases where collection of domestic trade data required for the Heid, Larch, and Yotov (2021) method are unfeasible.

Executive Briefings on Trade

"High-Income Countries Account for Small Share of UK Skilled Worker Visa Issuances"

https://www.usitc.gov/publications/332/executive_briefings/ebot_issuances_of_uk_skilled_worker_visa s.pdf

Jennifer Powell (Office of Industry and Competitiveness Analysis, Services Division), July 2024

Following the enactment of new immigration rules in December 2020, UK visa issuances to skilled workers grew substantially. However, trends in issuances to higher-income country workers—including U.S. workers—differed from issuances to workers from other countries. This brief reviews trends in UK visa issuances to skilled workers, an indicator of Mode 4 services trade, focusing on issuances to U.S. and other high-income country nationals.

Recent Trends in U.S. Services Trade: 2025 Annual Report

Appendix B Corresponding Tables: Corresponding Data Tables for Figures

Table B.1 Real value added by U.S. industry, 2019–23

In trillions of dollars. This table corresponds to figure 1.2.					
Type of industry	2019	2020	2021	2022	2023
Private goods-producing industries	3.6	3.5	3.6	3.6	3.7
Private services-producing industries	14.6	14.3	15.4	15.9	16.4

Source: USDOC, BEA, "Real Value Added by Industry," September 26, 2024.

Notes: Estimates are (inflation-adjusted) 2017 dollars. Private goods-producing industries include agriculture, forestry, fishing, and hunting; mining; construction; and manufacturing. Private service-producing industries include utilities; wholesale trade; retail trade; transportation and warehousing; information; finance, insurance, real estate, rental, and leasing; professional and business services; educational services, health care, and social assistance; arts, entertainment, recreation, accommodation, and food services; and other services, except government services.

Table B.2 Global services: cross-border exports of commercial services, by country, 2023

In billions of dollars (billion \$) and percentages (%). This table corresponds to figure 1.3.

Country	Billion \$	Share of total (%)
United States	994	12.7
United Kingdom	581	7.4
Germany	433	5.5
Ireland	397	5.1
China	380	4.9
France	359	4.6
India	337	4.3
Singapore	328	4.2
Netherlands	309	4.0
Japan	203	2.6
All other countries	3,501	44.8
Total	7,822	100.0

Source: WTO, "Trade in Commercial Services," accessed September 26, 2024.

Notes: Exports of commercial services exclude public-sector transactions. Because of rounding, shares may not add to 100 percent.

Table B.3 Global services: cross-border imports of commercial services, by country, 2023 In billions of dollars (billion \$) and percentages (%). This table corresponds to figure 1.4.

Country	Billion \$	Share of total (%)
United States	723	10.0
China	549	7.6
Germany	500	6.9
United Kingdom	389	5.4
Ireland	389	5.4
France	320	4.4
Netherlands	296	4.1
Singapore	295	4.1
India	245	3.4
Japan	223	3.1
All other countries	3,318	45.8
Total	7,249	100.0

Source: WTO, "Trade in Commercial Services," accessed September 26, 2024.

Notes: Imports of commercial services exclude public-sector transactions. Because of rounding, shares may not add to 100 percent.

III IIIIIIeiis ei aei		
Year	U.S. cross-border exports of private services	U.S. cross-border imports of private services
2019	868,642	569,313
2020	704,320	442,558
2021	782,032	544,469
2022	919,786	688,339
2023	993,829	722,681

Table B.4 U.S. private services: cross-border exports and imports, 2019–23

 In millions of dollars. This table corresponds to figure 1.5.

Source: USDOC, BEA, table 2.1, "U.S. Trade in Services, by Type of Service," July 3, 2024.

Table B.5 U.S. private services: cross-border exports, by category, 2023

In millions of dollars (million \$) and percentages (%). This table corresponds to figure 1.6.

Type of industry	Million \$	Share of total (%)
Professional services	378,177	38.1
Financial services	200,446	20.2
Travel services	174,765	17.6
Digital and electronic services	137,734	13.9
Distribution services	91,582	9.2
All other services	11,125	1.1
Total	993,829	100.0

Source: USDOC, BEA, table 2.1, "U.S. Trade in Services, by Type of Service," July 3, 2024.

Note: Because of rounding, shares may not add to 100 percent.

Table B.6 U.S. private services: cross-border imports, by category, 2023

In millions of dollars (million \$) and percentages (%). This table corresponds to figure 1.7.

Type of industry	Million \$	Share of total (%)
Travel services	211,784	29.3
Professional services	184,694	25.6
Financial services	127,298	17.6
Digital and electronic services	104,543	14.5
Distribution services	87,290	12.1
All other services	7,072	1.0
Total	722,681	100.0

Source: USDOC, BEA, table 2.1, "U.S. Trade in Services, by Type of Service," July 3, 2024. Note: Because of rounding, shares may not add to 100 percent.

Table B.7 U.S. private services: cross-border exports, by country, 2023

In millions of dollars (million \$) and percentages (%). This table corresponds to figure 1.8.

Country	Million \$	Share of total (%)
United Kingdom	90,445	9.1
Canada	85,437	8.6
Ireland	84,328	8.5
Switzerland	49,590	5.0
British Caribbean territories	49,310	5.0
China	46,170	4.6
Mexico	44,050	4.4
Japan	43,623	4.4
Germany	42,030	4.2
All other countries	458,846	46.2

Country	Million \$	Share of total (%)
Total	993,829	100.0

Source: USDOC, BEA, table 2.2, "U.S. Trade in Services, by Type of Service and by Country or Affiliation," July 3, 2024. Notes: Exports of commercial services exclude public sector transactions. Because of rounding, shares may not add to 100 percent. The BEA category "United Kingdom Islands, Caribbean" includes the following UK overseas territories: the British Virgin Islands, the Cayman Islands, Montserrat, and the Turks and Caicos Islands.

Table B.8 U.S. private services: cross-border imports, by country, 2023

In millions of dollars (million \$) and percentages (%). This table corresponds to figure 1.9.

Country	Million \$	Share of total (%)
United Kingdom	85,235	11.8
Canada	54,012	7.5
Germany	45,796	6.3
Mexico	44,772	6.2
India	36,351	5.0
Bermuda	33,545	4.6
Japan	32,953	4.6
Switzerland	30,341	4.2
France	27,301	3.8
All other countries	332,375	46.0
Total	722,681	100.0

Source: USDOC, BEA, table 2.2, "U.S. Trade in Services, by Type of Service and by Country or Affiliation," July 3, 2024.

Notes: Imports of commercial services exclude public sector transactions. Because of rounding, shares may not add to 100 percent.

In millions of dollars. This table corresponds to figure 1.10.			
Year	Sales by U.Sowned foreign affiliates	Purchases from foreign-owned U.S. affiliates	
2018	1,679,254	1,192,047	
2019	1,731,363	1,236,728	
2020	1,663,588	1,162,149	
2021	2,000,384	1,314,053	
2022	2,114,112	1,517,825	

Table B.9 U.S. services: affiliate sales and purchases, 2018–22

Source: USDOC, BEA, table 4.1, "Services Supplied to Foreign Persons by U.S. MNEs," October 8, 2024, and table 5.1, "Services Supplied to U.S. Persons by Foreign MNEs," October 8, 2024.

Notes: MNEs = multinational enterprises. Sales by U.S.-owned foreign affiliates include services supplied by majority-owned foreign affiliates of U.S. parent firms. Purchases from foreign-owned U.S. affiliates includes services supplied by majority-owned U.S. affiliates of foreign parent firms.

Table B.10 U.S. services: affiliate sales by U.S.-owned foreign affiliates by industry, 2022

In millions of dollars (million \$) and percentages (%). This table corresponds to figure 1.11.

Type of industry	Million \$	Share of total (%)
Distribution services	563,279	26.6
Financial services	319,160	15.1
Digital and electronic services	637,373	30.1
Professional services	176,511	8.3
Manufacturing	36,713	1.7
Mining	35,471	1.7
All other services	345,605	16.3
Total	2,114,112	100.0

Source: USDOC, BEA, table 4.1, "Services Supplied to Foreign Persons by U.S. MNEs," October 8, 2024.

Appendix B: Corresponding Tables: Corresponding Data Tables for Figures

Notes: MNEs = multinational enterprises. Professional services are underreported because health care and social assistance services data are suppressed. "Manufacturing" includes ancillary services provided by goods manufacturers. Other services include services supplied by majority-owned foreign affiliates of U.S. parent firms. Distribution services, digital and electronic services, professional services, and all other services are underreported because of suppression of data. Because of rounding, shares may not add to 100 percent.

in binons of donars (binon 3) and percentages (%). This table corresponds to <u>ingure 1.12</u> .			
Type of industry	Billion \$	Share of total (%)	
Distribution services	528,549	34.8	
Financial services	228,215	15.0	
Digital and electronic services	222,299	14.6	
Professional services	171,185	11.3	
Manufacturing	95,463	6.3	
Mining	42,884	2.8	
All other services	229,230	15.1	
Total	1,517,825	100.0	

Table B.11 U.S. services: purchases from foreign-owned U.S. affiliates, 2022 In billions of dollars (billion \$) and percentages (%). This table corresponds to figure 1.12.

Source: USDOC, BEA, table 5.1, "Services Supplied to U.S. Persons by Foreign MNEs," October 8, 2024.

Notes: MNEs = multinational enterprises. Digital and electronic services and all other services are underreported because of the suppression of data. "All other services" includes ancillary services provided in the mining, agriculture, and other sectors, as well as suppressed data. Beginning with the 2018 *Recent Trends in U.S. Services Trade* report, software publishing was reallocated from "All other services" to "Digital and Electronic Services" to better reflect the industry composition. Therefore, digital and electronic services data in this report, and in all *Recent Trends* reports published in 2018 and later, cannot be directly compared with such data in USITC reports published before 2018. Because of rounding, shares may not add to 100 percent.

Table B.12 Professional services: cross-border exports and imports, 2019–23

In billions of dollars (b	illion \$). This table	corresponds to	figure 2.1
---------------------------	------------------------	----------------	------------

Year	U.S. cross-border exports of professional services	U.S. cross-border imports of private services
2019	309	154
2020	291	146
2021	325	167
2022	365	187
2023	378	185

Source: USDOC, BEA, table 2.1, "U.S. Trade in Services, by Type of Service," July 3, 2024.

Note: Total imports and exports include services for education and health care as defined in appendix C, table C.2.

Table B.13 Professional services: U.S. cross-border exports, by country, 2023 Image: Service service

In millions of dollars (million \$) and percentages (%). This table corresponds to figure 2.2.

Country	Million \$	Share of total (%)
Ireland	55,931	14.8
Switzerland	35,288	9.3
Canada	25,895	6.8
Netherlands	25,151	6.7
China	22,016	5.8
United Kingdom	19,979	5.3

U.S. International Trade Commission | 177

Recent Trends in U.S. Services Trade: 2025 Annual Report

Country	Million \$	Share of total (%)
India	14,586	3.9
Singapore	13,048	3.5
Japan	12,671	3.4
Mexico	8,664	2.3
All other countries	144,947	38.3
Total	378,176	100.0

Source: USDOC, BEA, table 2.2, "U.S. Trade in Services, by Type of Service and by Country or Affiliation," July 3, 2024. Note: Because of rounding, figures may not add to 100 percent. Total exports include services for education and health care as defined in

appendix C, table C.2.

Table B.14 Professional services: U.S. cross-border imports, by country, 2023

In millions of dollars (million \$) and percentages (%). This table corresponds to figure 2.3.

Country	Million \$	Share of total (%)
United Kingdom	24,385	13.2
India	16,929	9.2
Canada	13,762	7.5
Germany	12,257	6.6
Japan	11,346	6.1
Ireland	10,759	5.8
Switzerland	7,779	4.2
China	7,138	3.9
Mexico	6,246	3.4
Netherlands	6,175	3.3
All other countries	67,918	36.8
Total	184,694	100.0

Source: USDOC, BEA, table 2.2, "U.S. Trade in Services, by Type of Service and by Country or Affiliation," December 3, 2024. Note: Because of rounding, figures may not add to 100 percent. Total imports include services for education and health care as defined in appendix C, table C.2.

Type of industry	Million \$	Share of total (%)
Research and development services	118,890	31.4
Management consulting services	104,441	27.6
Education services	54,848	14.5
Advertising services	24,295	6.4
Legal services	17,495	4.6
Maintenance and repair services	15,948	4.2
Architectural and engineering services	12,109	3.2
Accounting and auditing services	4,213	1.1
Health services	1,636	0.4
All other	24,649	6.5
Total	378,524	100.0

Table B.15 Professional services: U.S. cross-border exports, by service industry, 2023 In millions of dollars (million \$) and percentages (%). This table corresponds to figure 2.4.

Source: USDOC, BEA, table 2.1, "U.S. Trade in Services, by Type of Service," December 3, 2024.

Note: Because of rounding, figures may not add to 100 percent. Research and development (R&D) services include licenses for the outcomes of R&D. Education includes related travel services. See appendix C, table C.2 for a listing of service sectors included in professional services cross-border trade data.

Table B.16 Professional services: U.S. cross-border imports, by industry, 2023 In millions of dollars (million \$) and percentages (%). This table corresponds to figure 2.5.

Type of industry	Million \$	Share of total (%)
Research and development services	57,274	31.0
Management consulting services	53,672	29.1
Education services	13,664	7.4
Architectural and engineering services	12,681	6.9
Accounting and auditing services	8,378	4.5
Advertising services	6,660	3.6
Maintenance and repair services	6,470	3.5
All other	25,895	14.0
Total	184,694	100.0

Source: USDOC, BEA, table 2.1, "U.S. Trade in Services, by Type of Service," December 3, 2024.

Note: Because of rounding, figures may not add to 100 percent. Research and development (R&D) services include licenses for the outcomes of R&D. Education services include related travel services. See appendix C, table C.2 for a listing of service sectors included in professional services cross-border trade data.

Table B.17 Professional services: sales by U.S.-owned foreign affiliates, by industry, 2022In millions of dollars (million \$) and percentages (%). This table corresponds to figure 2.6.

Type of industry	Million \$	Share of total (%)
Architectural, engineering, and related services	29,805	16.9
Accounting, tax preparation, bookkeeping, and payroll services	16,357	9.3
Scientific research and development services	12,426	7.0
Legal services	9,775	5.5
Educational services	3,967	2.2
Waste management and remediation services	3,759	2.1
Management of nonbank companies and enterprises	2,311	1.3
All other services	98,111	55.6
Total	176,511	100.0

Source: USDOC, BEA, table 4.1, "Services Supplied to Foreign Persons by U.S. MNEs Through Their MOFAs, by Industry of Affiliate and by Country of Affiliate," July 3, 2024.

Notes: MNEs = multinational enterprises. Because of rounding, shares may not add to 100 percent. Of the professional services categories discussed in this report, data on advertising and related services and management, scientific, and technical consulting are suppressed in 2022 to avoid disclosure of individual company data. These sectors are included in the "All other" category.

in minoris of donars (minori ș) and percentages (%). This table corresponds to <u>figure 2.7</u> .			
Type of industry	Million \$	Share of total (%)	
Advertising and related services	42,874	25.0	
Management, scientific, and technical consulting	30,394	17.8	
Architectural, engineering, and related services	22,151	12.9	

Table B.18 Professional services: purchases from foreign-owned U.S. affiliates, 2022 In millions of dollars (million \$) and percentages (%). This table corresponds to figure 2.7

Recent Trends in U.S. Services Trade: 2025 Annual Report

Type of industry	Million \$	Share of total (%)
Scientific research and development services	6,680	3.9
Educational services	3,121	1.8
All other services	65,965	38.5
Total	171,185	100.0

Source: USDOC, BEA, table 5.1, "Services Supplied to U.S. Persons by Foreign MNEs Through Their MOUSAs, by Industry of Affiliate and by Country of UBO," October 8, 2024.

Note: MNEs = multinational enterprises; MOUSAs = majority-owned U.S. affiliates; UBO = ultimate beneficial owner. Because of rounding, figures may not add to 100 percent.

Table B.19 Legal services: U.S. cross-border exports and imports, 2019–23

In millions of dollars. This table corresponds to figure 2.8.

Year	U.S. cross-border exports	U.S. cross-border imports
2019	13,102	4,442
2020	13,680	4,786
2021	16,066	4,973
2022	16,277	5,053
2023	17,495	5,340

Source: USDOC, BEA, table 2.1, "U.S. Trade in Services, by Type of Service," July 3, 2024.

Table B.20 Legal services: U.S. affiliate sales and purchases, 2018–22

Year	Services supplied by U.S. firms' foreign affiliates	Services supplied by U.S. affiliates of foreign firms
2018	7,559	141
2019	7,759	92
2020	8,385	80
2021	9,957	97
2022	9,775	204

In millions of dollars. This table corresponds to figure 2.9.

Source: USDOC, BEA, table 4.1, "Services Supplied to Foreign Persons by U.S. MNEs Through Their MOFAs, by Industry of Affiliate and by Country of Affiliate," October 8, 2024, and USDOC, BEA, table 5.1, "Services Supplied to U.S. Persons by Foreign MNEs," October 8, 2024. Note: MNEs = multinational enterprises; MOFAs = majority-owned foreign affiliates; MOUSAs = majority-owned U.S. affiliates; UBO = ultimate beneficial owner.

Table B.21 Architectural, engineering, scientific, and other technical services: U.S. cross-border exports and imports, 2019–23

Year	U.S. cross-border exports	U.S. cross-border imports
2019	12,219	7,906
2020	10,080	9,035
2021	10,104	11,722
2022	11,482	15,157
2023	12,109	12,681

Source: USDOC, BEA, table 2.1, "U.S. Trade in Services, by Type of Service," July 3, 2024.
Year	Services supplied by U.S. firms' foreign affiliates	Services supplied by U.S. affiliates of foreign firms
2018	32,278	17,258
2019	30,176	19,509
2020	d.s.	19,504
2021	28,574	19,138
2022	29,805	22,151

Table B.22 Architectural, engineering and related services: U.S. affiliate sales and purchases, 2018–22 In millions of dollars. d.s. = data are suppressed to protect confidentiality. This table corresponds to figure 2.11.

Source: USDOC, BEA, table 4.1, "Services Supplied to Foreign Persons by U.S. MNEs Through Their MOFAs, by Industry of Affiliate and by Country of Affiliate," October 8, 2024, and USDOC, BEA, table 5.1, "Services Supplied to U.S. Persons by Foreign MNEs," October 8, 2024. Note: MNEs = multinational enterprises; MOFAs = majority-owned foreign affiliates; MOUSAs = majority-owned U.S. affiliates; UBO = ultimate beneficial owner.

Year	U.S. cross-border exports	U.S. cross-border imports
2019	2,566	5,028
2020	3,337	6,237
2021	3,247	6,546
2022	3,333	7,544
2023	4,213	8,378

Table B.23 Accounting and auditing services: U.S. cross-border exports and imports, 2019–23 In millions of dollars. This table corresponds to figure 2.12.

Source: USDOC, BEA, table 2.1, "U.S. Trade in Services, by Type of Service," July 3, 2024.

Table B.24 Accounting and auditing services: U.S. affiliate sales and purchases, 2018–22

In millions of dollars. d.s	s. = data are suppressed to	protect confidentiality	v. This table corres	ponds to figure 2.13

Year	Services supplied by U.S. firms' foreign affiliates	Services supplied by U.S. affiliates of foreign firms
2018	14,858	427
2019	14,465	385
2020	13,904	426
2021	15,830	444
2022	16,357	537

Source: USDOC, BEA, table 4.1, "Services Supplied to Foreign Persons by U.S. MNEs Through Their MOFAs, by Industry of Affiliate and by Country of Affiliate," October 8, 2024, and USDOC, BEA, table 5.1, "Services Supplied to U.S. Persons by Foreign MNEs," October 8, 2024. Note: MNEs = multinational enterprises; MOFAs = majority-owned foreign affiliates; MOUSAs = majority-owned U.S. affiliates; UBO = ultimate beneficial owner. Because of rounding, figures may not add to 100 percent.

Table B.25 Education services:	U.S. cross-border ex	ports and imports,	2019-23
--------------------------------	----------------------	--------------------	---------

In millions of dollars. This table corresponds to figure 2.14.

Year	U.S. cross-border exports	U.S. cross-border imports
2019	50,141	12,487
2020	42,895	7,530
2021	39,153	8,930

U.S. International Trade Commission | 181

Year	U.S. cross-border exports	U.S. cross-border imports
2022	45,360	11,083
2023	54,848	13,664

Source: USDOC, BEA, table 2.1, "U.S. Trade in Services, by Type of Service," July 3, 2024.

Table B.26 Education services: U.S. affiliate sales and purchases, 2018–22 In millions of dollars. This table corresponds to <u>figure 2.15</u>.

Year	Services supplied by U.S. firms' foreign affiliates	Services supplied by U.S. affiliates of foreign firms
2018	7,179	2,569
2019	6,243	2,339
2020	4,303	2,478
2021	3,998	1,963
2022	3,967	3,121

Source: USDOC, BEA, table 4.1, "Services Supplied to Foreign Persons by U.S. MNEs Through Their MOFAs, by Industry of Affiliate and by Country of Affiliate," October 8, 2024, and USDOC, BEA, table 5.1, "Services Supplied to U.S. Persons by Foreign MNEs," October 8, 2024. Note: MNEs = multinational enterprises; MOFAs = majority-owned foreign affiliates; MOUSAs = majority-owned U.S. affiliates; UBO = ultimate beneficial owner. Because of rounding, figures may not add to 100 percent.

Year	U.S. cross-border exports	U.S. cross-border imports
2019	65,697	39,436
2020	74,873	41,443
2021	90,775	46,521
2022	103,664	48,389
2023	104,441	53,672

Table B.27 Management consulting services: U.S. cross-border exports and imports, 2019–23 In millions of dollars. This table corresponds to figure 2.16.

Source: USDOC, BEA, table 2.1, "U.S. Trade in Services, by Type of Service," July 3, 2024.

Table B.28 Management, scientific, and technical consulting: U.S. affiliate purchases, 2018–22 In millions of dollars. d.s. = data are suppressed to protect confidentiality. This table corresponds to figure 2.17.

Year	Services supplied by U.S. affiliates of foreign firms
2018	26,280
2019	26,332
2020	23,128
2021	24,434
2022	30,394

Source: USDOC, BEA, table 4.1, "Services Supplied to Foreign Persons by U.S. MNEs," October 8, 2024; USDOC, BEA, table 5.1, "Services Supplied to U.S. Persons by Foreign MNEs," October 8, 2024.

Note: MNEs = multinational enterprises.

Year	U.S. cross-border exports	U.S. cross-border imports	
2019	21,523	6,002	
2020	20,776	4,309	
2021	23,280	5,395	
2022	22,280	5,882	
2023	24,295	6,660	

Table B.29 Advertising services: U.S. cross-border exports and imports,	2019–23	3
In millions of dollars. This table corresponds to figure 2.18.		

Source: USDOC, BEA, table 2.1, "U.S. Trade in Services, by Type of Service," July 3, 2024.

Table B.30 Advertising and related services: U.S. affiliate sales and purchases, 2018	3–22
---	------

In millions of dollars. d.s. = data are suppressed to protect confidentiality. This table corresponds to figure 2.19.

Year	Services supplied by U.S. firms' foreign affiliates	Services supplied by U.S. affiliates of foreign firms
2018	17,243	38,684
2019	15,814	38,753
2020	14,373	38,033
2021	d.s.	39,208
2022	d.s.	42,874

Source: USDOC, BEA, table 4.1, "Services Supplied to Foreign Persons by U.S. MNEs," October 8, 2024; USDOC, BEA, table 5.1, "Services Supplied to U.S. Persons by Foreign MNEs," October 8, 2024.

Notes: MNEs = multinational enterprises. Services supplied by foreign affiliates of U.S. firms in 2021 and 2022 are not shown because data were suppressed by the BEA to protect confidentiality.

Table B.31 Business enterprise research and development (BERD) spending in the top five countries,2017 and 2021

In billions of U.S. dollars. This table corresponds to figure 2.20.

Country	2017	2021
United States	405.4	565.4
China	341.8	488.5
Japan	132.9	135.1
South Korea	70.0	86.9
Germany	86.0	86.6

Source: OECD, "Main Science and Technology Indicators," March 2024.

Note: Due to classification issues, data for China is not strictly comparable to data for other economies.

Table B.32 R&D services: U.S. cross-border exports and imports, 2019–23

In millions of dollars. R&D = research and development. This table corresponds to figure 2.21.

Year	U.S. cross-border exports	U.S. cross-border imports
2019	102,845	56,829
2020	97,248	56,085
2021	112,085	61,310
2022	124,919	70,715
2023	118,890	57,274

Source: USDOC, BEA, table 2.1, "U.S. Trade in Services, by Type of Service," July 3, 2024.

Note: Exports and imports of R&D services include sale of licenses for outcomes of research and development and nonroutine testing and product development activities.

		Licenses for the use of outcomes	
Country	Trade Flow	of R&D	R&D services
Ireland	Export	13.8	15.0
Ireland	Import	1.4	3.1
Switzerland	Export	13.8	9.1
Switzerland	Import	3.3	0.6
Netherlands	Export	12.5	1.2
Netherlands	Import	1.5	1.3
Japan	Export	2.2	3.2
Japan	Import	7.7	0.8
India	Export	0.7	0.1
India	Import	1.2	4.4
United Kingdom	Export	1.6	1.7
United Kingdom	Import	2.4	2.7

Table B.33 R&D services: U.S. cross-border exports and imports, select countries, 2023In billions of dollars. This table corresponds to figure 2.22.

Source: USDOC, BEA, table 2.2, "U.S. Trade in Services, by Type of Service and by Country or Affiliation," December 3, 2024.

In minutions of donards. Rob – Research and development. This table corresponds to figure 2.2.	In millions of dollars.	. R&D = Research	and development.	. This table corres	ponds to fi	gure 2.23
--	-------------------------	------------------	------------------	---------------------	-------------	-----------

Year	Services supplied by U.S. firms' foreign affiliates	Services supplied by U.S. affiliates of foreign firms
2018	20,589	2,976
2019	9,399	3,886
2020	9,875	4,134
2021	11,286	5,293
2022	12,426	6,680

Source: USDOC, BEA, table 4.1, "Services Supplied to Foreign Persons by U.S. MNEs Through Their MOFAs, by Industry of Affiliate and by Country of Affiliate," October 8, 2024, and USDOC, BEA, table 5.1, "Services Supplied to U.S. Persons by Foreign MNEs," October 8, 2024. Note: MNEs = multinational enterprises

Table B.35 Total number of accountants and auditors in the United States and total revenue foraccounting services in the United States, 2019–23

Year	Revenue (million \$)	Number of accountants (thousands)
2019	41,744	1,964
2020	42,868	1,696
2021	46,169	1,630
2022	49,717	1,652
2023	53,236	1,624

In millions of dollars and thousands of accountants. This table corresponds to figure 3.1.

Appendix B: Corresponding Tables: Corresponding Data Tables for Figures

Source: Census, "Accounting, Tax Preparation, Bookkeeping, and Payroll Services: U.S. Total," Business and Industry, accessed January 8, 2025; BLS, "Labor Force Statistics-Employed–Accountants and Auditors," Current Population Survey, accessed January 8, 2025.

1 0	· · ·				
Year	Facebook	Instagram	TikTok	Snapchat	Х
2022	49.7	32.9	9.3	5.2	2.8
2023	49.6	33.2	11.6	4.3	1.3
2024	47.3	35.6	12.2	4.0	0.9

Table B.36 Share of total U.S. social media ad spending, by platform, 2022–24 In percentages. This table corresponds to figure 4.1.

Source: Emarketer, "US Social Media Ad Spending Share, by Platform, 2022–2024," August 22, 2

Recent Trends in U.S. Services Trade: 2025 Annual Report

Appendix C Categories by BEA Industries

Sector	BEA industries
Distribution services	Sea transport
	Air transport (freight and port)
	Other modes of transport
	Trade-related services
	 Franchises and trademarks licensing fees
Digital and electronic	 Audio-visual and related products
services	Telecommunications services
	Computer services
	Information services
	Licenses for reproduction of audio-visual and computer software
Financial services	Insurance services
	Financial services
Professional services	 Maintenance and repair services not included elsewhere
	Research and development services and licenses for the use of outcomes of
	research and development
	Legal services
	 Accounting, auditing, and bookkeeping services
	 Business and management consulting and public relations services
	Advertising
	Architectural services
	Engineering services
	Scientific and other technical services
	 Health services and travel services (health related)
	 Education services and travel services (education related)
	Other business services, not included elsewhere
	Waste treatment and depollution
Travel services n.e.s.o.i.	• Business
	Other personal
	Air transport, passenger
All other services	Construction
	Agricultural and mining services
	Operating leasing services
	Artistic-related services
	Heritage and recreational services

 Table C.37 Cross-border services trade by sector or industry

BEA = Bureau of Economic Analysis. n.e.s.o.i. = not elsewhere specified or included.

Source: USITC compiled sectors based on USDOC, BEA, categories presented in table 2.1, "U.S. Trade in Services, by Type of Service," July 3, 2024.

Table C.2 Professional services and related travel services cross-border trade by subsector or industry for chapter 2

Subsectors	BEA services types
Accounting, auditing, and related services	 Accounting, auditing, bookkeeping, and tax consulting services
Advertising and related services	Advertising and related services
Architectural and	Architectural services
engineering services	Engineering services
Education services and	Education services
related travel services	Education related (personal travel)
Health services and related	Health services
travel services	Health related (personal travel)
Legal services	Legal services
Maintenance and repair of services <u>n.e.s.o.i.</u>	Maintenance and repair services, not elsewhere specified or included
Management consulting services	 Business and management consulting and public relations services
Research and development	Research and development services
services	 Licenses for the use of outcomes of research and development
All other professional	Scientific and other technical services
services	Waste treatment and depollution
	Other business services, not included elsewhere

BEA = Bureau of Economic Analysis. n.e.s.o.i. = not elsewhere specified or included.

Source: USITC compiled sectors based on USDOC, BEA, categories presented in table 2.1, "U.S. Trade in Services, by Type of Service," July 3, 2024.

Sector	Affiliate services trades
Agriculture, forestry,	Agriculture, forestry, fishing, and hunting
fishing, and hunting	
Distribution services	Wholesale trade
	Retail trade
	Transportation and warehousing
Digital and electronic services	 Motion picture and sound recording industries
	Telecommunications
	Broadcasting
	 Data processing, hosting, and related services
	 Computer systems design and related services
	Software publishing
	Other information services
Financial services	Finance and insurance
	 Rental and leasing (except real estate)
Manufacturing	Manufacturing
Mining	Mining
Professional services	 Architectural, engineering, and other technical services
	 Management, scientific and technical consulting
	Legal services
	 Accounting, tax preparation, bookkeeping, and payroll services
	Specialized design services
	Scientific research and development services
	Advertising and related services
	Other professional, scientific, and technical
	 Management of nonbank companies and enterprises
	 Waste management and remediation services
	Health care and social assistance
	Educational services
All other services	 Newspaper, periodical, book, and database publishers
	Utilities
	Construction
	Administrative and support services
	Accommodations and food services
	Arts, entertainment, and recreation
	 Other services (except public administration and private households)
	Real estate

Table C.3 Affiliate services trade by sector or industry BEA = Bureau of Economic Analysis.

Source: USITC compiled sectors based on USDOC, BEA, categories presented in table 4.1, "Services Supplied to Foreign Persons by U.S. MNEs Through Their MOFAs, by Industry and Affiliate and by Country of Affiliate," October 8, 2024, and table 5.1, "Services Supplied to U.S. Persons by Foreign MNEs Through Their MOUSA, by Industry of Affiliate and by Country of UBO," October 8, 2024.